ATMOS ENERGY CORP Form 10-K November 16, 2009

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

#### Form 10-K

(Mark One)

**ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934** 

For the fiscal year ended September 30, 2009

OR

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

# Commission file number 1-10042 Atmos Energy Corporation

(Exact name of registrant as specified in its charter)

Texas and Virginia

(State or other jurisdiction of incorporation or organization)
Three Lincoln Centre, Suite 1800

5430 LBJ Freeway, Dallas, Texas

(Address of principal executive offices)

75-1743247

(IRS employer identification no.) **75240** 

(Zip code)

Registrant s telephone number, including area code: (972) 934-9227 Securities registered pursuant to Section 12(b) of the Act:

**Title of Each Class** 

Name of Each Exchange on Which Registered

Common stock, No Par Value

New York Stock Exchange

## Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes b No o

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes o No b

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was

required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes b No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files).\* Yes o No o

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of registrant s knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of large accelerated filer, accelerated filer and smaller reporting company in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer b Accelerated filer o Non-accelerated filer o Smaller reporting company o

(Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes o No b

The aggregate market value of the common voting stock held by non-affiliates of the registrant as of the last business day of the registrant s most recently completed second fiscal quarter, March 31, 2009, was \$2,072,764,690.

As of November 8, 2009, the registrant had 92,599,896 shares of common stock outstanding.

#### DOCUMENTS INCORPORATED BY REFERENCE

Portions of the registrant s Definitive Proxy Statement to be filed for the Annual Meeting of Shareholders on February 3, 2010 are incorporated by reference into Part III of this report.

<sup>\*</sup> The registrant has not been phased into the interactive data requirements.

# TABLE OF CONTENTS

		Page
Glossary of Key	<u>y Terms</u>	3
	Part I	
<u>Item 1.</u>	<u>Business</u>	4
Item 1A.	Risk Factors	22
Item 1B.	<u>Unresolved Staff Comments</u>	27
<u>Item 2.</u>	<u>Properties</u>	27
<u>Item 3.</u>	<u>Legal Proceedings</u>	29
<u>Item 4.</u>	Submission of Matters to a Vote of Security Holders	29
	<u>Part II</u>	
<u>Item 5.</u>	Market for Registrant s Common Equity, Related Stockholder Matters and Issuer	
	Purchases of Equity Securities	31
<u>Item 6.</u>	Selected Financial Data	34
<u>Item 7.</u>	Management s Discussion and Analysis of Financial Condition and Results of	
	<u>Operations</u>	35
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	63
Item 8.	Financial Statements and Supplementary Data	65
<u>Item 9.</u>	Changes in and Disagreements with Accountants on Accounting and Financial	120
Itaan 0.4	<u>Disclosure</u>	128
<u>Item 9A.</u> <u>Item 9B.</u>	Controls and Procedures Other Information	128 130
<u> 110111 9D.</u>	Other information	130
	Part III	
<u>Item 10.</u>	Directors, Executive Officers and Corporate Governance	130
<u>Item 11.</u>	Executive Compensation	130
<u>Item 12.</u>	Security Ownership of Certain Beneficial Owners and Management and Related	120
T. 12	Stockholder Matters	130
<u>Item 13.</u>	Certain Relationships and Related Transactions, and Director Independence	130
<u>Item 14.</u>	Principal Accountant Fees and Services	130
	Part IV	
<u>Item 15.</u>	Exhibits and Financial Statement Schedules	131
<u>EX-10.4(B)</u> <u>EX-12</u>		
EX-21		
EX-23.1		
EX-31 EX-32		
L/X <sup>-</sup> J <u>L</u>		

#### **GLOSSARY OF KEY TERMS**

AEC Atmos Energy Corporation
AEH Atmos Energy Holdings, Inc.
AEM Atmos Energy Marketing, LLC
APS Atmos Pipeline and Storage, LLC

ATO Trading symbol for Atmos Energy Corporation common stock on the New

York Stock Exchange

Bcf Billion cubic feet

COSO Committee of Sponsoring Organizations of the Treadway Commission

FASB Financial Accounting Standards Board FERC Federal Energy Regulatory Commission

Fitch Fitch Ratings, Ltd.

GRIP Gas Reliability Infrastructure Program
GSRS Gas System Reliability Surcharge

ISRS Infrastructure System Replacement Surcharge KPSC Kentucky Public Service Commission

LTIP 1998 Long-Term Incentive Plan

Mcf Thousand cubic feet

MDWQ Maximum daily withdrawal quantity

MMcf Million cubic feet

Moody s Moody s Investor Services, Inc.
NYMEX New York Mercantile Exchange, Inc.

NYSE
RRC
Railroad Commission of Texas
RRM
Rate Review Mechanism
RSC
Rate Stabilization Clause
S&P
Standard & Poor s Corporation

SEC United States Securities and Exchange Commission

Settled Cities Represents 438 of the 439 incorporated cities, or approximately 80 percent

of the Mid-Tex Division s customers, with whom a settlement agreement was

reached during the fiscal 2008 second quarter.

SRF Stable Rate Filing

TXU Gas Company, which was acquired on October 1, 2004

WNA Weather Normalization Adjustment

3

#### **PART I**

The terms we, our, us, Atmos Energy and the Company refer to Atmos Energy Corporation and its subsidiaries, the context suggests otherwise.

#### ITEM 1. Business.

# **Overview and Strategy**

Atmos Energy Corporation, headquartered in Dallas, Texas, is engaged primarily in the regulated natural gas distribution and transmission and storage businesses as well as other nonregulated natural gas businesses. Since our incorporation in Texas in 1983, we have grown primarily through a series of acquisitions, the most recent of which was the acquisition in October 2004 of the natural gas distribution and pipeline operations of TXU Gas Company. We are also incorporated in the state of Virginia.

Today, we distribute natural gas through regulated sales and transportation arrangements to over 3 million residential, commercial, public authority and industrial customers in 12 states located primarily in the South, which makes us one of the country s largest natural-gas-only distributors based on number of customers. We also operate one of the largest intrastate pipelines in Texas based on miles of pipe.

Through our nonregulated businesses, we primarily provide natural gas management and marketing services to municipalities, other local gas distribution companies and industrial customers primarily in the Midwest and Southeast and natural gas transportation along with storage services to certain of our natural gas distribution divisions and third parties.

Our overall strategy is to:

deliver superior shareholder value,

improve the quality and consistency of earnings growth, while operating our regulated and nonregulated businesses exceptionally well and

enhance and strengthen a culture built on our core values.

We have experienced more than 25 consecutive years of increasing dividends and earnings growth after giving effect to our acquisitions. Historically, we achieved this record of growth through acquisitions while efficiently managing our operating and maintenance expenses and leveraging our technology to achieve more efficient operations. In recent years, we have also achieved growth by implementing rate designs that reduce or eliminate regulatory lag and separate the recovery of our approved margins from customer usage patterns. In addition, we have developed various commercial opportunities within our regulated transmission and storage operations. Finally, we have strengthened our nonregulated businesses by increasing sales volumes and improving per-unit margins.

Our core values include focusing on our employees and customers while conducting our business with honesty and integrity. We continue to strengthen our culture through ongoing communications with our employees and enhanced employee training.

#### **Operating Segments**

We operate the Company through the following four segments:

The *natural gas distribution segment*, which includes our regulated natural gas distribution and related sales operations.

The *regulated transmission and storage segment*, which includes the regulated pipeline and storage operations of our Atmos Pipeline Texas Division.

The *natural gas marketing segment*, which includes a variety of nonregulated natural gas management services.

4

#### **Table of Contents**

The *pipeline*, *storage* and *other segment*, which is comprised of our nonregulated natural gas gathering, transmission and storage services.

These operating segments are described in greater detail below.

#### **Natural Gas Distribution Segment Overview**

Our natural gas distribution segment consists of the following six regulated divisions, presented in order of total customers served, covering service areas in 12 states:

Atmos Energy Mid-Tex Division,

Atmos Energy Kentucky/Mid-States Division,

Atmos Energy Louisiana Division,

Atmos Energy West Texas Division,

Atmos Energy Mississippi Division and

Atmos Energy Colorado-Kansas Division

Our natural gas distribution business is a seasonal business. Gas sales to residential and commercial customers are greater during the winter months than during the remainder of the year. The volumes of gas sales during the winter months will vary with the temperatures during these months.

Revenues in this operating segment are established by regulatory authorities in the states in which we operate. These rates are intended to be sufficient to cover the costs of conducting business and to provide a reasonable return on invested capital. Our primary service areas are located in Colorado, Kansas, Kentucky, Louisiana, Mississippi, Tennessee and Texas. We have more limited service areas in Georgia, Illinois, Iowa, Missouri and Virginia. In addition, we transport natural gas for others through our distribution system.

Rates established by regulatory authorities often include cost adjustment mechanisms for costs that (i) are subject to significant price fluctuations compared to our other costs, (ii) represent a large component of our cost of service and (iii) are generally outside our control.

Purchased gas cost mechanisms represent a common form of cost adjustment mechanism. Purchased gas cost adjustment mechanisms provide natural gas utility companies a method of recovering purchased gas costs on an ongoing basis without filing a rate case because they provide a dollar-for-dollar offset to increases or decreases in natural gas distribution gas costs. Therefore, although substantially all of our natural gas distribution operating revenues fluctuate with the cost of gas that we purchase, natural gas distribution gross profit (which is defined as operating revenues less purchased gas cost) is generally not affected by fluctuations in the cost of gas.

Additionally, some jurisdictions have introduced performance-based ratemaking adjustments to provide incentives to natural gas utilities to minimize purchased gas costs through improved storage management and use of financial instruments to lock in gas costs. Under the performance-based ratemaking adjustment, purchased gas costs savings are shared between the utility and its customers.

Finally, regulatory authorities have approved weather normalization adjustments (WNA) for over 90 percent of residential and commercial meters in our service areas as a part of our rates. WNA minimizes the effect of weather that is above or below normal by allowing us to increase customers bills to offset lower gas usage when weather is warmer than normal and decrease customers bills to offset higher gas usage when weather is colder than normal.

5

#### **Table of Contents**

As of September 30, 2009 we had WNA for our residential and commercial meters in the following service areas for the following periods:

Georgia October May Kansas October May Kentucky November April Louisiana December March Mississippi November April Tennessee November April Texas: Mid-Tex November April Texas: West Texas October May Virginia January December

Financial results for this segment are affected by the cost of natural gas and economic conditions in the areas that we serve. As discussed above, we are generally able to pass the cost of gas through to our customers under purchased gas adjustment clauses; therefore, the cost of gas typically does not have a direct impact on our gross profit. However, higher gas costs may cause customers to conserve or, in the case of industrial customers, to use alternative energy sources. Higher gas costs may also adversely impact our accounts receivable collections, resulting in higher bad debt expense and may require us to increase borrowings under our credit facilities resulting in higher interest expense.

Our supply of natural gas is provided by a variety of suppliers, including independent producers, marketers and pipeline companies and withdrawals of gas from proprietary and contracted storage assets. Additionally, the natural gas supply for our Mid-Tex Division includes peaking and spot purchase agreements.

Supply arrangements consist of both base load and swing supply (peaking) quantities and are contracted from our suppliers on a firm basis with various terms at market prices. Base load quantities are those that flow at a constant level throughout the month and swing supply quantities provide the flexibility to change daily quantities to match increases or decreases in requirements related to weather conditions.

Currently, all of our natural gas distribution divisions, except for our Mid-Tex Division, utilize 39 pipeline transportation companies, both interstate and intrastate, to transport our natural gas. The pipeline transportation agreements are firm and many of them have pipeline no-notice storage service, which provides for daily balancing between system requirements and nominated flowing supplies. These agreements have been negotiated with the shortest term necessary while still maintaining our right of first refusal. The natural gas supply for our Mid-Tex Division is delivered by our Atmos Pipeline Texas Division.

Except for local production purchases, we select our natural gas suppliers through a competitive bidding process by requesting proposals from suppliers that have demonstrated that they can provide reliable service. We select these suppliers based on their ability to deliver gas supply to our designated firm pipeline receipt points at the lowest cost. Major suppliers during fiscal 2009 were Anadarko Energy Services, Chesapeake Energy Marketing, Inc., ConocoPhillips Company, Devon Gas Services, L.P., Enbridge Marketing (US) L.P., Iberdrola Renewables, Inc., National Fuel Marketing Company, LLC, ONEOK Energy Services Company L.P., Tenaska Marketing and Atmos Energy Marketing, LLC, our natural gas marketing subsidiary.

The combination of base load, peaking and spot purchase agreements, coupled with the withdrawal of gas held in storage, allows us the flexibility to adjust to changes in weather, which minimizes our need to enter into long-term firm commitments. We estimate our peak-day availability of natural gas supply to be approximately 4.2 Bcf. The peak-day demand for our natural gas distribution operations in fiscal 2009 was on January 15, 2009, when sales to

customers reached approximately 3.1 Bcf.

To maintain our deliveries to high priority customers, we have the ability, and have exercised our right, to curtail deliveries to certain customers under the terms of interruptible contracts or applicable state regulations or statutes. Our customers demand on our system is not necessarily indicative of our ability to meet current or anticipated market demands or immediate delivery requirements because of factors such as the physical limitations of gathering, storage and transmission systems, the duration and severity of cold weather, the

6

#### **Table of Contents**

availability of gas reserves from our suppliers, the ability to purchase additional supplies on a short-term basis and actions by federal and state regulatory authorities. Curtailment rights provide us the flexibility to meet the human-needs requirements of our customers on a firm basis. Priority allocations imposed by federal and state regulatory agencies, as well as other factors beyond our control, may affect our ability to meet the demands of our customers. We anticipate no problems with obtaining additional gas supply as needed for our customers.

The following briefly describes our six natural gas distribution divisions. We operate in our service areas under terms of non-exclusive franchise agreements granted by the various cities and towns that we serve. At September 30, 2009, we held 1,111 franchises having terms generally ranging from five to 35 years. A significant number of our franchises expire each year, which require renewal prior to the end of their terms. We believe that we will be able to renew our franchises as they expire. Additional information concerning our natural gas distribution divisions is presented under the caption Operating Statistics .

Atmos Energy Mid-Tex Division. Our Mid-Tex Division serves approximately 550 incorporated and unincorporated communities in the north-central, eastern and western parts of Texas, including the Dallas/Fort Worth Metroplex. The governing body of each municipality we serve has original jurisdiction over all gas distribution rates, operations and services within its city limits, except with respect to sales of natural gas for vehicle fuel and agricultural use. The Railroad Commission of Texas (RRC) has exclusive appellate jurisdiction over all rate and regulatory orders and ordinances of the municipalities and exclusive original jurisdiction over rates and services to customers not located within the limits of a municipality.

Prior to fiscal 2008, this division operated under one system-wide rate structure. However, in 2008, we reached a settlement with cities representing approximately 80 percent of this division s customers (Settled Cities) that has allowed us, beginning in 2008, to update rates for customers in these cities through an annual rate review mechanism. Rates for the remaining 20 percent of this division s customers, primarily the City of Dallas, continue to be updated through periodic formal rate proceedings and filings made under Texas Gas Reliability Infrastructure Program (GRIP). GRIP allows us to include in our rate base annually approved capital costs incurred in the prior calendar year provided that we file a complete rate case at least once every five years.

Atmos Energy Kentucky/Mid-States Division. Our Kentucky/Mid-States Division operates in more than 420 communities across Georgia, Illinois, Iowa, Kentucky, Missouri, Tennessee and Virginia. The service areas in these states are primarily rural; however, this division serves Franklin, Tennessee, and other suburban areas of Nashville. We update our rates in this division through periodic formal rate filings made with each state s public service commission.

Atmos Energy Louisiana Division. In Louisiana, we serve nearly 300 communities, including the suburban areas of New Orleans, the metropolitan area of Monroe and western Louisiana. Direct sales of natural gas to industrial customers in Louisiana, who use gas for fuel or in manufacturing processes, and sales of natural gas for vehicle fuel are exempt from regulation and are recognized in our natural gas marketing segment. Our rates in this division are updated annually through a rate stabilization clause filing without filing a formal rate case.

Atmos Energy West Texas Division. Our West Texas Division serves approximately 80 communities in West Texas, including the Amarillo, Lubbock and Midland areas. Like our Mid-Tex Division, each municipality we serve has original jurisdiction over all gas distribution rates, operations and services within its city limits, with the RRC having exclusive appellate jurisdiction over the municipalities and exclusive original jurisdiction over rates and services provided to customers not located within the limits of a municipality. Prior to fiscal 2008, rates were updated in this division through formal rate proceedings. However, the West Texas Division entered into agreements with its West Texas service areas during 2008 and its Amarillo and Lubbock service area during 2009 to update rates for customers in these service areas through an annual rate review mechanism.

*Atmos Energy Mississippi Division*. In Mississippi, we serve about 110 communities throughout the northern half of the state, including the Jackson metropolitan area. Our rates in the Mississippi Division are updated annually through a stable rate filing without filing a formal rate case.

7

#### **Table of Contents**

Atmos Energy Colorado-Kansas Division. Our Colorado-Kansas Division serves approximately 170 communities throughout Colorado and Kansas and parts of Missouri, including the cities of Olathe, Kansas, a suburb of Kansas City and Greeley, Colorado, located near Denver. We update our rates in this division through periodic formal rate filings made with each state spublic service commission.

The following table provides a jurisdictional rate summary for our regulated operations. This information is for regulatory purposes only and may not be representative of our actual financial position.

exas Cexas Colorado Cansas Georgia	Action  5/24/04  4/28/09 10/1/07 5/12/08 9/22/08	\$417,111 755,038 81,208 (2)	Return <sup>(1)</sup> 8.258% 8.258% 8.45%	Equity <sup>(1)</sup> 10.00% 10.00% 11.25%
exas Colorado Cansas Georgia	4/28/09 10/1/07 5/12/08	755,038 81,208	8.258% 8.45%	10.00%
Colorado Cansas Georgia	10/1/07 5/12/08	81,208	8.45%	
Cansas Georgia	5/12/08	*		11 25%
leorgia		(2)	(2)	11.25/0
_	9/22/08		(2)	(2)
1	, . <del></del>	66,893	7.75%	10.70%
linois	11/1/00	24,564	9.18%	11.56%
owa	3/1/01	5,000	(2)	11.00%
Centucky	8/1/07	(2)	(2)	(2)
Iissouri	3/4/07	(2)	(2)	(2)
ennessee	4/1/09	190,100	8.24%	10.30%
			8.46% -	9.50% -
'irginia	9/30/08	33,194	8.96%	10.50%
				10.00% -
rans LA	4/1/09	96,570	(2)	10.80%
GS	7/1/09	236,600	(2)	10.40%
'exas	8/1/09	$1,262,969^{(3)}$	7.78%	9.60%
'exas	6/24/08	$1,127,924^{(3)}$	7.98%	10.00%
Iississippi	1/1/05	196,801	8.23%	9.80%
marillo	9/1/03	36,844	9.88%	12.00%
ubbock	3/1/04	43,300	9.15%	11.25%
Vest Texas	8/1/09	124,401	(2)	9.60%
	Q			
o'ile	inois wa entucky issouri ennessee arginia ans LA GS exas exas ississippi marillo ibbock	inois 11/1/00 wa 3/1/01 entucky 8/1/07 issouri 3/4/07 ennessee 4/1/09 irginia 9/30/08 rans LA 4/1/09 GS 7/1/09 exas 8/1/09 exas 6/24/08 ississispi 1/1/05 marillo 9/1/03 ibbock 3/1/04	eorgia 9/22/08 66,893 inois 11/1/00 24,564 wa 3/1/01 5,000 entucky 8/1/07 (2) issouri 3/4/07 (2) ennessee 4/1/09 190,100  arginia 9/30/08 33,194  ans LA 4/1/09 96,570 GS 7/1/09 236,600 exas 8/1/09 1,262,969(3) exas 6/24/08 1,127,924(3) ississispi 1/1/05 196,801 marillo 9/1/03 36,844 abbock 3/1/04 43,300 fest Texas 8/1/09 124,401	ansas 5/12/08 (2) (2) eorgia 9/22/08 66,893 7.75% inois 11/1/00 24,564 9.18% wa 3/1/01 5,000 (2) entucky 8/1/07 (2) (2) issouri 3/4/07 (2) (2) ennessee 4/1/09 190,100 8.24% eriginia 9/30/08 33,194 8.96%  ans LA 4/1/09 96,570 (2) exas 8/1/09 1,262,969(3) 7.78% exas 6/24/08 1,127,924(3) 7.98% ississispi 1/1/05 196,801 8.23% emarillo 9/1/03 36,844 9.88% elbbock 3/1/04 43,300 9.15% eest Texas 8/1/09 124,401 (2)

			Bad		Performance- Based	G 4
Division	Jurisdiction	Authorized Debt/ Equity Ratio	Debt Rider <sup>(4)</sup>	WNA	Rate Program <sup>(5)</sup>	Customer Meters
Atmos Pipeline Texas	Texas	50/50	No	N/A	N/A	N/A
Colorado-Kansas	Colorado	54/46	No(7)	No	No	111,382
	Kansas	(2)	Yes	Yes	No	129,983
Kentucky/Mid-States	Georgia	55/45	No	Yes	Yes	65,080
	Illinois	67/33	No	No	No	22,623
	Iowa	57/43	No	No	No	4,344
	Kentucky	(2)	No(7)	Yes	Yes	175,789
	Missouri	(2)	No	No(6)	No	57,332
	Tennessee	52/48	Yes	Yes	Yes	132,764
	Virginia	55/45	Yes	Yes	No	23,182
Louisiana	Trans LA	52/48	No	Yes	No	78,345
	LGS	52/48	No	Yes	No	277,648
Mid-Tex Settled Cities Mid-Tex Dallas &	Texas	52/48	Yes	Yes	No	1,227,598
Environs	Texas	52/48	Yes	Yes	No	306,899
Mississippi	Mississippi	47/53	No(7)	Yes	No	266,785
West Texas	Amarillo	50/50	Yes	Yes	No	69,836
	Lubbock	50/50	Yes	Yes	No	73,642
	West Texas	52/48	Yes	Yes	No	155,612

- (1) The rate base, authorized rate of return and authorized return on equity presented in this table are those from the last rate case or GRIP filing for each jurisdiction. These rate bases, rates of return and returns on equity are not necessarily indicative of current or future rate bases, rates of return or returns on equity.
- (2) A rate base, rate of return, return on equity or debt/equity ratio was not included in the respective state commission s final decision.
- (3) The Mid-Tex Rate Base amounts for the Settled Cities and Dallas and Environs both represent system-wide, or 100 percent, of the Mid-Tex Division s rate base. The difference in rate base amounts is due to two separate test filing periods covered.
- (4) The bad debt rider allows us to recover from ratepayers the gas cost portion of uncollectible accounts.
- (5) The performance-based rate program provides incentives to natural gas utility companies to minimize purchased gas costs by allowing the utility company and its customers to share the purchased gas costs savings.
- (6) The Missouri jurisdiction has a straight-fixed variable rate design which decouples gross profit margin from customer usage patterns.
- (7) The Company has pending requests in Colorado, Kentucky and Mississippi to move bad debt cost to the gas cost recovery mechanism. A hearing regarding the Mississippi request was held on September 1, 2009.

# Natural Gas Distribution Sales and Statistical Data

	2009	2008	ear Ended Septer 2007	2006	2005
METERS IN SERVICE,					
end of year					
Residential	2,901,577	2,911,475	2,893,543	2,886,042	2,862,822
Commercial	265,843	268,845	272,081	275,577	274,536
Industrial	2,193	2,241	2,339	2,661	2,715
Public authority and other	9,231	9,218	19,164	16,919	17,767
Total meters	3,178,844	3,191,779	3,187,127	3,181,199	3,157,840
INVENTORY STORAGE					
BALANCE Bcf	57.0	58.3	58.0	59.9	54.7
HEATING DEGREE DAYS <sup>(1)</sup>					
Actual (weighted average)	2,713	2,820	2,879	2,527	2,587
Percent of normal	100%	100%	100%	87%	89%
SALES VOLUMES					
MMcf <sup>(2)</sup>					
Gas Sales Volumes Residential	159,762	163,229	166,612	144,780	162,016
Commercial	91,379	93,953	95,514	87,006	92,401
Industrial	18,563	21,734	22,914	26,161	29,434
Public authority and other	12,413	13,760	12,287	14,086	12,432
Total gas salas valumas	202 117	202 676	207 227	272 022	296,283
Total gas sales volumes Transportation volumes	282,117 130,691	292,676 141,083	297,327 135,109	272,033 126,960	122,098
Transportation volumes	130,091	141,063	133,109	120,900	122,098
Total throughput	412,808	433,759	432,436	398,993	418,381
OPERATING REVENUES (000 §3)					
Gas Sales Revenues					
Residential	\$ 1,830,140	\$ 2,131,447	\$ 1,982,801	\$ 2,068,736	\$ 1,791,172
Commercial	838,184	1,077,056	970,949	1,061,783	869,722
Industrial	135,633	212,531	195,060	276,186	229,649
Public authority and other	89,183	137,821	114,298	144,600	114,742
Total gas sales revenues	2,893,140	3,558,855	3,263,108	3,551,305	3,005,285
Transportation revenues	59,914	60,504	59,813	62,215	59,996
Other gas revenues	31,711	35,771	35,844	37,071	37,859
Total operating revenues	\$ 2,984,765	\$ 3,655,130	\$ 3,358,765	\$ 3,650,591	\$ 3,103,140

Edgar Filing: ATMOS ENERGY CORP - Form 10-K

Average transportation					
revenue per Mcf	\$ 0.46	\$ 0.43	\$ 0.44	\$ 0.49	\$ 0.49
Average cost of gas per Mcf					
sold	\$ 6.95	\$ 9.05	\$ 8.09	\$ 10.02	\$ 7.41
Employees	4,691	4,558	4,472	4,402	4,327

See footnotes following these tables.

10

**Table of Contents** 

# Natural Gas Distribution Sales and Statistical Data By Division

					Fiscal	Yea	ar Ended S	epte	ember 30, 2	009					
		K	entucky/				West			C	olorado-				
	Mid-Tex	M	lid-States	L	ouisiana		Texas	Mi	ississippi	]	Kansas	C	ther <sup>(3)</sup>		Tot
RS IN															
CE															
tial	1,417,869		423,829		333,224		270,757		237,289		218,609				2,90
rcial	116,480		53,386		22,769		24,986		26,142		22,080				26
al	148		909				508		532		96				
uthority and															
·			2,555				2,839		2,822		1,015				
	1,534,497		480,679		355,993		299,090		266,785		241,800				3,17
NG DEGREE															
)	2,036		3,853		1,574		3,553		2,746		5,520				
of normal <b>VOLUMES</b>	100%	)	98%		101%		99%		103%		100%				
es Volumes															
tial	73,678		26,589		12,371		16,341		13,503		17,280				15
rcial	48,363		16,049		6,771		6,780		6,568		6,848				9
al	2,918		6,217		•		3,528		5,704		196				1
uthority and	,		•				,		,						
•			1,434				6,014		2,901		2,064				1
	124,959		50,289		19,142		32,663		28,676		26,388				28
rtation volumes	44,991		41,693		5,151		23,417		4,968		10,471				13
roughput	169,950		91,982		24,293		56,080		33,644		36,859				41
ATING															
IN (000 \$3) ATING	\$ 483,155	\$	163,602	\$	118,021	\$	89,982	\$	91,680	\$	78,188	\$		\$	1,02
(SES (000 \$3)															
on and															
ance	\$ 150,978	\$	68,823	\$	41,956	\$	35,126	\$	43,642	\$	32,935	\$	(4,031)	\$	36
ation and	7 -2 - 3,2 / -		00,0_0	_	,	7	,	_	,	_	,	_	(1,00-)	_	
ation	\$ 94,040	\$	32,755	\$	22,492	\$	15,242	\$	12,411	\$	15,334	\$		\$	19
other than	,	Ψ	,	Ψ	, · · · ·	*	, <b> ·-</b>	4	,	+	,	7		~	
	\$ 108,412	\$	13,261	\$	9,629	\$	15,863	\$	13,925	\$	8,222	\$		\$	16
npairments A <b>TING</b>	\$ 2,100	\$	785	\$	510	\$	413	\$	415	\$	376	\$		\$	10
TE (000 \$3)	\$ 127,625	\$	47,978	\$	43,434	\$	23,338	\$	21,287	\$	21,321	\$	4,031	\$	28

Table of Contents 19

33,960

\$ 22,173

\$ 24,726

24,871

\$ 42,626

\$ 57,943

173,201

# AL DITURES

00 s)

CRTY, PLANT QUIPMENT,

\$ 1,615,900

STICS, at year								
f pipe	28,996	12,158	8,321	7,702	6,540	7,162		70
ees	1,585	605	446	352	389	290	1,024	2

\$ 299,242 \$ 266,053

\$ 722,530 \$ 390,957

\$ 124,391 \$ 3,703

\$ 284,398

See footnotes following these tables.

11

# **Table of Contents**

		Fiscal Year Ended September 30, 2008 Kentucky/ West Colorado-														
	]	Mid-Tex		lid-States	L	ouisiana		vvest Texas	M	lississippi		olorado- Kansas	(	Other <sup>(3)</sup>		Tot
RS IN																
CE																
tial		1,414,543		431,880		336,211		270,990		240,113		217,738				2,91
rcial		117,022		54,538		23,059		25,226		27,219		21,781				26
al		163		930				497		562		89				
uthority and																
				2,563				2,888		2,822		945				
		1,531,728		489,911		359,270		299,601		270,716		240,553				3,19
NG DEGREE																
,		2,213		3,799		1,531		3,546		2,741		5,861				,
of normal		99%		96%		99%		99%		101%		105%				
VOLUMES				2,0		5		0				0				
es Volumes																
tial		76,296		26,009		12,475		17,190		12,882		18,377				16
rcial		50,348		15,731		6,858		7,162		6,590		7,264				9
al		3,293		7,740				3,876		6,580		245				2
uthority and																
				1,419				6,933		3,013		2,395				1.
		129,937		50,899		19,333		35,161		29,065		28,281				29
rtation volumes		49,606		44,796		6,136		26,411		4,219		9,915				14
roughput		179,543		95,695		25,469		61,572		33,284		38,196				433
ATING																
IN (000 \$3)	\$	478,622	\$	159,265	\$	110,754	\$	87,344	\$	91,749	\$	78,332	\$		\$	1,00
ATING																
SES (000 \$3)																
on and	_	42= -=		/# 4 · ·	_	40.05=	٠	26.50-		42.05		25.41	_	/A ====		-
ance	\$	167,497	\$	65,161	\$	42,367	\$	36,688	\$	46,024	\$	35,414	\$	(3,907)	\$	38
ation and	<b>*</b>	04.202	<i>*</i>	20.55 :	<b>A</b>	01.102	<i>_</i>	14501	_	11.550	<b>*</b>	14.500	<b>.</b>		<u></u>	
ntion	\$	84,202	\$	30,574	\$	21,193	\$	14,781	\$	11,752	\$	14,703	\$		\$	17′
ther than	Φ	111 014	φ	14.700	φ	0.104	φ	22.022	d	14.002	φ	7.000	φ		d	1.7
TING	\$	111,914	\$	14,799	\$	8,104	\$	22,032	\$	14,003	\$	7,600	\$		\$	173
TING IF (000 &3)	Φ	115 000	Φ	10 721	Φ	30.000	φ	12 042	Φ	10.070	Φ	20.615	Φ	2 007	ø	20
IE (000 \$3) AL	\$	115,009	\$	48,731	\$	39,090	\$	13,843	\$	19,970	\$	20,615	\$	3,907	\$	26
AL DITURES																
DII UNLO	\$	178,409	\$	59,274	\$	46,674	\$	34,354	\$	22,590	\$	20,331	\$	24,910	\$	38
	ф Ф	1,491,188	\$ \$	•	\$ \$	46,674 370,751	\$	•	\$ \$	22,390 254,452	\$	•	\$ \$	127,609	\$ \$	
	Φ	1,471,100	Φ	007,107	Φ	310,131	Ф	210,320	Φ	4J4,4J4	Ф	414,141	Ф	141,009	Φ	2,40.

CRTY, PLANT
QUIPMENT,
00 s)
R
STICS, at year

f pipe	28,697	12,104	8,277	14,697	6,537	7,150		
ees	1,506	635	427	342	393	281	974	

## Notes to preceding tables:

- (1) A heating degree day is equivalent to each degree that the average of the high and the low temperatures for a day is below 65 degrees. The colder the climate, the greater the number of heating degree days. Heating degree days are used in the natural gas industry to measure the relative coldness of weather and to compare relative temperatures between one geographic area and another. Normal degree days are based on National Weather Service data for selected locations. For service areas that have weather normalized operations, normal degree days are used instead of actual degree days in computing the total number of heating degree days.
- (2) Sales volumes, revenues, operating margins, operating expense and operating income reflect segment operations, including intercompany sales and transportation amounts.
- (3) The Other column represents our shared services function, which provides administrative and other support to the Company. Certain costs incurred by this function are not allocated.

# Regulated Transmission and Storage Segment Overview

Our regulated transmission and storage segment consists of the regulated pipeline and storage operations of our Atmos Pipeline Texas Division. This division transports natural gas to our Mid-Tex Division, transports natural gas for third parties and manages five underground storage reservoirs in Texas. We also provide ancillarly services customary in the pipeline industry including parking arrangements, lending and sales of inventory on hand. Parking arrangements provide short-term interruptible storage of gas on our pipeline.

12

#### **Table of Contents**

Lending services provide short-term interruptible loans of natural gas from our pipeline to meet market demands. Gross profit earned from our Mid-Tex Division and through certain other transportation and storage services is subject to traditional ratemaking governed by the RRC. However, Atmos Pipeline Texas existing regulatory mechanisms allow certain transportation and storage services to be provided under market-based rates with minimal regulation.

These operations include one of the largest intrastate pipeline operations in Texas with a heavy concentration in the established natural gas-producing areas of central, northern and eastern Texas, extending into or near the major producing areas of the Texas Gulf Coast and the Delaware and Val Verde Basins of West Texas. Nine basins located in Texas are believed to contain a substantial portion of the nation s remaining onshore natural gas reserves. This pipeline system provides access to all of these basins.

#### Regulated Transmission and Storage Sales and Statistical Data

	Fiscal Year Ended September 30									
	2009	2008	2007	2006	2005					
CUSTOMERS, end of year										
Industrial	68	62	65	67	66					
Other	168	189	196	178	191					
Total	236	251	261	245	257					
PIPELINE TRANSPORTATION										
VOLUMES MMéf)	706,132	782,876	699,006	581,272	554,452					
OPERATING REVENUES (000 \$)	\$ 209,658	\$ 195,917	\$ 163,229	\$ 141,133	\$ 142,952					
Employees, at year end	62	60	54	85	78					

<sup>(1)</sup> Transportation volumes and operating revenues reflect segment operations, including intercompany sales and transportation amounts.

#### **Natural Gas Marketing Segment Overview**

Our natural gas marketing activities are conducted through Atmos Energy Marketing (AEM), which is wholly-owned by Atmos Energy Holdings, Inc. (AEH). AEH is a wholly-owned subsidiary of AEC and operates primarily in the Midwest and Southeast areas of the United States.

AEM s primary business is to aggregate and purchase gas supply, arrange transportation and storage logistics and ultimately deliver gas to customers at competitive prices. In addition, AEM utilizes proprietary and customer-owned transportation and storage assets to provide various services our customers request, including furnishing natural gas supplies at fixed and market-based prices, contract negotiation and administration, load forecasting, gas storage acquisition and management services, transportation services, peaking sales and balancing services, capacity utilization strategies and gas price hedging through the use of financial instruments. AEM serves most of its customers under contracts generally having one to two year terms and sells natural gas to some of its industrial customers on a delivered burner tip basis under contract terms ranging from 30 days to two years. As a result, AEM s margins arise from the types of commercial transactions we have structured with our customers and our ability to identify the lowest cost alternative among the natural gas supplies, transportation and markets to which it has access to serve those customers.

AEM also seeks to maximize, through asset optimization activities, the economic value associated with the storage and transportation capacity we own or control in our natural gas distribution and natural gas marketing segments. We attempt to meet this objective by engaging in natural gas storage transactions in which we seek to find and profit through the arbitrage of pricing differences in various locations and by recognizing pricing differences that occur over time. This process involves purchasing physical natural gas, storing it in the storage and transportation assets to which AEM has access and selling financial instruments at advantageous prices to lock in a gross profit margin.

13

**Table of Contents** 

#### Natural Gas Marketing Sales and Statistical Data

	2009	2008	2007	2006	2005
CUSTOMERS, end of year					
Industrial	631	624	677	679	559
Municipal	63	55	68	73	69
Other	321	312	281	289	211
Total	1,015	991	1,026	1,041	839
INVENTORY STORAGE BALANCE Bcf	17.0	11.0	19.3	15.3	8.2
NATURAL GAS MARKETING SALES VOLUMES MM&	441,081	457,952	423,895	336,516	273,201
OPERATING REVENUES	441,001	437,932	,	550,510	273,201
( <b>000</b> §)	\$ 2,336,847	\$ 4,287,862	\$ 3,151,330	\$ 3,156,524	\$ 2,106,278

<sup>(1)</sup> Sales volumes and operating revenues reflect segment operations, including intercompany sales and transportation amounts.

## Pipeline, Storage and Other Segment Overview

Our pipeline, storage and other segment primarily consists of the operations of Atmos Pipeline and Storage, LLC (APS), which is wholly-owned by AEH. APS is engaged in nonregulated transmission, storage and natural gas gathering services. Its primary asset is a proprietary 21 mile pipeline located in New Orleans, Louisiana. It also owns or controls additional pipeline and storage capacity including interests in underground storage fields in Kentucky and Louisiana that are used to reduce the need of our natural gas distribution divisions to contract for pipeline capacity to meet customer demand during peak periods.

APS primary business is to provide storage and transportation services to our Louisiana and Kentucky/MidStates regulated natural gas distribution divisions, to our natural gas marketing segment and, on a more limited basis, to third parties. APS earns transportation fees and storage demand charges to aggregate and provide gas supply, provide access to storage capacity and transport gas for these customers.

APS also engages in various asset optimization activities. APS primary asset optimization activity involves the administration of two asset management plans with regulated affiliates of the Company. These arrangements provide APS the opportunity to maximize the economic value associated with the transportation and storage capacity assigned to these plans. APS attempts to meet this objective through a variety of activities including engaging in natural gas storage transactions and utilizing excess asset capacity to find and profit through the arbitrage of pricing differences in various locations and by recognizing pricing differences that occur over time. These plans require APS to share a portion of the economic value created by these activities with the regulated customers served by these affiliates. These arrangements have been approved by applicable state regulatory commissions and are subject to annual regulatory review intended to ensure proper allocation of economic value between our regulated customers and APS.

APS also seeks to maximize the economic value associated with the storage and transportation capacity it owns or controls. We attempt to meet this objective by engaging in natural gas storage transactions in which we seek to find and profit through the arbitrage of pricing differences in various locations and by recognizing pricing differences that occur over time. This process involves purchasing physical natural gas, storing it in the storage and transportation assets to which APS has access and, in transactions involving storage capacity, selling financial instruments at advantageous prices to lock in a gross profit margin.

14

#### **Table of Contents**

#### Pipeline, Storage and Other Sales and Statistical Data

		Fiscal Yo	ear Ended Sept	tember 30	
	2009	2008	2007	2006	2005
OPERATING REVENUES (000 \$) PIPELINE TRANSPORTATION	\$ 41,924	\$ 31,709	\$ 33,400	\$ 25,574	\$ 15,639
VOLUMES MMé‡)	6,395	5,492	7,710	9,712	7,593
INVENTORY STORAGE BALANCE					
Bcf	2.9	1.4	2.0	2.6	1.8

<sup>(1)</sup> Transportation volumes and operating revenues reflect segment operations, including intercompany sales and transportation amounts.

#### **Ratemaking Activity**

#### Overview

The method of determining regulated rates varies among the states in which our natural gas distribution divisions operate. The regulatory authorities have the responsibility of ensuring that utilities in their jurisdictions operate in the best interests of customers while providing utility companies the opportunity to earn a reasonable return on their investment. Generally, each regulatory authority reviews rate requests and establishes a rate structure intended to generate revenue sufficient to cover the costs of conducting business and to provide a reasonable return on invested capital.

Our current rate strategy is to focus on reducing or eliminating regulatory lag, obtaining adequate returns and providing stable, predictable margins. Atmos Energy has annual ratemaking mechanisms in place in three states that provide for an annual rate review and adjustment to rates for approximately 68 percent of our customers. Additionally, we have WNA mechanisms in eight states. These mechanisms work in tandem to provide insulation from volatile margins, both for the Company and our customers.

We will also continue to address various rate design changes, including the recovery of bad debt gas costs, inclusion of other taxes in gas costs and stratification of rates to benefit low income households in future rate filings. These design changes would address cost variations that are related to pass-through energy costs beyond our control.

Although substantial progress has been made in recent years by improving rate design across Atmos operating area, potential changes in federal energy policy and adverse economic conditions will necessitate continued vigilance by the Company and our regulators in meeting the challenges presented by these external factors.

15

#### Recent Ratemaking Activity

Substantially all of our natural gas distribution revenues in the fiscal years ended September 30, 2009, 2008 and 2007 were derived from sales at rates set by or subject to approval by local or state authorities. Annual net operating income increases resulting from ratemaking activity totaling \$54.4 million, \$40.6 million, and \$45.2 million became effective in fiscal 2009, 2008 and 2007 as summarized below:

	Annual Increase (Decrease) to Operating Income For the Fiscal Year Ended September 30					
Rate Action	2009	2008 (In thousands)	2007			
Rate case filings	\$ 2,959	•	\$ 7,793			
GRIP filings	11,443	8,101	25,624			
Annual rate filing mechanisms	38,764	3,275	12,963			
Other rate activity	1,237	1,424	(1,132)			
	\$ 54,403	\$ 40,638	\$ 45,248			

Additionally, the following ratemaking efforts were initiated during fiscal 2009 but had not been completed as of September 30, 2009:

Division	Rate Action	Jurisdiction	Operating Income Requested (In thousands)
Mid-Tex	Rate Case <sup>(1)</sup>	Dallas & Environs	\$ 7,743
Colorado/Kansas	Rate Case	Colorado	3,834
	GSRS <sup>(2)</sup>	Kansas	766
Kentucky/Mid-States	Rate Case <sup>(3)</sup>	Virginia	1,677
•	PRP Surcharge <sup>(4)</sup>	Georgia	909
West Texas	Rate Review Mechanism <sup>(5)</sup>	Lubbock	3,476
	Rate Review Mechanism <sup>(5)</sup>	Amarillo	2,285
Mississippi	Stable Rate Filing	Mississippi	10,195
			\$ 30,885

<sup>(1)</sup> Texas Railroad Commission Examiners issued a proposal for decision (PFD) on October 9, 2009. The PFD recommended a rate change of \$3.5 million applicable to the Dallas and Environs area of the Mid-Tex system. The Company has filed exceptions to the Examiner s proposal. A final Commission decision is expected before the end of the year.

<sup>(2)</sup> Gas System Reliability Surcharge (GSRS) relates to safety related investments made since the previous rate case.

- (3) The Company filed a Rate Case with the state of Virginia requesting a \$1.7 million increase. The staff has recommended an increase of \$1.4 million.
- (4) The Pipeline Replacement Program (PRP) surcharge relates to a long-term cast iron replacement program.
- (5) The Company filed Rate Review Mechanisms with the City of Lubbock requesting an increase of \$3.5 million and with the City of Amarillo requesting an increase of \$2.3 million. Effective October 1, 2009, the respective cities have approved increases of \$2.7 million and \$1.3 million.

In October 2009, we filed rate cases in Georgia and Kentucky, requesting an increase in operating income of \$3.8 million and \$9.5 million.

16

#### **Table of Contents**

Our recent ratemaking activity is discussed in greater detail below.

## Rate Case Filings

A rate case is a formal request from Atmos Energy to a regulatory authority to increase rates that are charged to customers. Rate cases may also be initiated when the regulatory authorities request us to justify our rates. This process is referred to as a show cause action. Adequate rates are intended to provide for recovery of the Company s costs as well as a fair rate of return to our shareholders and ensure that we continue to deliver reliable, reasonably priced natural gas service to our customers. The following table summarizes our recent rate cases:

Division	State	Increas Annu (In	Effective Date	
2009 Rate Case Filings:		Ф	2.512	444.400
Kentucky/Mid-States	Tennessee	\$	2,513	4/1/09
West Texas	Texas		446	Various
Total 2009 Rate Case Filings		\$	2,959	
2008 Rate Case Filings:				
Kentucky/Mid-States	Virginia	\$	869	9/30/08
Kentucky/Mid-States	Georgia		3,351	9/22/08
Mid-Tex <sup>(1)</sup>	Texas		5,430	6/24/08
Colorado-Kansas	Kansas		2,100	5/12/08
Mid-Tex <sup>(2)</sup>	Texas		8,000	4/1/08
Kentucky/Mid-States	Tennessee		8,088	11/4/07
Total 2008 Rate Case Filings		\$	27,838	
2007 Rate Case Filings:				
Kentucky/Mid-States	Kentucky <sup>(3)</sup>	\$	6,200	8/1/07
Mid-Tex	Texas <sup>(4)</sup>		4,793	4/1/07
Kentucky/Mid-States	Missouri <sup>(5)</sup>		1,500	3/4/07
Kentucky/Mid-States	Tennessee		(4,700)	12/15/06
Total 2007 Rate Case Filings		\$	7,793	

<sup>(1)</sup> Increase relates only to the City of Dallas and Environs areas of the Mid-Tex Division.

<sup>(2)</sup> Increase relates only to the Settled Cities area of the Mid-Tex Division.

<sup>(3)</sup> In February 2005, the Attorney General of the State of Kentucky filed a complaint with the Kentucky Public Service Commission (KPSC) alleging that our rates were producing revenues in excess of reasonable levels. In June 2007, the KPSC issued an order dismissing the case. In December 2006, the Company filed a rate

application for an increase in base rates. Additionally, we proposed to implement a process to review our rates annually and to collect the bad debt portion of gas costs directly rather than through the base rate. In July 2007, the KPSC approved a settlement we had reached with the Attorney General for an increase in annual operating income of \$6.2 million effective August 1, 2007.

(4) In March 2007, the RRC issued an order, which increased the Mid-Tex Division s annual operating income by approximately \$4.8 million beginning April 2007 and established a permanent WNA based on 10-year average weather effective for the months of November through April of each year. The RRC also approved a cost allocation method that eliminated a subsidy received from industrial and transportation customers and increased the revenue responsibility for residential and commercial customers. However, the order also required an immediate refund of amounts collected from our 2003 2005 GRIP filings of approximately

17

#### **Table of Contents**

- \$2.9 million and reduced our total return to 7.903 percent from 8.258 percent, based on a capital structure of 48.1 percent equity and 51.9 percent debt with a return on equity of 10 percent.
- (5) The Missouri Commission issued an order in March 2007 approving a settlement with rate design changes, including revenue decoupling through the recovery of all non-gas cost revenues through fixed monthly charges and an estimated increase in operating income of \$1.5 million.

# **GRIP Filings**

As discussed above in Natural Gas Distribution Segment Overview, GRIP allows natural gas utility companies the opportunity to include in their rate base annually approved capital costs incurred in the prior calendar year. The following table summarizes our GRIP filings with effective dates during the fiscal years ended September 30, 2009, 2008 and 2007:

		Incremental Net	Additiona Annual	&nbspB>Equity Reference Range	Per Share
for Anheuser-Busch	Merger Considerat			C	
\$58 \$80	\$	70			
		39			

#### Research Analyst Price Targets

Citi reviewed the reports of nine research analysts found in publicly available equity research. The analysis indicated the following per share reference range for the value of Anheuser-Busch s common stock:

	W	all Street	Resea	arch Price Target	Post	nheuser-Busch -Announcement of ated Financial Plan on	N	r Share Ierger sideration
		InBev nors		Post-InBev Rumors		June 27, 2008(*)		
High	\$	58	\$	65	\$	68	\$	70
Low	\$	51	\$	51	\$	61	\$	70
Median	\$	54	\$	59	\$	65	\$	70

<sup>(\*)</sup> Based on four research analysts who updated their price targets after the announcement of the updated Anheuser-Busch financial plan.

#### Selected Publicly Traded Companies Analysis

Citi reviewed financial and stock market information and public market trading multiples of Anheuser-Busch (based on analysts research for the period ending on May 22, 2008) and the following six companies:

Molson Coors Brewing Company

Grupo Modelo S.A.B. de C.V.

SABMiller plc

InBev N.V./S.A.

Carlsberg A/S

Heineken N.V.

These companies were selected because they are all publicly traded global brewing companies that Citi believed are most comparable to Anheuser-Busch. As part of its selected comparable company analysis, Citi calculated and analyzed the respective ratios of firm value (calculated as equity value plus straight debt, minority interest, straight preferred stock, and out-of-the-money convertibles, less cash and long term equity investments valued at the current market price where available, and at book value where market price is not available) to EBITDA (calculated as earnings before interest, taxes, depreciation and amortization) as estimated for the calendar year 2009. Citi also calculated and analyzed the respective ratio of stock price to estimated earnings per share (P/E) for calendar year 2009. The results of this selected publicly traded companies analysis are summarized below:

Edgar Filing: ATMOS ENERGY CORP - Form 10-K

	Firm Valu	Firm Value / EBITDA		Carnings
	2009E (as of 5/22/08)	2009E (as of 7/10/08)	2009E (as of 5/22/08)	2009E (as of 7/10/08)
High	9.5x	8.8x	16.6x	16.0x
Low	7.2x	6.3x	13.5x	11.0x
Median	8.1x	7.1x	14.2x	12.7x

Estimated financial data of the selected companies were based on research analysts estimates, public filings and other publicly available information, and estimated financial data of Anheuser-Busch was based on internal estimates of Anheuser-Busch s management. Based on the comparable company metrics analyzed, Citi selected a multiple range of 7.5x to 9.5x to the ratio of Anheuser-Busch s firm value to its EBITDA for the calendar year 2009 and 13.5x to 16.5x to Anheuser-Busch s estimated Earning Per Share for calendar year

# **Table of Contents**

2009. This analysis indicated the following selected per share equity reference range for Anheuser-Busch s common stock, as compared to the merger consideration of \$70.00 per share:

	200000	d Per Share quity	<b>:</b>		
	Reference Range for Anheuser-Busch			Per Share Merger Consideration	
2009 Estimated EBITDA	\$	54	\$66	\$	70
2009 Estimated Earning Per Share	\$	53	\$64	\$	70

#### Discounted Cash Flow Analysis

Citi performed a discounted cash flow analysis to calculate the estimated present value of the standalone unleveraged, after-tax free cash flows that Anheuser-Busch could generate for calendar years 2008 through 2018. The analysis was based on internal estimates provided by Anheuser-Busch s management as summarized on page 43. Citi also evaluated management s internal estimates after applying certain sensitivities to take into account the possibility that (i) only seventy-five percent of the cost-savings associated with management s updated financial plan are realized, which is referred to as Sensitivity Case 1, and (ii) only seventy-five percent of the outlined pricing-mix benefits associated with management s updated financial plan are realized, which is referred to as Sensitivity Case 2. Citi estimates for consolidated growth in unleveraged free cash flow for calendar years 2013 through 2018 were prepared after consulting with Anheuser-Busch management, resulting in a range of 3.9% to 1.8%. Estimated terminal values for Anheuser-Busch were calculated by applying a range of perpetuity growth rates for each of Anheuser-Busch s business units, resulting in a consolidated range of 1.6% to 2.2% to Anheuser-Busch s calendar year 2018 estimated unleveraged free cash flow. The cash flows and terminal values were then discounted to present value using discount rates ranging from 7.25% to 7.75%, which range was derived utilizing a weighted average cost of capital analysis based on certain financial metrics, and taking into account market volatility, for Anheuser-Busch and selected global brewing companies. This analysis indicated the following implied per share equity reference ranges for Anheuser-Busch s common stock, as compared to the per share merger consideration of \$70.00 per share:

Implied Per Share	Per Share Merger		
Management Plan	Sensitivity Case 1	Sensitivity Case 2	Consideration
\$69 \$79	\$66 \$75	\$62 \$70	\$70

#### Miscellaneous

Under the terms of Citi s engagement, Anheuser-Busch has agreed to pay Citi for its financial advisory services in connection with the Merger a quarterly fee of \$500,000, during the term of Citi s engagement, not to exceed \$2,000,000, and an additional fee of \$30 million, payable upon the consummation of the Merger. Anheuser-Busch also has agreed to reimburse Citi for reasonable travel and other out-of-pocket expenses incurred by Citi in performing its services, including reasonable fees and expenses of its legal counsel, and to indemnify Citi and related persons against liabilities, including liabilities under the federal securities laws, arising out of its engagement.

Citi and its affiliates in the past have provided services to Anheuser-Busch and InBev unrelated to the proposed Merger, for which services Citi and such affiliates have received compensation, including, without limitation, (i) for

Anheuser-Busch, acted as co-manager with respect to the offering of Anheuser-Busch s \$500,000,000 5.50% Notes due 2018 in November 2007, co-manager with respect to the offering of Anheuser-Busch s \$500,000,000 6.450% Debentures due 2037 in August 2007, bookrunner with respect to the offering of Anheuser-Busch s \$300,000,000 5.60% Notes due 2017 in February 2007, lender in connection with a \$55 million financing commitment and joint document agent on a \$500 million revolving credit facility in February 2008 and (ii) for InBev, in December 2007, provided an opinion with respect to AmBev s acquisition of the remaining stake in Quilmes Industrial SA, which is referred to as Quinsa, in July 2007, acted as bookrunner on AmBev s BRL 300 million senior unsecured notes due 2017 and in July 2006, acted as bookrunner on AmBev s BRL 2 billion local debentures (two tranches) to fund the acquisition of the remaining shares of Quinsa. In the ordinary course of Citi s business, Citi and its affiliates may actively trade or hold the securities of Anheuser-Busch and InBev for its own account or for the account of its customers

41

### **Table of Contents**

and, accordingly, may at any time hold a long or short position in such securities. In addition, Citi and its affiliates (including Citigroup Inc. and its affiliates) may maintain relationships with Anheuser-Busch, and InBev and their respective affiliates.

Anheuser-Busch selected Citi to provide certain financial advisory services in connection with the Merger based on Citi s reputation and experience. Citi is an internationally recognized investment banking firm which regularly engages in the valuation of businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive bids, secondary distributions of listed and unlisted securities, private placements and valuations for estate, corporate and other purposes.

### **Projected Financial Information**

Anheuser-Busch s senior management is wary of making detailed projections for extended periods due to the unpredictability of the underlying assumptions and estimates. However, certain financial forecasts were prepared by our senior management as part of an annual update of the long-term strategic plan in June 2008 and were thereafter made available to Goldman Sachs, Citi and InBev in connection with their consideration of the Merger. There were no material differences between the projections provided to Goldman Sachs and Citi in connection with their evaluation of InBev s \$70 per share proposal (which are summarized on page 43 of this proxy statement) and the projections provided to Goldman Sachs and Citi in connection with their evaluation of InBev s \$65 per share proposal. We have included the material projections in this proxy statement to provide our stockholders access to certain nonpublic information considered by InBev and/or the board of directors for purposes of considering and evaluating the Merger. The inclusion of this information should not be regarded as an indication that any of Goldman Sachs, Citi, InBev or any other recipient of this information considered, or now considers, it to be a reliable prediction of future results.

The projections reflect numerous estimates and assumptions with respect to industry performance, general business, economic, regulatory, market and financial conditions, as well as matters specific to Anheuser-Busch s business, many of which are beyond Anheuser-Busch s control. These projections were, in general, prepared solely for internal use and are subjective in many respects; they are susceptible to interpretations and periodic revision based on actual experience and business developments. As a result, there can be no assurance that the projected results will be realized or that actual results will not be significantly higher or lower than projected. Since the projections cover multiple years, such information by its nature becomes less reliable with each successive year. The projections were not prepared with a view toward public disclosure or toward complying with GAAP, the published guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. Neither Anheuser-Busch s independent registered public accounting firm, nor any other independent accountants, have compiled, examined, or performed any procedures with respect to the prospective financial information contained herein, nor have they expressed any opinion or any other form of assurance on such information or its achievability, and assume no responsibility for, and disclaim any association with, the prospective financial information. The PricewaterhouseCoopers LLP reports incorporated by reference in this proxy statement relate to Anheuser-Busch s historical financial information. They do not extend to the projected financial information and should not be read to do so. Furthermore, the financial projections do not take into account any circumstances or events occurring after the date they were prepared.

Anheuser-Busch has made publicly available its actual results of operations for the quarter and six months ended June 30, 2008. You should review Anheuser-Busch s Quarterly Report on Form 10-Q for the quarter ended June 30, 2008 to obtain this information. See Where You Can Find More Information. Readers of this proxy statement are cautioned not to place undue reliance on the projections set forth below. No one has made or makes any representation to any stockholder regarding the information included in these projections.

The inclusion of projections in this proxy statement should not be regarded as an indication that such projections will be an accurate prediction of future events, and they should not be relied on as such. Except as required by applicable securities laws, Anheuser-Busch does not intend to update, or otherwise revise the material projections to reflect circumstances existing after the date when made or to reflect the occurrence of

42

### **Table of Contents**

future events, even in the event that any or all of the assumptions are shown to be in error. These projections do not give effect to the Merger.

The projections, which assume full implementation of Anheuser-Busch s Blue Ocean program and pricing, are summarized below:

In millions of dollars, except per share amounts	2008	2009	2010	2011	2012
Net Sales	17,721	18,647	19,585	20,250	20,926
Operating Profit(1)	3,162	3,811	3,984	4,273	4,442
Earnings Per Share(1)	3.13	3.90	4.39	4.90	5.34
Free Cash Flow(2)	2,177	2,608	2,734	2,951	2,946
EBITDA without Equity Income(3)	4,200	4,800	5,000	5,300	5,400
Equity Income, Net of Tax	642	699	779	850	931

- (1) Results are normalized to exclude one-time gains, losses and expenses including the anticipated \$300 million to \$400 million (equal to 25 to 33 cents in earnings per share) associated with the Blue Ocean program.
- (2) Defined as operating cash flow, net of changes in working capital, minus capital expenditures.
- (3) EBITDA is calculated as pretax income plus depreciation and amortization expense, plus net interest cost (interest expense less capitalized interest). EBITDA was rounded as above in projections shown to Anheuser-Busch s board of directors and to InBev.

### **Financing of the Merger**

InBev and Merger Sub have estimated that the total amount of funds necessary to consummate the Merger and related transactions will be approximately \$54.8 billion, including (i) the financing of the Merger, including payment of related transaction charges, fees and expenses, and (ii) certain fees and expenses and accrued but unpaid interest on Anheuser-Busch s outstanding indebtedness. InBev has said it intends to fund the transaction with new credit facilities and equity financing. Funding of the equity and debt financing is subject to the satisfaction of the conditions set forth in the facilities under which the financing will be provided. Consummation of the Merger is not conditioned upon InBev s obtaining financing.

In connection with an anticipated transaction with Anheuser-Busch, InBev entered into the following definitive financing arrangements, copies of which have been provided to us:

US\$45,000,000,000 Senior Facilities Agreement, dated as of July 12, 2008 as amended as of July 23, 2008 for InBev and InBev Worldwide S.A R.L., arranged by Banco Santander, S.A., Barclays Capital, BNP Paribas, Deutsche Bank AG, London Branch, Fortis Bank SA/NV, ING Bank N.V., J.P. Morgan PLC, Mizuho Corporate Bank, LTD., the Bank of Tokyo-Mitsubishi UFJ, LTD. and The Royal Bank of Scotland PLC, as Mandated Lead Arrangers and Bookrunners, and Fortis Bank SA/NV, acting as Agent and Issuing Bank.

US\$5,600,000,000 Bridge Facility Agreement, dated as of July 12, 2008 as amended and increased to US\$9,800,000,000 as of July 23, 2008 for InBev, arranged by Banco Santander, S.A., BNP Paribas, Deutsche Bank AG, London Branch, Fortis Bank SA/NV, ING Bank N.V., J.P. Morgan PLC and The Royal Bank of Scotland PLC, as Mandated Lead Arrangers and Fortis Bank SA/NV, acting as Agent.

The availability of the debt financing is subject only to (a) satisfaction of customary corporate, acquisition-related and finance documentary conditions precedent; (b) there being no change of control of InBev; (c) in respect of each individual lender, there being no illegality affecting that lender s obligations to lend; (d) there being no continuing major event of default or breach of a major representation; (e) consummation of the Merger; and (f) InBev receiving investment grade credit ratings from S&P and Moody s. Major events of default are limited to non payment; breach of merger restrictions; fundamental change in the nature of business conducted; breach of restrictions on the granting of security; breach of certain obligations relating to the terms of the acquisition; insolvency related events; unlawfulness of the obligor s obligations or

43

### **Table of Contents**

repudiation of the finance documents. Major representations are limited to those relating to legal status; power to carry on operations; binding nature of obligations under finance documents; non conflict of obligation under finance documents with law or other agreements; power and authority to enter into finance documents; validity and admissibility in evidence of finance documents and pari passu ranking of obligations under the finance documents. Furthermore, major events of default and major representations relate only to InBev, Merger Sub, and InBev Worldwide S.A R.L. and material subsidiaries of InBev, excluding Anheuser-Busch or any of its subsidiaries.

In accordance with the Merger Agreement, InBev has agreed to satisfy on a timely basis all conditions applicable to InBev under the senior debt and bridge facility agreements and must obtain and consummate the financing contemplated by the senior debt and bridge facility agreements and if such financing is unavailable, obtain and consummate alternative financing.

InBev must give Anheuser-Busch prompt notice upon becoming aware of any termination of the senior debt and bridge facility agreements and of the occurrence of any event that would prevent InBev from satisfying one or more of the conditions to initial utilization in the senior debt and bridge facility agreements. In the event that InBev becomes aware of any event or circumstance that makes any portion of the financing of the senior debt and bridge facility agreements unavailable prior to the Termination Date (as described below) on the terms and conditions set forth in the senior debt and bridge facility agreements, InBev must secure as promptly as practicable any such unavailable portion from alternative sources.

Each of InBev and Merger Sub have expressly acknowledged and agreed that the closing and the consummation of the Merger is not conditioned on the availability of the funding under the senior debt and bridge facility agreements or any alternative financing arrangement.

### Interests of Anheuser-Busch s Directors and Executive Officers in the Merger

In considering the recommendation of our board of directors with respect to the Merger Agreement, holders of shares of Anheuser-Busch common stock should be aware that Anheuser-Busch executive officers and directors have certain interests in the Merger that may be different from, or in addition to, those of our stockholders generally. These interests may create potential conflicts of interest. Our board of directors was aware that these interests existed when it approved the Merger Agreement. Such interests relate to, or arise from, among other things:

the fact that InBev and Mr. Busch IV contemplate entering into a consulting agreement under which Mr. Busch will provide consulting services in exchange for the payments and benefits described below;

the fact that unvested equity awards, including stock options, restricted stock and deferred stock units held by Anheuser-Busch s directors and executive officers, will become vested upon approval by stockholders of the Merger Agreement or upon consummation of the Merger, as applicable, and all these equity awards will be converted into the right to receive an amount in cash pursuant to the terms of the Merger Agreement (minus the exercise price, if any);

the fact that upon the adoption by stockholders of the Merger Agreement or upon consummation of the Merger, as applicable, (i) executive officers will be entitled to pro-rata bonuses for the current performance period under the Officer Bonus Plan and (ii) executive officers and directors will be entitled to accelerated vesting and payment of their accrued benefits and distribution of their accounts under Anheuser-Busch supplemental retirement plans and deferred compensation plans;

the fact that certain executive officers will be entitled to integration bonuses and/or enhanced severance payments and benefits in the event of an involuntary or constructive termination within two years following the

closing of the Merger and, in certain circumstances, a modified gross-up payment on amounts that are subject to the excise tax imposed by Code Section 4999;

the fact that InBev has agreed, after consultation with the Anheuser-Busch board of directors, to nominate and cause to be elected two current or former directors of Anheuser-Busch (one of whom will

44

### **Table of Contents**

be August Busch IV) to the board of directors of InBev, and to confirm each such director for a three-year term at the first annual general meeting of InBev following the closing of the Merger; and

the fact that InBev has agreed to provide Anheuser-Busch s directors and officers with certain rights to indemnification and insurance.

Anheuser-Busch s directors and executive officers have entered into or participate in, as applicable, the various agreements and arrangements discussed below. In the case of each plan or agreement discussed below to which the term change in control applies, adoption by stockholders of the Merger Agreement or consummation of the Merger, as applicable, will constitute a change in control for all purposes under the applicable Anheuser-Busch Benefit Plans (as that term is defined in the Merger Agreement). InBev may also seek to enter into other arrangements with one or more of Anheuser-Busch s executive officers regarding their ongoing employment with Anheuser-Busch.

### Arrangements with Executive Officers and Directors of Anheuser-Busch

Consulting Agreement with Mr. Busch IV . In connection with the Merger, InBev and Mr. Busch IV are currently negotiating the terms of a consulting agreement (the <u>Consulting Agreement</u>) to be effective as of the closing of the Merger and continuing until December 31, 2013 (the <u>Consulting Period</u>), substantially on the terms described below. In his role as consultant, it is contemplated that Mr. Busch IV will, at the request of the CEO of InBev, provide advice to InBev on Anheuser-Busch new products and new business opportunities; review Anheuser-Busch marketing programs; meet with retailers, wholesalers and key advertisers of Anheuser-Busch; attend North American media events; provide advice with respect to Anheuser-Busch s relationship with charitable organizations and the communities in which it operates; and provide advice on the taste, profile, and characteristics of the Anheuser-Busch malt-beverage products.

Under the terms of the Consulting Agreement, as contemplated, at the time of the Merger, Mr. Busch IV will receive a lump sum cash payment equal to \$10,350,000, less any applicable withholding. During the Consulting Period, it is contemplated that Mr. Busch IV will be paid a fee of approximately \$120,000 per month. In addition, it is contemplated that Mr. Busch IV will be provided with an appropriate office in St. Louis, Missouri, administrative support, and insured medical, dental, vision and prescription drug benefits that are materially similar to those provided to full-time salaried employees of Anheuser-Busch. It is contemplated that he will also be provided with personal security services through December 31, 2011 (in St. Louis, Missouri) in accordance with the Company s past practices including an income tax gross-up and with complimentary tickets to Anheuser-Busch-sponsored events. Mr. Busch IV will also be eligible for a Code Section 280G gross-up payment (estimated to be approximately \$13.3 million) on various change in control payments and benefits to which he is entitled to in connection with the Merger.

It is contemplated that Mr. Busch IV will be subject to restrictive covenants relating to non-competition and non-solicitation of employees and customers which will be in effect for the Consulting Period and a confidentiality covenant. It is also contemplated that the parties will be subject to a mutual non-disparagement covenant.

It is contemplated that the Consulting Agreement will be terminable by either party on sixty days written notice. If terminated by reason of a notice given by Mr. Busch IV, he would no longer be entitled to any rights, payments or benefits under the Consulting Agreement (with the exception of accrued but unpaid consulting fees, business expense reimbursements, any Code Section 280G gross-up payment, indemnification by InBev, and continued office and administrative support for 90 days following termination of the agreement) and the non-compete and non-solicitation restrictive covenants would survive for two years following termination of the Consulting Agreement (but not beyond December 31, 2013). If terminated by reason of a notice given by InBev for any reason other than for cause, Mr. Busch IV would continue to have all rights (including the right to payments and benefits) provided for in the Consulting Agreement and would continue to be bound by the non-compete and non-solicitation restrictive covenants

through December 31, 2013.

It is contemplated that Mr. Busch IV would generally be indemnified by InBev from and against all claims arising from the performance of his duties as a consultant during the Consulting Period. In addition, it

45

### **Table of Contents**

is contemplated that Mr. Busch IV and InBev will execute a mutual release of claims regarding all pre-closing matters.

<u>Retention Plan</u>. In connection with the transaction, Anheuser-Busch and InBev have agreed to establish an employee retention program providing for integration bonuses and severance benefits for certain key employees of Anheuser-Busch. This program would be effective as of the closing of the Merger and is summarized below.

Integration Bonus. Approximately 60 key employees of Anheuser-Busch (including Anheuser-Busch s current executive officers, other than Mr. Busch IV who will terminate employment with Anheuser-Busch upon the occurrence of the Merger as described above) will be eligible to receive an additional bonus for 2008 equal to 40% of the target bonus otherwise payable to that employee under current Anheuser-Busch bonus programs. The amount of the additional bonus will depend upon the extent of achievement of projected 2008 savings under the Blue Ocean program. In addition, approximately 360 key employees of Anheuser-Busch (including Anheuser-Busch s current executive officers, other than Mr. Busch IV) will be eligible to receive an additional bonus ranging from 80% to 110% of the employee s 2009 target bonus award. The extent to which the bonus exceeds 80% of the employee s 2009 target bonus will depend on the extent of achievement of the Blue Ocean operating goals through 2009. In order to qualify for this integration bonus, the employee generally must be employed until the date on which annual bonuses for 2009 are paid in the ordinary course, though an employee who is involuntarily or constructively terminated after the closing and prior to the bonus payment will be eligible to receive a pro-rata payment.

The executive officers would be entitled to the following estimated integration bonuses under the retention plan assuming (1) the executive officer continues employment through the applicable bonus payment date and (2) in the case of the 2009 Integration Bonus, each officer s 2009 target bonus is equal to his/her 2008 target bonus.

	2009 Integration Bonus (assuming continued employment and Blue Ocean performance targets are not achieved) 2009 Integration Bonus (assuming continue employment and maximum Blue Ocean performance targets are achieved)		(assur emp ma	Bonus ming continued ployment and ximum Blue Ocean	Bonus (assuming continued employment and maximum Blue Ocean performance targets	
Name			e achieved)	aı	re achieved)	
W. R. Baker	\$	542,430	\$	678,038	\$	271,215
D. J. Muhleman	\$	456,237	\$	570,297	\$	228,819
M. J. Owens	\$	398,520	\$	498,150	\$	199,260
Remaining Executive Officers as a group (approx. 13 executives)	\$	3,375,833	\$	4,219,791	\$	1,687,917

Enhanced Severance. The retention plan also provides that the same group of approximately 360 employees (including Anheuser-Busch's current executive officers, other than Mr. Busch IV) will be eligible for enhanced severance benefits payable upon an involuntary or constructive termination of employment within two years following the closing of the transaction. These severance benefits will range from 15 months of base salary to 2 times the sum of base salary and target bonus, and will include continuation of medical, dental, vision, life insurance, prescription drug plan coverage and other welfare benefits ranging from 15 to 24 months (in each case, depending on the particular employee category; executive officers will be eligible for severance ranging from 1.75 to 2 times the sum of base salary and target bonus and benefits ranging from 21 to 24 months). The retention plan provides that approximately 60

of these employees (including Anheuser-Busch's current executive officers, other than Mr. Busch IV) will, if necessary, be eligible for a modified gross-up payment on amounts that are subject to the excise tax imposed by Code Section 4999, but only if the total value of all parachute payments to the individual exceeds 110% of the individual s safe harbor amount. The enhanced severance program will also contain customary restrictive obligations, including an agreement not to compete with Anheuser-Busch for a period ranging from 12 to 24 months. Constructive termination is defined to include a material reduction of compensation, a reduction in duties and responsibilities from those in effect immediately prior to closing of the transaction and relocation of more than 50 miles.

46

### **Table of Contents**

The executive officers would be entitled to the following estimated payments (assuming termination of employment immediately following the closing of the Merger) pursuant to the enhanced severance portion of the retention plan.

Named Executive			Esti	mated Section 280G
	Severance			
Officers	Payment	<b>Benefit Continuation</b>	Gros	ss-Up Payment
W. R. Baker	\$2,712,152	24 months	\$	0
D. J. Muhleman	\$2,407,920	24 months	\$	2,200,000
M. J. Owens	\$2,103,300	24 months	\$	2,000,000
Remaining Executive Officers as				
a group (approx. 13 executives)	\$19,630,604	Ranges from 21 to 24 months	\$	12,200,000

### Accelerated Payments under Anheuser-Busch Benefit Plans

Officer Bonus Plan. Pursuant to the terms of the Officer Bonus Plan, upon the occurrence of a change in control, all bonuses for the current performance period will be immediately payable in cash on a pro-rated basis. Under the 2008 officer bonus program, bonus payments will be calculated as percentages of Anheuser-Busch s adjusted pretax earnings for the period prior to the Merger. The estimate of bonus payments set forth in the table below assumes that the Merger is consummated on November 1, 2008 and is based upon Anheuser-Busch s projection of its pretax earnings for the ten-month period ending October 31, 2008. As required by the bonus plan, pretax earnings have been adjusted to normalize earnings and therefore the calculation has excluded expenses associated with Anheuser-Busch s enhanced retirement program, investment banking, legal and other expenses relating to the Merger and the potential alternative transactions with a third party and expenses under executive and director compensation and retirement programs arising as a result of the Merger.

Supplemental Retirement Plans. Executive officers participate in the Anheuser-Busch 401(k) Restoration Plan (the Restoration Plan ) and the Anheuser-Busch Companies, Inc. Supplemental Executive Retirement Plan (the SERP ). Upon a change in control of Anheuser-Busch, the entire value of accounts maintained under the Restoration Plan will be distributed to participating executives in a single lump sum within thirty days of the date of the change in control. In addition, the actuarial equivalent of benefits accrued under the SERP will fully vest and will be paid in a single lump sum within thirty days of the date of the change in control. With respect to the SERP, if an executive has otherwise satisfied the eligibility requirements for early or normal retirement benefits on the date of the change in control, the amount payable to such executive is the amount he or she would have been paid in the event of actual retirement on the change in control date.

Deferred Compensation Plans. Executive officers of Anheuser-Busch are eligible to participate in the Anheuser-Busch Executive Deferred Compensation Plan (the Executive Deferred Compensation Plan ) and non-employee directors of Anheuser-Busch are eligible to participate in the Anheuser-Busch Companies, Inc. Deferred Compensation Plan for Non-Employee Directors (the Director Deferred Compensation Plan ). The Executive Deferred Compensation Plan permits officers to defer salary and bonus compensation in excess of \$200,000 for a specified period of time or until termination of employment. The Director Deferred Compensation Plan permits non-employee directors to defer compensation into cash accounts and/or share accounts. In the event of a change in control of Anheuser-Busch, the entire amount of participating officers and directors accounts will be paid to such individuals in a single lump sum within 30 days of the date of the change in control.

Excise Tax Gross Ups. In the event that an excise tax or other special tax is imposed on any payment made to an executive officer under the Officer Bonus Plan, the Executive Deferred Compensation Plan, the SERP, or the 401(k) Restoration Plan due to a change in control, the payment amount will be increased to cover such tax on a grossed-up basis. Such gross-up payments are included in, and are not additional to, the gross-up payments provided in the *Enhanced Severance* chart above.

The table below summarizes the value of the estimated payments that would be received by Anheuser-Busch s executive officers and directors upon adoption by stockholders of the Merger Agreement or

47

### **Table of Contents**

consummation of the Merger, as applicable, assuming stockholder approval and consummation of the Merger occur on November 1, 2008 (other than the amounts calculated under the 401(k) Restoration Plan and the Deferred Compensation Plan which are calculated as of June 30, 2008 and may change prior to adoption by stockholders of the Merger Agreement).

		o Rata 2008		Value of Accrued		Volue of		Volue of	
	Bonus under the			Benefits under the		Value of Accounts under the		Value of Accounts	
	Of	fficer Bonus	us SERP (Vested/Unvested)		401(k) Restoration Plan(1)		under Deferred Compensation Plan(2)		
Name		Plan							
Executives									
A. A. Busch IV	\$	6,000,000	\$	0/\$5,638,214	\$	1,004,889	\$	0	
W.R. Baker	\$	2,439,000	\$	6,991,782/\$0	\$	846,164	\$	9,472,840	
D. J. Muhleman	\$	2,045,000	\$	0/\$3,190,641	\$	453,437	\$	0	
M. J. Owens	\$	1,782,000	\$	0/\$2,381,538	\$	331,836	\$	0	
M. T. Bobak(3)		N/A		N/A	\$	328,228	\$	250,550	
Remaining Executive									
Officers as a group									
(approx. 13 executives)	\$	14,633,000	\$	14,663,874/\$5,133,795	\$	2,278,838	\$	4,542,725	
Directors									
August A. Busch III		N/A		N/A		N/A	\$	153,727	
Carlos Fernandez(4)		N/A		N/A		N/A	\$	14,579	
James J. Forese		N/A		N/A		N/A	\$	0	
John E. Jacob(5)		N/A		N/A		N/A	\$	1,038,205	
James R. Jones		N/A		N/A		N/A	\$	119,192	
Charles F. Knight(6)		N/A		N/A		N/A		N/A	
Vernon R. Loucks, Jr.		N/A		N/A		N/A	\$	329,270	
Vilma S. Martinez		N/A		N/A		N/A	\$	3,422,109	
William Porter Payne		N/A		N/A		N/A	\$	274,794	
Joyce M. Roché		N/A		N/A		N/A	\$	600,546	
Henry Hugh Shelton		N/A		N/A		N/A	\$	110,003	
Patrick T. Stokes		N/A		N/A		N/A	\$	160,795	
Andrew C. Taylor		N/A		N/A		N/A	\$	155,710	
Douglas A. Warner III		N/A		N/A		N/A	\$	201,575	
Edward E. Whitacre, Jr.		N/A		N/A		N/A	\$	2,140,266	

<sup>(1)</sup> The executives account balances under the 401(k) Restoration Plan are fully vested and therefore no acceleration of vesting will occur as a result of the Merger.

(3) Mr. Bobak resigned from Anheuser-Busch in December 2007.

<sup>(2)</sup> The executives and directors account balances under the Deferred Compensation Plan are fully vested and therefore no acceleration of vesting will occur as a result of the Merger.

- (4) Mr. Fernandez resigned from the board of directors in June 2008.
- (5) Mr. Jacob resigned from the board of directors in April 2008.
- (6) Mr. Knight resigned from the board of directors in April 2008.

## Treatment of Outstanding Equity Awards

Acceleration of Equity Awards. Pursuant to the terms of Anheuser-Busch s 2008 Long-Term Equity Incentive Plan for Non-Employee Directors, 2007 Equity and Incentive Plan, 2006 Restricted Stock Plan for Non-Employee Directors, 1998 Incentive Stock Plan, 1989 Incentive Stock Plan, Global Employee Stock Purchase Plan, the Stock Plan for Non-Employee Directors, and the Non-Employee Director Elective Stock

48

### **Table of Contents**

Acquisition Plan (the Stock Plans ), upon adoption by stockholders of the Merger Agreement or upon consummation of the Merger, as applicable, each equity award (including options, restricted shares, and restricted share units) outstanding under the Stock Plans will vest and all restrictions will lapse (including equity awards held by our executive officers and directors).

Cancellation of Equity Awards under the Merger Agreement. The Merger Agreement provides that at the consummation of the Merger, each outstanding option to purchase shares under the Stock Plans will be cancelled and converted into the right to receive an amount in cash equal to the product of (x) the total number of shares subject to the option times (y) the excess of \$70.00 over the exercise price per share under the option, less applicable taxes required to be withheld.

In addition, the Merger Agreement provides that at the consummation of the Merger, all other equity outstanding awards will be cancelled and converted into the right to receive an amount in cash equal to (x) the number of shares subject to the awards times (y) \$70.00, less applicable taxes required to be withheld.

Assuming a closing date of November 1, 2008, the estimated aggregate number and value of equity awards held by executive officers and directors is as follows:

	Unvested Opti Accelerate U <sub>l</sub> in	pon a Change	Rest Stock and D Ui That Will	I Shares of cricted deferred Stock nits Accelerate don a	Aggregate Value of Equity Awards Payable Upon Consummation of Merger (Vested and			
Name	Con (#/S		Change in Control (#/\$)*			Unvested) (\$)*		
Executive Officers								
A. A. Busch IV(1)	1,389,060/\$	28,867,668	67,847/\$	4,749,290	\$	88,638,310		
W.R. Baker	311,166/\$	7,968,908	20,306/\$	1,421,420	\$	51,474,383		
D. J. Muhleman	331,326/\$	6,865,573	16,815/\$	1,177,050	\$	38,454,580		
M. J. Owens	293,127/\$	6,061,254	14,628/\$	1,023,960	\$	26,559,998		
M. T. Bobak(2)	371,462/\$	7,716,192	18,585/\$	1,300,950	\$	27,322,957		
Remaining Executive								
Officers (approx. 13								
executives)	1,905,389/\$	39,633,757	88,554/\$	6,198,780	\$	193,287,387		
Directors								
August A. Busch III(3)	3,333/\$	68,543	9,275/\$	649,250	\$	103,594,266		
Carlos Fernandez(4)		0		0	\$	510,007		
James J. Forese	4,999/\$	110,443	2,913/\$	203,910	\$	741,935		
John E. Jacob(5)		0	859/\$	60,130	\$	28,286,916		
James R. Jones	4,999/\$	110,443	2,925/\$	204,750	\$	1,251,100		
Charles F. Knight(6)		0		0	\$	1,046,350		
Vernon R. Loucks, Jr.	4,999/\$	110,443	2,925/\$	204,750	\$	1,251,100		
Vilma S. Martinez	4,999/\$	110,443	2,913/\$	203,910	\$	1,250,260		
William Porter Payne	4,999/\$	110,443	2,925/\$	204,750	\$	1,251,100		
Joyce M. Roché	4,999/\$	110,443	2,925/\$	204,750	\$	1,251,100		

Henry Hugh Shelton	4,999/\$	110,443	2,925/\$	204,750	\$ 960,475
Patrick T. Stokes	3,333/\$	68,543	15,872/\$	1,111,040	\$ 140,766,897
Andrew C. Taylor	4,999/\$	110,443	2,913/\$	203,910	\$ 1,250,260
Douglas A. Warner III	4,999/\$	110,443	2,925/\$	204,750	\$ 1,251,100
Edward E. Whitacre, Jr.	4,999/\$	110,443	2,925/\$	204,750	\$ 825,200

<sup>\*</sup> The value of the stock option awards is equal to (a) the product of the number of shares underlying the award and \$70 minus (b) the aggregate exercise price. The value of the restricted stock awards and deferred stock unit awards is equal to the product of the number of shares subject to the award and \$70.

49

<sup>(1)</sup> Amount does not include 50,000 vested options transferred to Mr. Busch IV by Mr. Busch III.

## **Table of Contents**

- (2) Mr. Bobak resigned from Anheuser-Busch in December 2007.
- (3) Amount includes approximately 100,000 transferred, vested options (50,000 of which were transferred to Mr. Busch IV).
- (4) Mr. Fernandez resigned from the board of directors in June 2008.
- (5) Mr. Jacob resigned from the board of directors in April 2008.
- (6) Mr. Knight resigned from the board of directors in April 2008.

### Indemnification and Insurance

Prior to the effective time of the Merger, Anheuser-Busch will, and if Anheuser-Busch is unable to, InBev will cause the surviving corporation to, obtain and fully pay for tail insurance policies with a claims period of at least six years from and after the effective time of the Merger from an insurance carrier with the same or better credit rating as Anheuser-Busch s current insurance carrier with respect to directors and officers liability insurance and fiduciary liability insurance (collectively, D&O Insurance ), for the persons who, as of the date of the Merger Agreement, were covered by Anheuser-Busch s existing D&O Insurance, with terms, conditions, retentions and levels of coverage at least as favorable as Anheuser-Busch s existing D&O Insurance with respect to matters existing or occurring at or prior to the effective time of the Merger (including in connection with the Merger Agreement or the transactions or actions contemplated thereby). If Anheuser-Busch and the surviving corporation for any reason fail to obtain such tail insurance policies, InBev has agreed to cause the surviving corporation to continue to maintain in effect Anheuser-Busch s current D&O Insurance, at no expense to the beneficiaries, for a period of at least six years from and after the effective time of the Merger. If such insurance is unavailable, InBev has agreed to cause the surviving corporation to purchase the best available D&O Insurance for such six-year period from an insurance carrier with the same or better credit rating as Anheuser-Busch s current insurance carrier with respect to Anheuser-Busch s existing D&O Insurance with terms, conditions, retentions and with levels of coverage at least as favorable as provided in Anheuser-Busch s existing policies as of the date of the Merger Agreement. In no event will InBev or the surviving corporation be required to expend an annual premium amount in excess of 300% of the annual premiums currently paid by Anheuser-Busch for such insurance; if the annual premiums of such insurance coverage exceed that amount, the surviving corporation is required to obtain a policy with the greatest coverage available for a cost not exceeding such amount.

Following the effective time of the Merger, InBev and the surviving corporation will indemnify, defend and hold harmless each present and former director and officer of Anheuser-Busch or any of its subsidiaries and any fiduciary under any Anheuser-Busch benefit plan (the indemnified parties), and will also promptly advance expenses as incurred, against any costs or expenses (including attorneys fees and disbursements), judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the indemnified party is or was an officer, director, employee or fiduciary of Anheuser-Busch or any of its subsidiaries or a fiduciary under any Anheuser-Busch benefit plan, whether asserted or claimed prior to, at or after the effective time of the Merger (including with respect to any acts or omissions in connection with the Merger Agreement and the transactions and actions contemplated thereby), to the fullest extent permitted under Delaware law and our certificate of incorporation or bylaws and any indemnification agreement in effect on the date of the Merger Agreement. Any determination required to be made with respect to whether an officer s or director s conduct complied with the standards set forth under Delaware law and our certificate of incorporation and bylaws will be made by independent counsel selected by the indemnified party. In the event of any claim, action, suit, proceeding or investigation, (x) neither InBev nor the

surviving corporation may settle, compromise or consent to the entry of any judgment in any claim, action, suit, proceeding or investigation, unless such settlement, compromise or consent includes an unconditional release of the indemnified party from all liability arising out of such claim, action, suit, proceeding or investigation or the indemnified party otherwise consents, and (y) the surviving corporation will cooperate in the defense of such matter.

50

### **Table of Contents**

Additionally, the charter and bylaws of the surviving corporation must contain provisions no less favorable with respect to indemnification of and advancement of expenses to individuals who were directors and officers prior to the effective time of the Merger than those contained in our certificate of incorporation and bylaws as of the date of the Merger Agreement, and such provisions may not be amended, repealed or otherwise modified for a period of six years from the effective time of the Merger in any manner that would adversely affect the rights of any indemnified party.

The rights of the indemnified parties under the Merger Agreement are in addition to any rights such indemnified parties may have under the certificate of incorporation or bylaws of Anheuser-Busch or any of its subsidiaries, or under any applicable contracts or laws. InBev has agreed to honor and perform under all indemnification agreements entered into by Anheuser-Busch or any of its subsidiaries.

Each of Anheuser-Busch s directors and executive officers is party to an indemnification agreement with Anheuser-Busch which provides indemnitees with, among other things, certain indemnification and advancement rights in third-party proceedings, proceedings by or in the right of Anheuser-Busch, proceedings in which the indemnitee is wholly or partly successful, and for an indemnitee s expenses incurred as a witness in a proceeding by reason of his or her corporate status. In the event of a potential change of control of Anheuser-Busch, each of the directors and executive officers has the right to request that Anheuser-Busch fund a trust in an amount sufficient to satisfy any and all expenses reasonably anticipated at the time of request to be incurred in connection with investigating, preparing for and defending any claim relating to an indemnifiable event, and any and all judgments, fines, penalties and settlement amounts of any and all claims relating to an indemnifiable event from time to time actually paid or claimed, reasonably anticipated or proposed to be paid.

## Material U.S. Federal Income Tax Consequences of the Merger to Our Stockholders

The following discussion is a summary of the material U.S. federal income tax consequences of the Merger to U.S. holders (as defined below) of Anheuser-Busch common stock whose shares are converted into the right to receive cash in the Merger. This summary is based on the provisions of the Internal Revenue Code of 1986, as amended (the Code), U.S. Treasury regulations promulgated thereunder, judicial authorities and administrative rulings, all as in effect as of the date of the proxy statement and all of which are subject to change, possibly with retroactive effect.

For purposes of this discussion, the term U.S. holder means a beneficial owner of shares of our common stock that is, for U.S. federal income tax purposes:

an individual who is a citizen or resident of the United States;

a corporation (including any entity treated as a corporation for U.S. federal income tax purposes) created or organized under the laws of the United States, any state thereof, or the District of Columbia;

a trust that (i) is subject to the supervision of a court within the United States and the control of one or more U.S. persons or (ii) has a valid election in effect under applicable U.S. Treasury regulations to be treated as a U.S. person; or

an estate the income of which is subject to U.S. federal income tax regardless of its source.

Holders of our common stock who are not U.S. holders may be subject to different tax consequences than those described below and are urged to consult their tax advisors regarding their tax treatment under U.S. and non-U.S. tax laws.

The following does not purport to consider all aspects of U.S. federal income taxation of the Merger that might be relevant to U.S. holders in light of their particular circumstances, or those U.S. holders that may be subject to special rules (for example, dealers in securities or currencies, brokers, banks, financial institutions, insurance companies, mutual funds, tax-exempt organizations, stockholders subject to the alternative minimum tax, partnerships (or other flow-through entities and their partners or members), persons whose functional currency is not the U.S. dollar, stockholders who hold our stock as part of a hedge, straddle, constructive sale or conversion transaction or other integrated investment, stockholders that acquired our common stock pursuant to the exercise of an employee stock option or otherwise as compensation, or U.S. holders who have perfected

51

### **Table of Contents**

and not withdrawn a demand for statutory appraisal rights), nor does it address the U.S. federal income tax consequences to U.S. holders that do not hold our common stock as capital assets within the meaning of Section 1221 of the Code (generally, property held for investment). In addition, the discussion does not address any aspect of foreign, state, local, estate, gift or other tax law that may be applicable to a U.S. holder.

The tax consequences to stockholders that hold our common stock through a partnership or other pass-through entity generally will depend on the status of the stockholder and the activities of the partnership. Partners in a partnership or other pass-through entity holding our common stock should consult their tax advisors.

This summary of certain material U.S. federal income tax consequences is for general information only and is not tax advice. Holders are urged to consult their tax advisors with respect to the application of U.S. federal income tax laws to their particular situations as well as any tax consequences arising under the U.S. federal estate or gift tax rules, or under the laws of any state, local, foreign or other taxing jurisdiction or under any applicable tax treaty.

Exchange of Shares of Common Stock for Cash Pursuant to the Merger Agreement. The receipt of cash in exchange for shares of our common stock pursuant to the Merger will be a taxable transaction for U.S. federal income tax purposes. In general, a U.S. holder whose shares of common stock are converted into the right to receive cash in the Merger will recognize capital gain or loss for U.S. federal income tax purposes in an amount equal to the difference, if any, between the amount of cash received with respect to such shares and the stockholder s adjusted tax basis in such shares. Gain or loss will be determined separately for each block of shares (i.e., shares acquired at the same cost in a single transaction) surrendered for cash pursuant to the Merger. Such gain or loss will be long-term capital gain or loss provided that a stockholder s holding period for such shares is more than 12 months at the time of the consummation of the Merger. Long-term capital gains of non-corporate U.S. holders, including individuals, recognized in taxable years beginning before January 1, 2011 are generally taxed at a maximum rate of 15%. Short-term capital gains are taxed at ordinary income rates.

Backup Withholding and Information Reporting. A U.S. holder may be subject to backup withholding at the applicable rate (currently 28 percent) on the cash payments to which such U.S. holder is entitled pursuant to the Merger, unless the U.S. holder properly establishes an exemption or provides a taxpayer identification number and otherwise complies with the backup withholding rules. Each U.S. holder should complete and sign the substitute IRS Form W-9 included as part of the letter of transmittal and return it to the paying agent in order to provide the information and certification necessary to avoid backup withholding, unless an applicable exemption applies and is established in a manner satisfactory to the paying agent. Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules generally will be allowable as a refund or a credit against a U.S. holder s U.S. federal income tax liability provided the required information is timely furnished to the IRS.

### **Regulatory Approvals**

Under the HSR Act and the rules promulgated thereunder by the FTC, the Merger cannot be completed until Anheuser-Busch and InBev each file a notification and report form under the HSR Act and the applicable waiting period has expired or been terminated. On July 15, 2008, InBev filed, and on July 18, 2008, Anheuser-Busch filed notification and report forms under the HSR Act with the FTC and the Antitrust Division of the DOJ. The applicable waiting period under the HSR Act has not yet expired or been terminated. At any time before or after consummation of the Merger, notwithstanding the termination of the waiting period under the HSR Act, the Antitrust Division of the DOJ or the FTC could take such action under the antitrust laws as it deems necessary or desirable in the public interest, including seeking to enjoin the consummation of the Merger or seeking divestiture of substantial assets of Anheuser-Busch or InBev. At any time before or after the consummation of the Merger, and notwithstanding the termination of the waiting period under the HSR Act, any state could take such action under antitrust laws as it deems

necessary or desirable in the public interest. Such action could include seeking to enjoin the consummation of the Merger or seeking divestiture of

52

## **Table of Contents**

substantial assets of Anheuser-Busch or InBev. Private parties may also seek to take legal action under antitrust laws under certain circumstances.

In addition, the Merger is subject to various foreign antitrust laws. To the extent required, Anheuser-Busch and InBev expect to file notifications in certain foreign jurisdictions, and to observe the applicable waiting periods prior to completing the Merger.

While there can be no assurance that the Merger will not be challenged by a governmental authority or private party on antitrust grounds, Anheuser-Busch, based on a review of information provided by InBev relating to the businesses in which it and its affiliates are engaged, believes that the Merger can be effected in compliance with federal, state and foreign antitrust laws. The term antitrust laws means the Sherman Act, as amended, the Clayton Act, as amended, the HSR Act, the Federal Trade Commission Act, as amended, and all other Federal, state and foreign statutes, rules, regulations, orders, decrees, administrative and judicial doctrines, and other laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade.

### **Delisting and Deregistration of Common Stock**

If the Merger is completed, our common stock will be delisted from the NYSE and deregistered under the Exchange Act and we will no longer file periodic reports with the SEC on account of our common stock.

### Litigation Related to the Merger

#### Missouri Shareholder Suits

On June 4, 2008 and July 16, 2008, two substantially similar putative shareholder class and derivative actions were filed in the Circuit Court of the City of St. Louis, Missouri against Anheuser-Busch s board of directors and Anheuser-Busch (in part as nominal defendant), styled as *Pick v. Busch, et al.*, C.A. No. 0822-CC02134, and *New Jersey Carpenters Pension and Annuities Funds v. Busch, et al.*, C.A. No. 0822-CC07280. These plaintiffs alleged that the defendants breached their fiduciary duties by failing to give adequate consideration to the InBev non-binding proposal and taking improper defensive actions against the offer in an attempt to maintain their positions on Anheuser-Busch s board of directors. The plaintiffs generally sought declaratory relief that the defendants breached their fiduciary duties, injunctive relief to prevent such breaches and/or fees and expenses. On July 2, 2008, a different plaintiff filed an action with similar allegations in United Stated District Court for the Eastern District of Missouri, styled as *United Food & Commercial Workers Pension Fund of Northeastern PA v. Anheuser-Busch Companies, Inc., et al.*, C.A. 4:08-cv-00968. The plaintiff in the federal action sought damages in addition to the other forms of relief sought in the state actions.

The plaintiffs in the Missouri state court actions filed a motion to consolidate those two cases and for the appointment of lead counsel on June 26, 2008 and a motion for expedited discovery on June 30, 2008. Anheuser-Busch filed motions to stay the two Missouri state actions in favor of the substantially similar lawsuits pending in the Delaware Court of Chancery on June 27, 2008. On July 2, 2008, the Circuit Court of the City of St. Louis held a hearing on these motions and, on July 7, 2008, granted Anheuser-Busch s motion to stay the two Missouri state actions in favor of substantially similar Delaware actions, which are described below. The Circuit Court also ruled that the plaintiffs motion to consolidate the Missouri actions and motion for expedited discovery were moot. On July 11, 2008, the plaintiffs filed a motion to lift the stay and Anheuser-Busch filed an opposition brief on July 18, 2008. On July 21, 2008, the court removed the motion to lift the stay from its hearing docket, to be reset on application.

### **Delaware Shareholder Suits**

Between June 12, 2008 and July 2, 2008, a total of 11 substantially similar putative shareholder class and derivative actions were filed in the Delaware Court of Chancery against Anheuser-Busch s board of directors and Anheuser-Busch (in part as a nominal defendant), alleging that the defendants breached their fiduciary duties by failing to maximize shareholder value and adopting unreasonable defensive measures in the face of

53

### **Table of Contents**

the InBev non-binding proposal, including undertaking merger negotiations with Grupo Modelo. The plaintiffs generally sought declaratory relief that the defendants breached their fiduciary duties, injunctive relief to prevent such breaches, damages, and/or fees and expenses.

On June 18, 2008 and June 24, 2008, the plaintiffs in two of these actions moved for expedited discovery. The Court denied both motions. On June 25, 2008, one of the plaintiffs filed an Order of Dismissal seeking a voluntary dismissal of its action without prejudice. On July 10, 2008, the court consolidated the remaining ten actions under the caption *In re Anheuser- Busch Companies, Inc. Shareholders Litigation*, C.A. No. 3851, appointed the following as lead plaintiffs: Deka International S.A. Luxemburg, International Fund Management S.A. Luxemburg, Helaba Invest Kapitalanlagegesellschaft MBH, Deka Inestmentgesellschaft MBH, Deka Fundmaster Investmentgesellschaft MBH, the General Retirement System of the City of Detroit, and Sjunde AP-Fonden. The court further ordered the lead plaintiffs to file a consolidated amended derivative and class action complaint and that the defendants need not respond to any of the previously filed-complaints. The lead plaintiffs have not yet filed their consolidated amended complaint.

54

### **Table of Contents**

#### THE MERGER AGREEMENT

The summary of the material terms of the Merger Agreement below and elsewhere in this proxy statement is qualified in its entirety by reference to the Merger Agreement, a copy of which is attached to this proxy statement as Annex A and which we incorporate by reference into this document. This summary does not purport to be complete and may not contain all of the information about the Merger Agreement that is important to you. We encourage you to carefully read the Merger Agreement in its entirety.

The Merger Agreement has been included to provide you with information regarding its terms. It is not intended to provide any other factual information about Anheuser-Busch, InBev or Merger Sub. Such information can be found elsewhere in this proxy statement and in the other public filings Anheuser-Busch makes with the SEC, which are available without charge at www.sec.gov.

The representations, warranties and covenants contained in the Merger Agreement were made only for purposes of the Merger Agreement and as of a specific date and may be subject to more recent developments, were solely for the benefit of the parties to the Merger Agreement, may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures made for the purposes of allocating risk between parties to the Merger Agreement instead of establishing these matters as facts, and may apply standards of materiality in a way that is different from what may be viewed as material by you or by other investors. For the foregoing reasons, you should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of Anheuser-Busch, InBev or Merger Sub or any of their respective subsidiaries or affiliates.

## The Merger

The Merger Agreement provides for the Merger of Merger Sub with and into Anheuser-Busch upon the terms, and subject to the conditions, of the Merger Agreement. As the surviving corporation, Anheuser-Busch will continue to exist following the Merger as an indirect wholly-owned subsidiary of InBev. Upon consummation of the Merger, the directors of Merger Sub will be the initial directors of the surviving corporation and the officers of Anheuser-Busch will be the initial officers of the surviving corporation.

We or InBev may terminate the Merger Agreement prior to the consummation of the Merger in some circumstances, whether before or after the adoption by our stockholders of the Merger Agreement. Additional details on the parties ability to terminate the Merger Agreement are described in Termination of the Merger Agreement beginning on page 66.

### **Effective Time**

The Merger will be effective at the time the certificate of merger is filed with the Secretary of State of the State of Delaware or such later time as agreed by the parties to the Merger Agreement in writing and specified in the certificate of merger. We expect to complete the Merger as soon as practicable after our and InBev s stockholders adopt the Merger Agreement and all other conditions to closing have been satisfied or waived.

Unless otherwise agreed by the parties to the Merger Agreement, the parties are required to close the Merger no later than the third business day after the satisfaction or waiver of the conditions described under Conditions to the Merger beginning on page 65.

# **Merger Consideration**

Each share of our common stock issued and outstanding immediately prior to the effective time of the Merger will be converted into the right to receive \$70.00 in cash, without interest and less any applicable withholding taxes, other than the following shares:

owned by InBev, Merger Sub or any other direct or indirect wholly owned subsidiary of InBev (other than shares of common stock held on behalf of third parties);

55

### **Table of Contents**

owned by Anheuser-Busch or any direct or indirect wholly owned subsidiary of Anheuser-Busch (other than shares of common stock held on behalf of third parties); and

owned by stockholders who have perfected and not withdrawn a demand for appraisal rights pursuant to Section 262 of the DGCL.

After the effective time of the Merger, each holder of any shares of our common stock (other than shares for which appraisal rights have been properly demanded and perfected and not withdrawn) will no longer have any rights with respect to the shares, except for the right to receive the merger consideration. See Dissenters Rights of Appraisal beginning on page 75.

### **Treatment of Options and Other Awards**

Anheuser-Busch Stock Options. At the effective time of the Merger, each outstanding option to acquire our common stock under equity incentive plans, vested or unvested, will be cancelled and converted into the right to receive within 3 business days following the Merger a cash payment equal to the number of shares of our common stock underlying such option multiplied by the amount by which \$70.00 exceeds the exercise price of such option, less any applicable withholding taxes.

Anheuser-Busch Awards. At the effective time of the Merger, each right of any kind, contingent or accrued, to acquire or receive stock payable under Anheuser-Busch benefit plans, vested or unvested, will be cancelled and converted into the right to receive within 3 business days following the Merger a cash payment equal to the number of shares of our common stock subject to any such award immediately prior to the effective time multiplied by either (i) \$70.00 or (ii) if the Anheuser-Busch award has a specified reference price, the amount by which \$70.00 exceeds such reference price, in each case less any applicable withholding taxes.

*Employee Stock Purchase Plan.* Anheuser-Busch s Global Employee Stock Purchase Plan will terminate as of immediately prior to the effective time of the Merger, and no new share offering date will commence under the plan following the date of the Merger Agreement, and unused amounts held in participant savings accounts will be returned to participants.

The effect of the Merger upon our other employee benefit plans is more fully described under beginning on page 69.

### **Payment for the Shares of Common Stock**

InBev will select a paying agent with our prior approval to make payment of the merger consideration as described above. Simultaneously with the effective time of the Merger, InBev will deposit or cause to be deposited with such paying agent a cash amount in immediately available funds to pay the aggregate merger consideration payable to our stockholders as well as any amounts described above that are payable to option holders and Anheuser-Busch award recipients.

Following the effective time of the Merger, there will be no further transfer of shares of our common stock.

Promptly after the effective time of the Merger and, in any event, within five business days in the case of record holders and three business days in the case of street holders, the surviving corporation will cause the paying agent to send you a letter of transmittal and instructions advising you how to surrender your shares of common stock in exchange for the merger consideration. The paying agent will pay you the merger consideration to which you are

entitled after you have provided to the paying agent your signed letter of transmittal and any other items specified by the letter of transmittal. Interest will not be paid or accrue in respect of the merger consideration. The surviving corporation will reduce the amount of any merger consideration paid to you by any applicable withholding taxes.

If any cash deposited with the paying agent is not claimed within nine months following the effective time of the Merger, such cash will be returned to the surviving corporation. Holders who have not received the merger consideration due to non-compliance with the exchange of certificates procedures must seek payment from the surviving corporation.

56

### **Table of Contents**

## **Representations and Warranties**

In the Merger Agreement, Anheuser-Busch made representations and warranties relating to, among other things:

the corporate organization, good standing and qualification of Anheuser-Busch and its subsidiaries;

its capital structure;

its corporate power and authority to enter into the Merger Agreement;

the fairness opinions delivered by Goldman Sachs and Citi;

required regulatory filings, consents and approvals of governmental entities;

absence of conflicts with law, Anheuser-Busch s organizational documents and contracts to which Anheuser-Busch is a party;

Anheuser-Busch s financial statements filed with the SEC and related matters;

compliance with NYSE listing and corporate governance rules, and Sarbanes-Oxley;

absence of changes since December 31, 2007;

absence of litigation and unknown liabilities of Anheuser-Busch;

employee benefits;

Anheuser-Busch s compliance with laws;

material contracts;

Anheuser-Busch s owned and leased United States properties;

lack of applicability of takeover statutes to the Merger;

environmental matters;

taxes:

United States labor matters;

intellectual property;

brokers and finders; and

Anheuser-Busch s solvency as of immediately prior to the effective time of the Merger.

In the Merger Agreement, InBev and Merger Sub each made representations and warranties relating to:

the corporate organization, good standing and qualification of InBev and Merger Sub;

corporate power and authority to enter into and consummate the transactions contemplated by the Merger Agreement;

required regulatory filings, consents and approvals of governmental entities;

the availability of the funds necessary to perform their obligations under the Merger Agreement;

the capitalization of Merger Sub;

absence of litigation and judgments that are reasonably likely to prevent or materially impair the consummation of the transactions contemplated by the Merger Agreement;

solvency of the surviving corporation immediately after the effective time of the Merger;

their ownership of our common stock; and

brokers and finders.

57

### **Table of Contents**

Many of Anheuser-Busch s representations and warranties are qualified by a Material Adverse Effect standard. For purposes of the Merger Agreement, Material Adverse Effect is defined to mean:

A material adverse effect on the financial condition, business or results of operations of Anheuser-Busch and its subsidiaries taken as a whole, provided that none of the following, in and of itself or themselves, shall constitute (or be taken into account in determining the occurrence of) a Material Adverse Effect:

effects resulting from changes in the economy or financial, credit, banking, currency, commodities or capital markets generally in the United States or other countries in which Anheuser-Busch conducts material operations or any changes in currency exchange rates, interest rates, monetary policy or inflation, except any change, event, circumstance or development disproportionately adversely affects Anheuser-Busch and its subsidiaries compared to other companies operating in the beer, packaging or theme park industries (but only to the extent of such disproportionate effect);

effects resulting from changes that are the result of factors generally affecting the beer, packaging or theme park industries, except any change, event, circumstance or development disproportionately adversely affects Anheuser-Busch and its subsidiaries compared to other companies operating in the beer, packaging or theme park industries (but only to the extent of such disproportionate effect);

effects resulting from changes in law or in United States generally accepted accounting principles (GAAP) or rules and policies of the Public Company Accounting Oversight Board, except any change, event, circumstance or development disproportionately adversely affects Anheuser-Busch and its subsidiaries compared to other companies operating in the beer, packaging or theme park industries (but only to the extent of such disproportionate effect);

any act of God or other calamity, national or international, political or social conditions (including the engagement by any country in hostilities, whether commenced before or after the date of the Merger Agreement, and whether or not pursuant to the declaration of a national emergency or war), or the occurrence of any military or terrorist attack, except any change, event, circumstance or development disproportionately adversely affects Anheuser-Busch and its subsidiaries compared to other companies operating in the beer, packaging or theme park industries (but only to the extent of such disproportionate effect);

effects resulting from any failure by Anheuser-Busch to meet any estimates of revenues or earnings on or after the date of the Merger Agreement, provided that this exception shall not prevent or otherwise affect any change, effect, circumstance or development underlying such failure from being taken into account in determining whether a Material Adverse Effect has occurred;

the announcement or the existence of the Merger Agreement and the transactions contemplated by the Merger Agreement (including any related or resulting loss of or change in relationship with any customer, supplier, distributor, wholesaler or other business partner, or departure of any employee or officer, or any litigation or other proceeding), including by reason of the identity of InBev or any plans or intentions of InBev with respect to the conduct of the business of Anheuser-Busch or its subsidiaries;

compliance with the terms of, any actions taken pursuant to, or any failure to take action prohibited by, the Merger Agreement, or such other changes or events to which InBev has expressly consented in writing; or

certain other agreed upon exceptions, including (A) any impact of the Merger Agreement or the transactions contemplated thereby on Anheuser-Busch s investment in Grupo Modelo would not constitute a breach of the

Merger Agreement or relieve or excuse InBev or Merger Sub of their obligations under the Merger Agreement, and (B) with respect to adverse developments, if any, in Anheuser-Busch s labor relations.

## Litigation; Consent Solicitation

The parties have agreed to voluntarily dismiss all pending litigation between themselves relating to the removal or removability of directors, the transactions contemplated by the Merger Agreement and InBev s

58

### **Table of Contents**

consent solicitation. InBev has also agreed to terminate its proposed consent solicitation to remove and replace Anheuser-Busch s board of directors.

### **Conduct of Business Prior to Closing**

We have agreed in the Merger Agreement that, until the effective time of the Merger, except as expressly contemplated by the Merger Agreement or required by applicable law, we and our subsidiaries will:

operate our businesses in all material respects in the ordinary and usual course of business;

to the extent consistent with the above, use commercially reasonable efforts to preserve our business organizations and maintain existing relations and goodwill with governmental entities, customers, suppliers, licensors, licensees, distributors, creditors, lessors, employees and business associates (except as announced by Anheuser-Busch prior to the date of the Merger Agreement); and

prior to making any written communications to the employees of Anheuser-Busch or its subsidiaries pertaining to material compensation or benefit matters contemplated by the Merger Agreement, provide InBev with such a copy of such communication, provide a reasonable period of time to review and comment on such communications, and consider any comments that InBev may have.

We have also agreed that, until the consummation of the Merger, except as expressly contemplated by the Merger Agreement or required by applicable law or consented to in writing by InBev (which consent will not be unreasonably withheld or delayed) or set forth in the disclosure schedule that Anheuser-Busch delivered to InBev in connection with the execution of the Merger Agreement, we and our subsidiaries will not:

adopt or propose any change in our organizational or governing documents;

merge or consolidate Anheuser-Busch or any of its subsidiaries (except for transactions with wholly owned subsidiaries);

restructure, reorganize, liquidate or otherwise enter into any contracts imposing material changes or restrictions on our assets, operations or businesses, other than in the ordinary course;

acquire stock or assets other than in the ordinary course of business or, if outside the ordinary course of business, in an amount not to exceed \$50 million for any transaction or related series of transactions;

issue, sell, pledge, transfer or otherwise dispose of any shares of capital stock of Anheuser-Busch and its subsidiaries or any rights to acquire Anheuser-Busch securities (other than issuances of common stock pursuant to outstanding awards under Anheuser-Busch s equity award plans);

create or incur any lien on intellectual property owned or exclusively licensed by Anheuser-Busch or its subsidiaries or on other material intellectual property or on assets having a value in excess of \$50 million;

make any loans, advances, guarantees or capital contributions or investments (other than to Anheuser-Busch or its wholly owned subsidiaries) in excess of \$10 million in the aggregate;

declare or pay dividends (except for regular quarterly dividends not to exceed \$0.37 per share) or enter into any contract with respect to the voting of Anheuser-Busch s capital stock;

reclassify, split, combine, subdivide or redeem, purchase or otherwise acquire, any Anheuser-Busch capital stock;

incur, alter, amend or modify any indebtedness for borrowed money or guarantee of a third party s indebtedness (other than ordinary course borrowing consistent with past practice that (i) does not exceed \$200 million in the aggregate, (ii) is in replacement of existing Anheuser-Busch indebtedness, (iii) guarantees the indebtedness of wholly owned subsidiaries otherwise in compliance with Anheuser-Busch s operational covenants or (iv) interest rate swaps consistent with past practice not to exceed \$500 million of notional debt in the aggregate);

59

### **Table of Contents**

make or authorize capital expenditures in excess of \$875 million in the aggregate during any 12 month period;

enter into any contract that would be classified as a material contract under the Merger Agreement, other than in the ordinary course of business or in connection with an acquisition permitted under the terms of the Merger Agreement;

make material changes in accounting policies or procedures, except as required by changes in GAAP;

settle any action before a governmental authority or any obligation or liability of Anheuser-Busch above a certain threshold amount (excluding amounts covered by insurance) or agree to any material limitation/restriction on any aspect of Anheuser-Busch s or its subsidiaries business;

amend or modify in any material respect or terminate any material contract or intellectual property contract or cancel, modify or waive any material debts or claims other than, in each case, in the ordinary course of business:

make any tax election, amend any tax return, settle any controversy with respect to taxes or change any method of tax accounting, where the amount of taxes in question is greater than \$50 million;

transfer, sell or otherwise dispose of (i) any of Anheuser-Busch s material intellectual property, other than pursuant to non-material contracts granted in the ordinary course of business or (ii) any of Anheuser-Busch s material assets, product lines or businesses, except for dispositions in the ordinary course of business and sale of obsolete assets or other dispositions of less than \$50 million in the aggregate; and

except pursuant to written, binding plans or as required by applicable law:

grant or provide any severance or termination payments or benefits to any director, officer or employee (other than certain severance and retention plans approved by InBev which are described under The Merger Agreement Employee Benefits beginning on page 69 of this proxy statement);

increase the compensation, bonus or benefits of, pay any bonus to, or make any new equity awards to any director, officer or employee (except to non-officer employees in the ordinary course of business consistent with past practice);

establish, adopt, amend or terminate any Anheuser-Busch benefit plan (except as required by law) or amend the terms of any outstanding equity-based awards;

accelerate vesting or payment, or fund or secure payment, of compensation or benefits under any Anheuser-Busch benefit plan (if not already contemplated by such plan);

materially change any actuarial or other assumptions used to calculate funding obligations with respect to any Anheuser-Busch benefit plan or change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as required by GAAP; and

forgive any loans made to directors, officers or employees of Anheuser-Busch or its subsidiaries.

InBev and Merger Sub have agreed not to (i) take any action that is reasonably likely to prevent or materially impair the consummation of the Merger or (ii) enter into any acquisition agreement, or make any acquisition, that is

reasonably likely to prevent, materially delay or impair the consummation of the Merger. InBev has also agreed not to issue any common stock or subscription rights with respect thereto prior to obtaining approval of its shareholders of the Merger unless the InBev s controlling shareholder, Stichting InBev AK, agrees to vote any shares of InBev common stock that it received in such issuance in favor of the Merger and the other transactions contemplated by the Merger Agreement.

We have also agreed to provide to InBev, upon reasonable notice, reasonable access during normal business hours to our employees, books, contracts and records.

60

#### **Table of Contents**

### Agreements to Take Further Action and to Use Reasonable Best Efforts

Subject to the terms and conditions set forth in the Merger Agreement, each of Anheuser-Busch and InBev has agreed to use their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on their part under the Merger Agreement and applicable law to consummate the Merger and the other transactions contemplated by the Merger Agreement as soon as practicable, including preparing and filing all documentation to effect all necessary notices, reports and other filings and to obtain all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or governmental entity to consummate the Merger or any of the other transactions contemplated by the Merger Agreement.

Except as may be required to protect privilege or work product protections, InBev and Anheuser-Busch have the right to review in advance all of the information relating to the other party and their subsidiaries (or in certain circumstances, such other party s counsel) that appears in any filing made with, or written materials submitted to, any third party and/or any governmental entity in connection with the Merger and the other transactions contemplated by the Merger Agreement.

InBev s obligation to use reasonable best efforts to obtain clearance from a governmental entity has been qualified so as to not require InBev to proffer to, or agree to, divest, dispose of or otherwise encumber or hold separate before or after the effective time of the Merger any assets, licenses, operations, rights, product lines or businesses of InBev or Anheuser-Busch or any of their respective affiliates or to agree to any material changes (including through licensing arrangements) or restriction on, or other impairment of, InBev s ability to own or operate any such assets, licenses, operations, rights, product lines or businesses or InBev s ability to vote or otherwise exercise full ownership rights with respect to the stock of Anheuser-Busch to the extent such actions would, individually or in the aggregate, result in reduction of assets, categories of assets, businesses or investments that generated in the aggregate more than 5% of the sum of the 2007 gross sales and investment income of InBev and Anheuser-Busch, including their respective subsidiaries, on a combined basis net of any ongoing annual royalty payments or other ongoing revenue contributions expected to be received as a result of effectuating such divestitures or other arrangements.

Anheuser-Busch and InBev have agreed to promptly provide to federal, state, local or foreign courts or governmental antitrust entities with non-privileged information and documents (i) requested by any governmental antitrust entity or (ii) that are necessary for the consummation of the transactions contemplated by the Merger Agreement.

Additionally, Anheuser-Busch has agreed to prepare and file this proxy statement as promptly as practicable following the date the parties entered into the Merger Agreement and InBev agreed to promptly convene a meeting of its shareholders to approve, among other things, the acquisition of Anheuser-Busch.

Anheuser-Busch and InBev have agreed to, upon request by the other, furnish all information concerning itself, its subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with the proxy statement, the InBev prospectus, the InBev shareholders meeting materials or any other statement, filing, notice or application.

### **Financing**

#### **Debt Financing**

In accordance with the Merger Agreement, InBev has agreed to satisfy on a timely basis all conditions applicable to InBev under the senior debt and bridge facility agreements and must obtain and consummate the financing contemplated by the senior debt and bridge facility agreements and if such financing is unavailable, obtain and consummate alternative financing.

InBev must give Anheuser-Busch prompt notice upon becoming aware of any termination of the senior debt and bridge facility agreements and of the occurrence of any event that would prevent InBev from satisfying one or more of the conditions to initial utilization in the senior debt and bridge facility agreements. In the event that InBev becomes aware of any event or circumstance that makes any portion of the financing

61

#### **Table of Contents**

of the senior debt and bridge facility agreements unavailable prior to the Termination Date (as described below) on the terms and conditions set forth in the senior debt and bridge facility agreements, InBev must secure as promptly as practicable any such unavailable portion from alternative sources.

Each of InBev and Merger Sub have expressly acknowledged that the Merger is not conditioned on the availability of the funding under the senior debt and bridge facility agreements or any alternative financing arrangement.

### Cooperation of Anheuser-Busch

We have agreed to use our commercially reasonable efforts to provide InBev and its representatives with reasonable access during normal business hours to our books, records, personnel and auditors. Additionally, in connection with InBev s debt financing and equity or equity-linked capital raising activities, we have agreed to:

use commercially reasonable efforts to make our personnel reasonably available for participation in meetings, presentations, road shows, due diligence sessions (including accounting due diligence sessions) and sessions with rating agencies;

use commercially reasonable efforts to provide InBev with all information within our possession as is reasonably requested by InBev for the preparation of materials for rating agency presentations, offering documents, private placement memoranda, bank information memoranda, prospectuses and similar documents required in connection therewith (including preparation of combined financial information on a pro forma basis);

use commercially reasonable efforts to provide InBev with all information within our possession as is reasonably requested by InBev for preparation of additional definitive financing documents, and consider in good faith InBev s reasonable requests for certificates, legal opinions or documents as may be reasonably required under the Facilities or in connection with InBev s equity financing (it being understood that the failure of Anheuser-Busch to provide any such certificate, legal opinion or document will not constitute a breach of the Merger Agreement by Anheuser-Busch);

furnish InBev and its lenders or the underwriters of InBev s equity financing with financial and other pertinent information regarding our and our subsidiaries businesses that is within our possession as may be reasonably requested by InBev (including providing monthly financial statements (excluding footnotes) and in the form that Anheuser-Busch customarily prepares such financial statements within the time such statements are customarily prepared);

use commercially reasonable efforts to enable InBev to obtain accountants comfort letters, as reasonably requested by InBev (whether addressed to the underwriters of InBev s equity financing and/or InBev) (it being understood that the failure of Anheuser-Busch s accountants to agree to provide such comfort letters or the failure to provide such comfort letters will not constitute a breach of the Merger Agreement by Anheuser-Busch, in each case so long as Anheuser-Busch has used such commercially reasonable efforts); and

use commercially reasonable efforts to permit the lenders under the Facilities or the underwriters of InBev s equity financing, as well as InBev, to evaluate the assets of Anheuser-Busch and its Subsidiaries, and the cash management and accounting systems, policies and procedures associated therewith.

We are not required to take any of the actions described above to the extent taking such action would, in our reasonable judgment, interfere unreasonably with our or our subsidiaries business or operations or would require our board or the board of directors of any of our subsidiaries to take any action.

Additionally, the Merger Agreement limits our obligation to incur any fees or liabilities with respect to the debt or equity financing prior to the effective time of the Merger. InBev has also agreed to reimburse us for all reasonable out-of-pocket costs and to indemnify and hold harmless Anheuser-Busch, our subsidiaries, directors, officers, employees, representatives and advisors from and against all losses, damages, claims, costs

62

#### **Table of Contents**

or expenses suffered or incurred in connection with the debt financing and any information utilized in connection therewith.

### **Post-Closing Agreements of InBev**

InBev has agreed that, effective upon the closing of the transactions contemplated by the Merger Agreement:

Anheuser-Busch s current headquarters in St. Louis, Missouri will be the surviving corporation s headquarters, InBev s headquarters for North America (excluding Cuba) and the global home of the flagship Budweiser brand:

the current name of Anheuser-Busch will be the name of the surviving corporation, and the name of InBev will be Anheuser-Busch InBev N.V./S.A.; and

InBev will, after consultation with the Anheuser-Busch board of directors, nominate and cause to be elected two current or former directors of Anheuser-Busch to the board of directors of InBev, and each such director shall be confirmed for a three-year term at the first annual general meeting of InBev following the closing of the transactions contemplated by the Merger Agreement.

Following the closing of the transactions contemplated by the Merger Agreement, InBev will:

cause the surviving corporation to preserve Anheuser-Busch sheritage and continue to support philanthropic and charitable causes in St. Louis and other communities in which Anheuser-Busch operates, including Grant s Farm and the Clydesdales operations;

confirm the surviving corporation s good faith commitment that it will not close any of Anheuser-Busch s current 12 breweries located in the United States, provided there are no new or increased federal or state excise taxes or other unforeseen extraordinary events which negatively impact Anheuser-Busch s business;

reaffirm its commitment to the three-tier distribution system in the United States and agree to work with Anheuser-Busch s existing wholesaler panel to strengthen the relationship between the surviving corporation and its wholesalers; and

honor Anheuser-Busch s obligations under the Naming Rights and Sponsorship Agreement, dated August 3, 2004, as amended, between Busch Media Group, Inc., as authorized agent for Anheuser-Busch, Incorporated and Cardinals Ballpark, LLC relating to Busch Stadium.

#### No Solicitation of Other Offers

In connection with the Merger Agreement, we have agreed not to:

initiate, solicit or knowingly encourage any inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any acquisition proposal;

engage in, continue or otherwise participate in any discussions or negotiations regarding, or provide any non-public information or data to any person relating to, any acquisition proposal; or

otherwise knowingly facilitate any effort or attempt to make an acquisition proposal.

Notwithstanding these restrictions, prior to the time that our stockholders approve the Merger, if Anheuser-Busch has otherwise complied in all material respects with its obligations under the no solicitation covenant of the Merger Agreement, Anheuser-Busch may provide information with respect to Anheuser-Busch and its subsidiaries to any person who has made an unsolicited, written acquisition proposal so long as:

our board of directors believes in good faith that such acquisition proposal is bona fide;

such acquisition proposal provides for the acquisition of more than 50% of our assets (on a consolidated basis) or total voting power of our equity securities; and

63

#### **Table of Contents**

our board of directors concludes in good faith, after consultation with its financial advisors and outside legal counsel, that such acquisition proposal constitutes (in the event our board of directors proposes to approve, recommend or otherwise declare advisable such acquisition proposal) or is reasonably likely to result in a superior proposal.

We are also permitted to participate in discussions or negotiations with any person making any acquisition proposal that is consistent with the terms described above.

In the cases described above, we are not permitted to disclose any non-public information to a person without entering into a confidentiality agreement that contains provisions that are no less favorable in the aggregate to us than those contained in our confidentiality agreement with InBev. In addition, we will promptly provide to InBev any non-public information concerning us or our subsidiaries provided to such other person which was not previously provided to InBev.

We have agreed to promptly (and, in any event, within 24 hours) notify InBev if any proposals or offers with respect to an acquisition proposal are received by, any such information is requested from, or any such discussions or negotiation are sought to be initiated or continued with, us or any of our representatives indicating the name of such person and the material terms and conditions of any proposals or offers (including, if applicable, copies of any written requests, proposals or offers, including proposed agreements), and to keep InBev informed, on a reasonably current basis, of the status and terms of any such proposals or offers (including any amendments).

An acquisition proposal means any proposal or offer with respect to a merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction involving Anheuser-Busch or any of its significant subsidiaries, and any acquisition by any person, or proposal or offer, which if consummated would result in any person becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions, 15% or more of the total voting power or of any class of equity securities of Anheuser-Busch or those of any of its significant subsidiaries, or 15% or more of the consolidated total assets (including equity securities of its subsidiaries) of Anheuser-Busch, in each case other than the transactions contemplated by the Merger Agreement.

A superior proposal means a bona fide acquisition proposal in connection with which there has been no material violation of the no solicitation covenant contained in the Merger Agreement that would result in any person (or its stockholders) becoming the beneficial owner, directly or indirectly, of more than 50% of the assets (on a consolidated basis) or more than 50% of the total voting power of the equity securities of Anheuser-Busch that our board of directors has determined in its good faith judgment is reasonably likely to be consummated in accordance with its terms, taking into account all legal, financial and regulatory aspects of the proposal and the person making the proposal, and if consummated, would result in a transaction more favorable to Anheuser-Busch s stockholders from a financial point of view than the transactions contemplated by the Merger Agreement (after taking into account any revisions to the terms of the Merger proposed by InBev).

In addition to the rights described above, we may terminate the Merger Agreement and enter into a definitive agreement with respect to a superior proposal under certain circumstances. See Recommendation Withdrawal/Termination in Connection with a Superior Proposal.

#### Recommendation Withdrawal/Termination in Connection with a Superior Proposal

Our board of directors has unanimously resolved to recommend that our stockholders adopt the Merger Agreement. However, if our board of directors determines in good faith, after consultation with outside counsel, that the failure to

take such action would be inconsistent with the board s fiduciary duties to our stockholders under applicable law, it may, at any time prior to, but not after, the adoption of the Merger Agreement by our stockholders:

withhold, withdraw, qualify or modify, or publicly propose to withhold, withdraw, qualify or modify, in a manner adverse to InBev, its recommendation that our stockholders adopt the Merger Agreement;

64

#### **Table of Contents**

enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other agreement relating to any acquisition proposal.

However, in order to enter into an agreement relating to an acquisition proposal, our board must also have determined in good faith, after consultation with our financial advisors, that such acquisition proposal is a superior proposal and remains so after our board follows the procedures described below.

To the extent the board proposes to take the foregoing actions with regard to its recommendation, it may only do so if:

we have not materially breached our obligations under the no solicitation provision of the Merger Agreement;

we have given InBev at least 72 hours notice of our intention to change our recommendation or terminate the Merger Agreement;

prior to terminating the Merger Agreement, our board of directors has taken into account any changes to the terms of the Merger Agreement proposed by InBev and other information provided by InBev during such 72-hour period; and

Any material amendment to an acquisition proposal will be deemed a new acquisition proposal that will entitle InBev to an additional 72 hour period to propose changes to the Merger Agreement and provide additional information.

In addition, we are not entitled to enter into any agreement with respect to a superior proposal unless we have concurrently paid to InBev the applicable termination fee as described in further detail in Termination Fees beginning on page 67.

#### **Conditions to the Merger**

Conditions to Each Party s Obligations. Each party s obligation to complete the Merger is subject to the satisfaction or waiver at or prior to the effective time of the Merger of each of the following conditions:

the Merger Agreement must have been adopted by the affirmative vote of the holders of a majority of the outstanding shares of our common stock, and a resolution approving the Merger (and the change of InBev s name) must have been passed by the holders of 75% of the ordinary shares of InBev common stock present in person or by proxy at an extraordinary shareholders meeting of InBev;

any applicable waiting period under the HSR Act must have expired or been earlier terminated and all other required approvals and authorizations from other applicable antitrust authorities must have been obtained; and

no court or other governmental entity shall have enacted, issued, promulgated, enforced or entered any law (temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Merger.

Conditions to InBev s and Merger Sub s Obligations. The obligation of InBev and Merger Sub to complete the Merger is subject to the satisfaction or waiver at or prior to the effective time of the Merger of each of the following additional conditions:

as of the date of the Merger Agreement and as of the Closing Date, (i) our representations and warranties that are qualified by Material Adverse Effect must be true and correct (unless such representation or warranty

expressly speaks as of an earlier date, in which case such representation or warranty shall be true and correct as of such earlier date) and (ii) our representations and warranties that are not so qualified must be true and correct provided that the condition set forth in (ii) will be deemed to have been satisfied even if any representations and warranties are not true and correct unless such failure, individually or in the aggregate, would, or would be reasonably likely to, result in a material adverse effect (except for representations and warranties with respect to our capital structure, corporate power and authority to enter into the Merger Agreement, the fairness opinions delivered by

65

#### **Table of Contents**

Goldman Sachs and Citi and the applicability of anti-takeover statues, which must be true in all material respects);

we must have performed in all material respects all obligations required to be performed under the Merger Agreement at or prior to the closing date;

since the date of the Merger Agreement, there must not have been any change, event, circumstance or development that has had, or would be reasonably likely to have, a Material Adverse Effect; and

we must deliver to InBev at closing a certificate with respect to the satisfaction of the foregoing conditions relating to representations, warranties and obligations.

Conditions to Anheuser-Busch s Obligations. Our obligation to complete the Merger is subject to the satisfaction or waiver at or prior to the effective time of the Merger of each of the following additional conditions:

as of the date of the Merger Agreement and as of the Closing Date (unless such representation or warranty expressly speaks as of an earlier date, in which case such representation or warranty shall be true and correct as of such earlier date), InBev s representations and warranties must be true and correct in all material respects;

InBev and Merger Sub must have performed in all material respects all obligations required to be performed by them under the Merger Agreement at or prior to the closing date; and

InBev must deliver to us at closing a certificate with respect to the satisfaction of the foregoing conditions relating to representations, warranties and obligations.

### **Termination of the Merger Agreement**

The Merger Agreement may be terminated at any time prior to the consummation of the Merger, whether before or after stockholder approval has been obtained:

by mutual written consent of Anheuser-Busch and InBev;

by either Anheuser-Busch or InBev if:

the Merger is not completed on or before March 19, 2009 (the Termination Date), so long as the failure of the Merger to be completed by such date is not the result of, or caused by, the failure of the terminating party to comply in all material respects with the covenants and agreements set forth in the Merger Agreement;

our stockholders do not adopt the Merger Agreement at the special meeting or any adjournment or postponement thereof;

a resolution by InBev s shareholders approving the Merger has not been obtained at the extraordinary shareholders meeting of InBev (however, InBev may not terminate the Merger Agreement if Stichting InBev AK is in breach of its voting agreement with Anheuser-Busch); or

any order, decree or ruling permanently restraining, enjoining or otherwise prohibiting consummation of the Merger shall become final and non-appealable;

provided, that the party seeking to terminate the Merger Agreement has not breached in any material respect its obligations under the Merger Agreement in any manner that shall have resulted in the failure of a condition to the consummation of the Merger;

by Anheuser-Busch, if:

prior to obtaining stockholder approval, we terminate the Merger Agreement in order to enter into an agreement with respect to a superior proposal and, concurrently with such termination, pay InBev the required termination fee;

66

#### **Table of Contents**

InBev or Merger Sub has breached any of its representations, warranties, covenants or agreements under the Merger Agreement, or any such representation or warranty made by them shall have become untrue, such that certain conditions to closing would not be satisfied and such breach or condition is not curable, or if curable, is not cured, within the earlier of 30 days after written notice of such breach or the Termination Date; or

all of the conditions to the obligations of InBev and Merger Sub have been satisfied and InBev has failed to consummate the Merger no later than 30 calendar days after the satisfaction of such conditions;

by InBev, if:

our board of directors adversely changes or withdraws its recommendation that our stockholders adopt the Merger Agreement;

following the receipt of an acquisition proposal, we fail to reaffirm the recommendation of our board of directors that our stockholders adopt the Merger Agreement within 10 business days of InBev s reasonable written request that we do so;

following the public disclosure (other than by InBev or its affiliates) of a tender offer or exchange offer for outstanding shares of Anheuser-Busch s common stock, we fail to recommend unequivocally against such tender or exchange offer within the earlier of one day prior to the special meeting or 11 business days after the commencement of such offer; or

we have breached any of our representations, warranties, covenants or agreements under the Merger Agreement, or any such representation or warranty made by us shall have become untrue, such that certain conditions to closing would not be satisfied and such breach or condition is not curable, or if curable, is not cured, within the earlier of 30 days after written notice of such breach or the Termination Date.

### **Termination Fees**

If the Merger Agreement is terminated by us or by InBev or Merger Sub under the conditions described in further detail below, a termination fee in the amount of \$1,250,000,000 (the termination fee ) may be payable to Anheuser-Busch or InBev, as applicable.

We must pay the termination fee at the direction of InBev if:

we terminate the Merger Agreement prior to the special meeting in order to enter into a definitive agreement for a superior proposal;

InBev terminates the Merger Agreement because:

our board of directors adversely changes or withdraws its recommendation that our stockholders adopt the Merger Agreement;

following receipt of an acquisition proposal, our board of directors fails to reaffirm our approval or recommendation of the Merger within 10 business days of InBev s reasonable written request that we do so;

following the public disclosure (other than by InBev or its affiliates) of a tender offer or exchange offer for outstanding shares of Anheuser-Busch s common stock, we fail to recommend unequivocally against such tender or exchange offer within the earlier of one day prior to the special meeting or 11 business days after the commencement of such offer; or

we have breached any of our representations, warranties, covenants or agreements under the Merger Agreement, or any such representation or warranty made by us shall have become untrue, such that certain conditions to closing would not be satisfied and such breach or condition is not curable, or if curable, is not cured, within the earlier of 30 days after written notice of such breach or the

67

#### **Table of Contents**

Termination Date (or if we have terminated the Merger Agreement following the failure of our stockholders to adopt the Merger Agreement and any of the above events have occurred);

we or InBev terminate the Merger Agreement because our stockholders fail to adopt the Merger Agreement or because the Merger is not consummated by the Termination Date (unless the Merger has not been consummated by the Termination Date because of a breach by InBev of its obligations under the Merger Agreement) and, in each case,

a bona fide acquisition proposal has been made to Anheuser-Busch or any of its subsidiaries or to its stockholders generally, or any person has publicly announced an intention (whether or not conditional) to make a bona fide acquisition proposal, and such acquisition proposal or publicly announced intention has not been publicly withdrawn without qualification at least (A) 30 business days prior to the date of termination (in the case of termination of failure to consummate the Merger by the Termination Date), and (B) at least 10 business days prior to the date of the special meeting (in the case of termination for failure to obtain stockholder approval); and

within twelve months after such termination we enter into an agreement with respect to, or consummate, any acquisition proposal.

InBev must pay the termination fee to Anheuser-Busch if we or InBev terminate the Merger Agreement for failure to obtain a resolution of InBev s shareholders approving the Merger.

If InBev pays a termination fee to us and (i) InBev has not breached its obligations (x) to not enter into any acquisition agreement, or make any acquisition, that is reasonably likely to prevent, materially delay or impair the consummation of the Merger, (y) to not issue any InBev common stock (or subscription rights with respect thereto) prior to obtaining the approval of InBev shareholders at the extraordinary shareholders meeting of InBev (unless Stichting InBev AK has agreed to vote any additional InBev shares received by it, or received by others who are bound to vote together with Stichting InBev AK, in favor of the proposal to approve the Merger) and (z) to call the extraordinary shareholders meeting of InBev s shareholders and recommend the approval of the Merger at such extraordinary shareholders meeting, and (ii) Stichting InBev AK has not breached any of its obligations under the voting agreement, then such termination fee shall be Anheuser-Busch s sole and exclusive remedy for monetary damages under the Merger Agreement. If Anheuser-Busch pays a termination fee to InBev, such termination fee shall be InBev s and Merger Sub s sole and exclusive remedy for monetary damages under the Merger Agreement.

### **Specific Performance**

Each of the parties is specifically authorized to obtain an order of specific performance to enforce performance of any covenant or obligation under the Merger Agreement or injunctive relief to restrain any breach or threatened breach, including, in the case of Anheuser-Busch, to specifically enforce InBev s obligations to seek to cause its financing to be funded and InBev s obligations to consummate the Merger.

### **Indemnification and Insurance**

Prior to the effective time of the Merger, Anheuser-Busch will, and if Anheuser-Busch is unable to, InBev will cause the surviving corporation to, obtain and fully pay for tail insurance policies with a claims period of at least six years from and after the effective time of the Merger from an insurance carrier with the same or better credit rating as Anheuser-Busch s current insurance carrier with respect to directors and officers liability insurance and fiduciary liability insurance (collectively, D&O Insurance), for the persons who, as of the date of the Merger Agreement, were covered by Anheuser-Busch s existing D&O Insurance, with terms, conditions, retentions and levels of coverage at

least as favorable as Anheuser-Busch s existing D&O Insurance with respect to matters existing or occurring at or prior to the effective time of the Merger (including in connection with the Merger Agreement or the transactions or actions contemplated thereby). If Anheuser-Busch and the surviving corporation for any reason fail to obtain such tail insurance policies, InBev has agreed to cause the surviving corporation to continue to maintain in effect Anheuser-Busch s current D&O Insurance, at no expense to the beneficiaries, for a period of at least six years from and after the effective time

68

#### **Table of Contents**

of the Merger. If such insurance is unavailable, InBev has agreed to cause the surviving corporation to purchase the best available D&O Insurance for such six-year period from an insurance carrier with the same or better credit rating as Anheuser-Busch s current insurance carrier with respect to Anheuser-Busch s existing D&O Insurance with terms, conditions, retentions and with levels of coverage at least as favorable as provided in Anheuser-Busch s existing policies as of the date of the Merger Agreement. In no event will InBev or the surviving corporation be required to expend an annual premium amount in excess of 300% of the annual premiums currently paid by Anheuser-Busch for such insurance; if the annual premiums of such insurance coverage exceed that amount, the surviving corporation is required to obtain a policy with the greatest coverage available for a cost not exceeding such amount.

Following the effective time of the Merger, InBev and the surviving corporation will indemnify, defend and hold harmless each present and former director and officer of Anheuser-Busch or any of its subsidiaries and any fiduciary under any Anheuser-Busch benefit plan (the indemnified parties ), and will also promptly advance expenses as incurred, against any costs or expenses (including attorneys fees and disbursements), judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the indemnified party is or was an officer, director, employee or fiduciary of Anheuser-Busch or any of its subsidiaries or a fiduciary under any Anheuser-Busch benefit plan, whether asserted or claimed prior to, at or after the effective time of the Merger (including with respect to any acts or omissions in connection with the Merger Agreement and the transactions and actions contemplated thereby), to the fullest extent permitted under Delaware law and our certificate of incorporation or bylaws and any indemnification agreement in effect on the date of the Merger Agreement. Any determination required to be made with respect to whether an officer s or director s conduct complied with the standards set forth under Delaware law and our certificate of incorporation and bylaws will be made by independent counsel selected by the indemnified party. In the event of any claim, action, suit, proceeding or investigation, (x) neither InBev nor the surviving corporation may settle, compromise or consent to the entry of any judgment in any claim, action, suit, proceeding or investigation, unless such settlement, compromise or consent includes an unconditional release of the indemnified party from all liability arising out of such claim, action, suit, proceeding or investigation or the indemnified party otherwise consents, and (y) the surviving corporation will cooperate in the defense of such matter.

Additionally, the charter and bylaws of the surviving corporation must contain provisions no less favorable with respect to indemnification of and advancement of expenses to individuals who were directors and officers prior to the effective time of the Merger than those contained in our certificate of incorporation and bylaws as of the date of the Merger Agreement, and such provisions may not be amended, repealed or otherwise modified for a period of six years from the effective time of the Merger in any manner that would adversely affect the rights of any indemnified party.

The rights of the indemnified parties under the Merger Agreement are in addition to any rights such indemnified parties may have under the certificate of incorporation or bylaws of Anheuser-Busch or any of its subsidiaries, or under any applicable contracts or laws. InBev has agreed to honor and perform under all indemnification agreements entered into by Anheuser-Busch or any of its subsidiaries.

### **Employee Benefits**

From and after the effective time of the Merger, InBev will, and will cause the surviving corporation to, honor, in accordance with their terms, all benefit plans. In addition, InBev or the surviving corporation will, as applicable, pay the annual bonuses for the 2008 calendar year to employees of Anheuser-Busch and its subsidiaries who remain employed through December 31, 2008 or who are involuntarily terminated without cause, between the effective time of the Merger and December 31, 2008 (other than employees who are given a notice of termination prior to the effective time of the Merger), based on the performance for the 2008 year in accordance with Anheuser-Busch s practices and policies in effect on the date of the Merger Agreement.

Until the first anniversary of the effective time of the Merger or December 31, 2009, whichever is later, InBev shall, or shall cause the surviving corporation and each of its subsidiaries to, provide employees of Anheuser-Busch and its subsidiaries at the effective time of the Merger with compensation and benefits that

69

#### **Table of Contents**

are not less favorable in the aggregate than the compensation and benefits provided to current employees immediately prior to the effective time of the Merger.

InBev will cause the surviving corporation to cause service rendered by employees of Anheuser-Busch and its subsidiaries prior to the consummation of the Merger to be taken into account for vesting and eligibility purposes (but not for accrual purposes) under employee benefit plans of the surviving corporation and its subsidiaries, to the same extent as such service was taken into account under the corresponding benefit plans of Anheuser-Busch and its subsidiaries for those purposes. Current employees will not be subject to any pre-existing condition limitation under any health plan of the surviving corporation or its subsidiaries for any condition for which they would have been entitled to coverage under the corresponding benefit plan of Anheuser-Busch and its subsidiaries in which they participated prior to the effective time of the Merger. InBev will cause the surviving corporation and its subsidiaries to give such current employees credit under such plans for co-payments made and deductibles satisfied prior to the effective time of the Merger.

Until the first anniversary of the effective time of the Merger or December 31, 2009, whichever is later, InBev or the surviving corporation, or InBev s or the surviving corporation s subsidiaries, as applicable, will pay severance benefits to non-union employees of Anheuser-Busch and its Subsidiaries who are involuntarily terminated without cause during such period (other than those employees who are given a notice of termination prior to the effective time of the Merger) that are not less favorable than the severance benefits payable under Anheuser-Busch s Severance Pay Program as in effect immediately prior to the effective time of the Merger.

### Amendment, Extension and Waiver

To the extent permitted by applicable law, the parties may amend the Merger Agreement at any time, provided that, pursuant to Delaware law, after our stockholders have adopted the Merger Agreement, there shall be no amendment that decreases the merger consideration or which adversely affects the rights of our stockholders without the approval of our stockholders. The Merger Agreement may not be amended, changed, supplemented or otherwise modified except by a written agreement executed by all of the parties to the Merger Agreement.

At any time before the consummation of the Merger, each of the parties to the Merger Agreement may waive compliance with any of the agreements or conditions contained in the Merger Agreement to the extent permitted by applicable law.

### **Fees and Expenses**

Whether or not the Merger is consummated, all costs and expenses incurred in connection with the Merger Agreement and the Merger and the other transactions contemplated by the Merger Agreement will be paid by the party incurring such expense, except that (i) InBev will reimburse Anheuser-Busch for charges and expenses incurred by Anheuser-Busch in connection with its cooperation with InBev in InBev s efforts to obtain financing, and (ii) the party paying a termination fee under the terms of the Merger Agreement will reimburse the other party for charges and expenses incurred in connection with any suit brought to enforce payment of the termination fee.

### **VOTING AGREEMENT OF STICHTING INBEV AK**

In connection with the Merger Agreement, the controlling stockholder of InBev (holding voting power of, or having the right to direct the voting by certain affiliated parties or parties acting in concert of, in the aggregate, voting power of approximately 63.47% of InBev s outstanding common stock) has entered into a voting agreement with Anheuser-Busch pursuant to which such stockholder has agreed, among other things, to vote at an extraordinary shareholders meeting of InBev in favor of, among other things, InBev s acquisition of Anheuser-Busch. The voting

agreement will terminate upon the earlier of the effective time of the Merger and the termination of the Merger Agreement.

70

### ADJOURNMENT OF THE SPECIAL MEETING

### The Adjournment Proposal

If the number of shares of common stock present in person or represented by proxy at the special meeting voting in favor of the proposal to adopt the Merger Agreement is insufficient to adopt the Merger Agreement at the time of the special meeting, then we intend to move to adjourn the special meeting in order to enable our board of directors to solicit additional proxies in respect of such proposal. In that event, we will ask our stockholders to vote only upon the adjournment proposal, and not the proposal to adopt the Merger Agreement.

In this proposal, we are asking you to authorize the holder of any proxy solicited by our board of directors to vote in favor of granting discretionary authority to the proxy or attorney-in-fact to adjourn the special meeting for the purpose of soliciting additional proxies. If our stockholders approve the adjournment proposal, we could adjourn the special meeting and any adjourned session of the special meeting and use the additional time to solicit additional proxies, including the solicitation of proxies from stockholders that have previously returned properly executed proxies or authorized a proxy by using the Internet or a toll-free telephone number. Among other things, approval of the adjournment proposal could mean that, even if we had received proxies representing a sufficient number of votes against the adoption of the Merger such that the proposal to adopt the Merger Agreement would be defeated, we could adjourn the special meeting without a vote on the adoption of the Merger Agreement and seek to convince the holders of those shares to change their votes to votes in favor of adoption of the Merger Agreement. Additionally, we may seek to adjourn the special meeting if a quorum is not present at the special meeting.

### **Vote Required and Board Recommendation**

Approval of the proposal to adjourn the special meeting requires an affirmative vote of a majority of the votes cast that are entitled to vote at the special meeting, assuming a quorum is present. No proxy that is specifically marked AGAINST adoption of the Merger will be voted in favor of the adjournment proposal, unless it is specifically marked FOR the proposal to adjourn the special meeting.

Our board of directors believes that if the number of shares of common stock present in person or represented by proxy at the special meeting voting in favor of the proposal to adopt the Merger is not a sufficient number of shares to approve the proposal to adopt the Merger Agreement, it is in the best interests of Anheuser-Busch and its stockholders to enable our board of directors to continue to seek to obtain a sufficient number of additional votes in favor of approval of the proposal to adopt the Merger Agreement.

Our board of directors unanimously recommends that you vote FOR the proposal to adjourn the special meeting to a later date, if necessary or appropriate, to solicit additional proxies in favor of the proposal to adopt the Merger Agreement if there are insufficient votes at the time of the special meeting to adopt the Merger Agreement.

71

### MARKET PRICE OF COMMON STOCK

Our common stock is listed for trading on the NYSE under the symbol BUD . The following table sets forth, for the fiscal quarters indicated, the high and low sales prices per share as reported on the NYSE composite tape. In addition, the table also sets forth the quarterly cash dividends per share declared and paid by Anheuser-Busch with respect to its common stock.

	Common Stock		Quarterly Dividend	
	High	Low	DIV	aena
Year Ended December 31, 2006				
First Quarter	\$ 44.17	\$ 40.17	\$	0.270
Second Quarter	\$ 47.07	\$ 41.89	\$	0.270
Third Quarter	\$ 50.00	\$ 44.75	\$	0.295
Fourth Quarter	\$ 49.75	\$ 46.05	\$	0.295
Year Ended December 31, 2007				
First Quarter	\$ 51.74	\$ 47.95	\$	0.295
Second Quarter	\$ 55.19	\$ 49.18	\$	0.295
Third Quarter	\$ 52.50	\$ 47.38	\$	0.330
Fourth Quarter	\$ 53.91	\$ 48.16	\$	0.330
Year Ended December 31, 2008				
First Quarter	\$ 54.67	\$ 45.55	\$	0.330
Second Quarter	\$ 62.99	\$ 46.72	\$	0.330
Third Quarter (through , 2008)				

The closing sale price of our common stock on the New York Stock Exchange (the NYSE) on May 22, 2008, the last trading day before the initial market rumors surrounding a potential proposal by InBev to purchase Anheuser-Busch, was \$52.58. The closing sale price of our common stock on the NYSE on June 10, 2008, the last trading day before the announcement by InBev of its proposal to purchase Anheuser-Busch, was \$57.15. The closing sale price of our common stock on the NYSE on July 11, 2008, the last trading day prior to the announcement of the Merger, was \$66.50. On \$\, 2008, the most recent practicable date before this proxy statement was printed, the closing price for our common stock on the NYSE was \$\, \text{per share.} On July 23, 2008, our board of directors declared a quarterly dividend of \$0.37 to be paid on September 9, 2008 to stockholders of record as of August 11, 2008. You are encouraged to obtain current market quotations for our common stock in connection with voting your shares.

72

### SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table sets forth information concerning beneficial ownership of our common stock as of June 30, 2008 for: (a) each of Anheuser-Busch s directors, (b) each of Anheuser-Busch s current and former executive officers for whom disclosure is required; (c) Anheuser-Busch s directors and executive officers as a group; and (d) each beneficial holder of more than five percent of our common stock.

Except as otherwise noted in the footnotes below, each person or entity identified below has sole voting and investment power to such securities, and, unless otherwise indicated, the address of each person named in the table below is c/o Anheuser-Busch Companies, Inc., One Busch Place, St. Louis, Missouri 63118.

	Number of Shares	Percentage
	of Common Stock Beneficially	of
Name	Owned	class
Barclays Global Investors, NA and Affiliates	41,597,470(1)	5.78%
W. Randolph Baker	2,194,606(2)	*
Mark T. Bobak	864,392(3)	*
August A. Busch III	9,178,439(4)	1.27%
August A. Busch IV	2,680,831(5)	*
James J. Forese	33,313(6)	*
James R. Jones	42,797(7)(8)	*
Vernon R. Loucks, Jr.	42,883(7)	*
Vilma S. Martinez	41,899(9)	*
Douglas J. Muhleman	1,345,244(10)	*
Michael J. Owens	955,085(11)	*
William Porter Payne	44,449(7)	*
Joyce M. Roché	40,125(7)	*
Henry Hugh Shelton	39,573(12)	*
Patrick T. Stokes	6,811,751(13)	*
Andrew C. Taylor	84,617(7)	*
Douglas A. Warner III	53,001(7)	*
Edward E. Whitacre, Jr.	29,883(14)	*
All directors and executive officers as a group (30 persons)	31,526,776(15)	4.24%

<sup>\*</sup> less than 1% of the common stock outstanding

(1) This information is based on the Schedule 13G dated January 10, 2008 filed by Barclays Global Investors, NA and affiliates, 45 Fremont Street, San Francisco, CA 94105 with the Securities and Exchange Commission reporting on beneficial ownership as of December 31, 2007. In addition to Barclays Global Investors, NA, affiliates on the filing are Barclays Global Fund Advisors, Barclays Global Investors, LTD, Barclays Global Investors Japan Trust and Banking Company Limited, Barclays Global Investors Japan Limited, Barclays Global Investors Canada Limited, Barclays Global Investors Australia Limited, and Barclays Global Investors

(Deutschland) AG. According to the filing, the reporting persons have sole voting power with respect to 36,285,922 shares and sole investment power with respect to 41,597,470 shares.

- (2) The number of shares includes 1,877,309 shares that are subject to currently exercisable stock options, of which 245,737 are held in a family partnership, and 20,306 shares of unvested restricted stock.
- (3) Mr. Bobak retired from Anheuser-Busch on December 31, 2007. The number of shares includes 843,785 shares that are subject to currently exercisable stock options and 18,585 shares of unvested restricted stock.
- (4) The number of shares includes 4,630,129 shares that are subject to currently exercisable stock options, of which 100,000 are held in trusts for the benefit of children of Mr. Busch III, and 9,275 shares of unvested

73

#### **Table of Contents**

restricted stock. Of the shares shown, Mr. Busch III has shared voting and shared investment power as to 1,059,836 shares and 2,048,064 shares are held in trusts of which Mr. Busch III is income beneficiary and as to which he has certain rights, but as to which he has no voting or investment power. 85,348 shares beneficially owned by members of his immediate family are not included.

- (5) The number of shares includes 2,557,779 shares that are subject to currently exercisable stock options. Of those options, 50,000 were granted to Mr. Busch III and presently are held in trust for the benefit of Mr. Busch IV. Also included in the total are 67,847 shares of unvested restricted stock.
- (6) The number of shares includes 20,001 shares that are subject to currently exercisable stock options and 2.913 shares of unvested restricted stock.
- (7) The number of shares includes 38,001 shares that are subject to currently exercisable stock options and 499 shares of unvested restricted stock. 3,383 of the shares held by Mr. Loucks have been pledged as security.
- (8) Mr. Jones has shared voting and shared investment power with respect to 2,256 of these shares.
- (9) The number of shares includes 38,001 shares that are subject to current exercisable stock options and 2,913 shares of unvested restricted stock.
- (10) The number of shares includes 1,300,701 shares that are subject to currently exercisable stock options and 16,815 shares of unvested restricted stock.
- (11) The number of shares includes 892,624 shares that are subject to currently exercisable stock options and 14,628 shares of unvested restricted stock.
- (12) The number of shares includes 30,001 shares that are subject to currently exercisable stock options and 499 shares of unvested restricted stock.
- (13) The number of shares includes 6,405,126 shares that are subject to currently exercisable stock options (of which 1,200,833 are held in a family partnership), 351,252 shares that are held in a family partnership for which Mr. Stokes wife has shared voting and shared investment power, and 15,645 shares that are held in a trust in which Mr. Stokes and his wife have an economic interest, but as to which they have no voting or investment power. Also included are 13,446 shares of unvested restricted stock.
- (14) The number of shares includes 25,001 shares that are subject to currently exercisable stock options and 2,913 shares of unvested restricted stock.
- (15) The number of shares stated includes 25,403,814 shares that are subject to currently exercisable stock options, 261,688 shares of unvested restricted stock, 2,048,064 of the shares that are referred to in Note 4, as to which Mr. Busch III has no voting or investment power, and 366,897 of the shares that are referred to in Note 13 for which Mr. Stokes has no voting or investment power. 3,383 of the shares are pledged as security. The directors and executive officers as a group have sole voting and sole investment power as to 2,645,880 shares and shared voting and shared investment power as to 1,062,121 shares. 98,259 shares held by immediate family members or family trusts are not included and beneficial ownership of such shares is disclaimed.

74

#### DISSENTERS RIGHTS OF APPRAISAL

Under the General Corporation Law of the State of Delaware (the DGCL), you have the right to dissent from the Merger and to receive payment in cash for the fair value of your common stock as determined by the Delaware Court of Chancery, together with a fair rate of interest, if any, as determined by the court, in lieu of the consideration you would otherwise be entitled to receive pursuant to the Merger Agreement. These rights are known as appraisal rights. Anheuser-Busch s stockholders electing to exercise appraisal rights must comply with the provisions of Section 262 of the DGCL in order to perfect their rights. Anheuser-Busch will require strict compliance with the statutory procedures.

The following is intended as a brief summary of the material provisions of the Delaware statutory procedures required to be followed by a stockholder in order to dissent from the Merger and perfect appraisal rights.

This summary, however, is not a complete statement of all applicable requirements and is qualified in its entirety by reference to Section 262 of the DGCL, the full text of which appears in Annex D to this proxy statement. Failure to precisely follow any of the statutory procedures set forth in Section 262 of the DGCL may result in a termination or waiver of your appraisal rights.

Section 262 requires that stockholders be notified that appraisal rights will be available not less than 20 days before the stockholders meeting to vote on the Merger. A copy of Section 262 must be included with such notice. This proxy statement constitutes Anheuser-Busch s notice to its stockholders of the availability of appraisal rights in connection with the Merger in compliance with the requirements of Section 262. If you wish to consider exercising your appraisal rights, you should carefully review the text of Section 262 contained in Annex D since failure to timely and properly comply with the requirements of Section 262 will result in the loss of your appraisal rights under the DGCL.

If you elect to demand appraisal of your shares, you must satisfy each of the following conditions:

You must deliver to Anheuser-Busch a written demand for appraisal of your shares before the vote with respect to the Merger is taken. This written demand for appraisal must be in addition to and separate from any proxy or vote abstaining from or voting against the adoption of the Merger Agreement. Voting against or failing to vote for the adoption of the Merger Agreement by itself does not constitute a demand for appraisal within the meaning of Section 262.

You must not vote in favor of the adoption of the Merger Agreement. A vote in favor of the adoption of the Merger Agreement, by proxy, over the Internet, by telephone or in person, will constitute a waiver of your appraisal rights in respect of the shares so voted and will nullify any previously filed written demands for appraisal. If you fail to comply with either of these conditions and the Merger is completed, you will be entitled to receive the cash payment for your shares of common stock as provided for in the Merger Agreement, but you will have no appraisal rights with respect to your shares of common stock.

All demands for appraisal should be addressed to Anheuser-Busch Companies, Inc., One Busch Place, St. Louis, Missouri 63118, Attention: Vice President and Secretary, and must be delivered before the vote on the Merger Agreement is taken at the special meeting, and should be executed by, or on behalf of, the record holder of the shares of our common stock. The demand must reasonably inform Anheuser-Busch of the identity of the stockholder and the intention of the stockholder to demand appraisal of his, her or its shares.

To be effective, a demand for appraisal by a holder of our common stock must be made by, or in the name of, such registered stockholder, fully and correctly, as the stockholder is name appears on his or her stock certificate(s). Beneficial owners who do not also hold the shares of record may not directly make appraisal demands to Anheuser-Busch. The beneficial holder must, in such cases, have the registered owner, such as a broker or other nominee, submit the required demand in respect of those shares. If shares are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of a demand for appraisal should be made by or for the fiduciary; and if the shares are owned of record by more than one person, as in a joint tenancy or tenancy in common, the demand should be executed by or for all

75

#### **Table of Contents**

joint owners. An authorized agent, including an authorized agent for two or more joint owners, may execute the demand for appraisal for a stockholder of record; however, the agent must identify the record owner or owners and expressly disclose the fact that, in executing the demand, he or she is acting as agent for the record owner. A record owner, such as a broker, who holds shares as a nominee for others, may exercise his or her right of appraisal with respect to the shares held for one or more beneficial owners, while not exercising this right for other beneficial owners. In that case, the written demand should state the number of shares as to which appraisal is sought. Where no number of shares is expressly mentioned, the demand will be presumed to cover all shares held in the name of the record owner.

If you hold your shares of common stock in a brokerage account or in other nominee form and you wish to exercise appraisal rights, you should consult with your broker or the other nominee to determine the appropriate procedures for the making of a demand for appraisal by the nominee.

Within 10 days after the effective time of the Merger, the surviving corporation must give written notice that the Merger has become effective to each Anheuser-Busch stockholder who has properly filed a written demand for appraisal and who did not vote in favor of the Merger Agreement. At any time within 60 days after the effective time, any stockholder who has demanded an appraisal has the right to withdraw the demand and to accept the cash payment specified by the Merger Agreement for his or her shares of common stock. Within 120 days after the effective date of the Merger, any stockholder who has complied with Section 262 shall, upon written request to the surviving corporation, be entitled to receive a written statement setting forth the aggregate number of shares not voted in favor of the Merger Agreement and with respect to which demands for appraisal rights have been received and the aggregate number of holders of such shares. Such written statement will be mailed to the requesting stockholder within 10 days after such written request is received by the surviving corporation or within 10 days after expiration of the period for delivery of demands for appraisal, whichever is later. Within 120 days after the effective time, either the surviving corporation or any stockholder who has complied with the requirements of Section 262 may file a petition in the Delaware Court of Chancery demanding a determination of the fair value of the shares held by all stockholders entitled to appraisal. Upon the filing of the petition by a stockholder, service of a copy of such petition shall be made upon the surviving corporation. The surviving corporation has no obligation to file such a petition in the event there are dissenting stockholders. Accordingly, the failure of a stockholder to file such a petition within the period specified could nullify the stockholder s previously written demand for appraisal.

If a petition for appraisal is duly filed by a stockholder and a copy of the petition is delivered to the surviving corporation, the surviving corporation will then be obligated, within 20 days after receiving service of a copy of the petition, to provide the Court of Chancery with a duly verified list containing the names and addresses of all stockholders who have demanded an appraisal of their shares and with whom agreements as to the value of their shares have not been reached by the surviving corporation. After notice to dissenting stockholders who demanded appraisal of their shares, the Court of Chancery is empowered to conduct a hearing upon the petition, and to determine those stockholders who have complied with Section 262 and who have become entitled to the appraisal rights provided thereby.

After determination of the stockholders entitled to appraisal of their shares of our common stock, the Court of Chancery will appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the Merger, together with a fair rate of interest, if any. When the value is determined, the Court of Chancery will direct the payment of such value, with interest thereon accrued during the pendency of the proceeding, if the Court of Chancery so determines, to the stockholders entitled to receive the same, upon surrender by such holders of those shares. Unless the Court of Chancery in its discretion determines otherwise for good cause shown, interest from the effective date of the Merger through the date of payment of the judgment shall be compounded quarterly and shall accrue at 5% over the Federal Reserve discount rate (including any surcharge) as established from time to time during the period between the effective date of the Merger and the date of payment of

the judgment.

In determining fair value, the Court of Chancery is required to take into account all relevant factors. You should be aware that the fair value of your shares as determined under Section 262 could be more than,

76

#### **Table of Contents**

the same as, or less than the value that you are entitled to receive under the terms of the Merger Agreement.

Costs of the appraisal proceeding may be imposed upon the surviving corporation and the stockholders participating in the appraisal proceeding by the Court of Chancery as the Court of Chancery deems equitable in the circumstances. Upon the application of a stockholder, the Court of Chancery may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorneys fees and the fees and expenses of experts, to be charged pro rata against the value of all shares entitled to appraisal. Any stockholder who had demanded appraisal rights will not, after the effective time of the Merger, be entitled to vote shares subject to that demand for any purpose or to receive payments of dividends or any other distribution with respect to those shares, other than with respect to payment as of a record date prior to the effective time of the Merger; however, if no petition for appraisal is filed within 120 days after the effective time of the Merger, or if the stockholder delivers a written withdrawal of his or her demand for appraisal and an acceptance of the terms of the Merger within 60 days after the effective time of the Merger, then the right of that stockholder to appraisal will cease and that stockholder will be entitled to receive the cash payment for shares of his, her or its common stock pursuant to the Merger Agreement. Any withdrawal of a demand for appraisal made more than 60 days after the effective time of the Merger may only be made with the written approval of the surviving corporation and must, to be effective, be made within 120 days after the effective time.

In view of the complexity of Section 262, Anheuser-Busch s stockholders who may wish to dissent from the Merger and pursue appraisal rights should consult their legal advisors.

#### SUBMISSION OF STOCKHOLDER PROPOSALS

If the Merger is consummated, we will not have public stockholders and there will be no public participation in any future meeting of stockholders. However, if the Merger is not completed or if we are otherwise required to do so under applicable law, we would hold a 2009 annual meeting of stockholders. We had previously set the deadline for submitting proposals for inclusion in Anheuser-Busch s proxy statement relating to the 2009 annual meeting to be November 10, 2008. Any stockholder proposal received after November 10, 2008 will not be considered timely for inclusion in the proxy materials. Under our bylaws, in order for a stockholder proposal submitted outside of Rule 14a-8, and therefore not included in our proxy materials, to be considered timely, such proposal must be received by our Corporate Secretary not later than the tenth business day following the date on which notice of the date of the 2009 annual meeting is mailed to stockholders or we otherwise make public disclosure of the date of such annual meeting, whichever occurs first.

#### HOUSEHOLDING OF SPECIAL MEETING MATERIALS

Some banks, brokers, and other nominee record holders may be participating in the practice of householding proxy statements and annual reports. This means that only one copy of this notice and proxy statement may have been sent to multiple stockholders in your household unless we have received contrary instructions. If you would prefer to receive separate copies of a proxy statement or annual report either now or in the future, please contact BNY Mellon Shareowner Services, P.O. Box 358015, Pittsburgh, PA 15252, attention: Shareholder Correspondence. Upon written or oral BNY Mellon Shareowner Services, we will provide a separate copy of the annual reports and proxy statements. In addition, security holders sharing an address can request delivery of a single copy of annual reports or proxy statements if you are receiving multiple copies upon written or oral request to BNY Mellon Shareowner Services at the address stated above.

### **OTHER MATTERS**

At this time, we know of no other matters to be submitted to our stockholders at the special meeting or any adjournment or postponement of the special meeting. If any other matters properly come before the special meeting or any adjournment or postponement of the special meeting, it is the intention of the persons named in the enclosed proxy card to vote the shares they represent in accordance with their best judgment.

77

#### WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any document we file at the SEC s public reference room located at 100 F Street, N.E., Room 1580, Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the public reference room. Our SEC filings are also available to the public at the SEC s website at http://www.sec.gov. You also may obtain free copies of the documents Anheuser-Busch files with the SEC by going to the Investors section of our website at www.anheuser-busch.com/InformationRequest. Our website address is provided as an inactive textual reference only. The information provided on our website is not part of this proxy statement, and therefore is not incorporated by reference.

Reports, proxy statements or other information concerning us may also be inspected at the offices of the NYSE at 20 Broad Street, New York, New York 10005.

Statements contained in this proxy statement, or in any document incorporated in this proxy statement by reference regarding the contents of any contract or other document, are not necessarily complete and each such statement is qualified in its entirety by reference to that contract or other document filed as an exhibit with the SEC. The SEC allows us to incorporate by reference into this proxy statement documents we file with the SEC. This means that we can disclose important information to you by referring you to those documents. The information incorporated by reference is considered to be a part of this proxy statement, and later information that we file with the SEC will update and supersede that information. We incorporate by reference the documents listed below and any documents filed by us pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this proxy statement and before the date of the special meeting:

Annual Report on Form 10-K filed on March 29, 2008, and the amendment thereto filed April 25, 2008;

Quarterly Reports on Form 10-Q for the quarterly periods ended March 31, 2008 and June 30, 2008; and

Current Reports on Form 8-K filed on June 23, 2008, June 26, 2008, June 27, 2008, June 30, 2008, July 14, 2008, July 16, 2008 and August 12, 2008.

Notwithstanding the foregoing, information furnished under Items 2.02 and 7.01 of any Current Report on Form 8-K, including the related exhibits, is not incorporated by reference in this proxy statement.

Any person, including any beneficial owner, to whom this proxy statement is delivered may request copies of proxy statements and any of the documents incorporated by reference in this document or other information concerning us, without charge, by written or telephonic request directed to Anheuser-Busch Investor Relations, One Busch Place, St. Louis, Missouri 63118, telephone: (800) 342-5283, on Anheuser-Busch s website at www.anheuser-busch.com/informationrequest.com or from the SEC through the SEC s website at the address provided above. Documents incorporated by reference are available without charge, excluding any exhibits to those documents unless the exhibit is specifically incorporated by reference into those documents.

THIS PROXY STATEMENT DOES NOT CONSTITUTE THE SOLICITATION OF A PROXY IN ANY JURISDICTION TO OR FROM ANY PERSON TO WHOM OR FROM WHOM IT IS UNLAWFUL TO MAKE SUCH PROXY SOLICITATION IN THAT JURISDICTION. YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED OR INCORPORATED BY REFERENCE IN THIS PROXY STATEMENT TO VOTE YOUR SHARES AT THE SPECIAL MEETING. WE HAVE NOT AUTHORIZED ANYONE TO PROVIDE

YOU WITH INFORMATION THAT IS DIFFERENT FROM WHAT IS CONTAINED IN THIS PROXY STATEMENT. THIS PROXY STATEMENT IS DATED , 2008. YOU SHOULD NOT ASSUME THAT THE INFORMATION CONTAINED IN THIS PROXY STATEMENT IS ACCURATE AS OF ANY DATE OTHER THAN THAT DATE, AND THE MAILING OF THIS PROXY STATEMENT TO STOCKHOLDERS DOES NOT CREATE ANY IMPLICATION TO THE CONTRARY.

78

Annex A

AGREEMENT AND PLAN OF MERGER
By and Among
ANHEUSER-BUSCH COMPANIES, INC.,
INBEV N.V./S.A.
and
PESTALOZZI ACQUISITION CORP.
Dated as of July 13, 2008

A-1

### **Table of Contents**

### **TABLE OF CONTENTS**

		<u>Page</u>
	Article I	
	The Merger; Closing; Effective Time	
1.1.	The Merger	A-4
1.2.	Closing	A-4
1.3.	Effective Time	A-4
	Article II	
	Certificate of Incorporation and By-Laws of the Surviving Corporation	
2.1.	The Certificate of Incorporation	A-5
2.2.	The By-Laws	A-5
	Article III	
	Directors and Officers of the Surviving Corporation	
3.1.	Directors	A-5
3.2.	Officers	A-5
	Article IV	
	Effect of the Merger on Capital Stock; Exchange of Certificates	
4.1.	Effect on Capital Stock	A-5
4.2.	Exchange of Certificates	A-6
4.3.	Treatment of Stock Plans	A-7
4.4.	Adjustments to Prevent Dilution	A-8
	Article V	
	Representations and Warranties	
5.1.	Representations and Warranties of the Company	A-8
5.2.	Representations and Warranties of Parent and Merger Sub	A-20
	Article VI	
	Covenants	
6.1.	Interim Operations	A-23
6.2.	Acquisition Proposals	A-25
6.3.	Proxy Filing; Information Supplied	A-28
6.4.	Stockholders Meeting	A-28
6.5.	Filings; Other Actions; Notification	A-29
6.6.	Access and Reports	A-31
6.7.	Stock Exchange Delisting	A-31
6.8.	Publicity	A-31
6.9.	Employee Benefits	A-31
6.10.	Debt and Equity Financing	A-33
6.11.	Expenses	A-34
6.12.	Indemnification; Directors and Officers Insurance	A-34
6.13.	Takeover Statutes	A-35
6.14.	Communications	A-36

6.15.	Release and Termination of Litigation	A-36
6.16.	Retention Plan	A-36
6.17.	Other Matters	A-36
	A-2	

## **Table of Contents**

		Page
	Article VII	
	Conditions	
7.1.	Conditions to Each Party s Obligation to Effect the Merger	A-37
7.2.	Conditions to Obligations of Parent and Merger Sub	A-37
7.3.	Conditions to Obligation of the Company	A-38
	Article VIII	
	Termination	
8.1.	Termination by Mutual Consent	A-38
8.2.	Termination by Either Parent or the Company	A-38
8.3.	Termination by the Company	A-38
8.4.	Termination by Parent	A-39
8.5.	Effect of Termination and Abandonment	A-39
	Article IX	
	Miscellaneous and General	
9.1.	Survival	A-40
9.2.	Modification or Amendment	A-41
9.3.	Waiver of Conditions	A-41
9.4.	Counterparts	A-41
9.5.	GOVERNING LAW AND VENUE; WAIVER OF JURY TRIAL; SPECIFIC	
	PERFORMANCE	A-41
9.6.	Notices	A-42
9.7.	Entire Agreement	A-43
9.8.	No Third Party Beneficiaries	A-43
9.9.	Obligations of Parent and of the Company	A-43
9.10.	Transfer Taxes	A-44
9.11.	Definitions	A-44
9.12.	Severability	A-44
9.13.	Interpretation; Construction	A-44
9.14.	Assignment	A-44
Annex A	Defined Terms	A-46
Annex B	Charter	B-1
	A-3	

#### **Table of Contents**

#### AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (hereinafter called this <u>Agreement</u>), dated as of July 13, 2008, by and among Anheuser-Busch Companies, Inc., a Delaware corporation (the <u>Company</u>), InBev N.V./S.A., a public company organized under the laws of Belgium (<u>Parent</u>), and Pestalozzi Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of Parent (<u>Merger Sub</u>, the Company and Merger Sub sometimes being hereinafter collectively referred to as the <u>Constituent Corporations</u>).

#### RECITALS

WHEREAS, the respective boards of directors of each of Parent, Merger Sub and the Company have approved the merger of Merger Sub with and into the Company (the <u>Merger</u>) upon the terms and subject to the conditions set forth in this Agreement and have approved and declared advisable this Agreement;

WHEREAS, the board of directors of the Company has determined that the Merger and the other transactions contemplated by this Agreement are in the best interests of the Company s stockholders and has resolved to recommend the approval of the Merger to the Company s stockholders;

WHEREAS, the board of directors of Parent has determined that the Merger and the other transactions contemplated by this Agreement are in the best interests of Parent;

WHEREAS, contemporaneously with the execution and delivery of this Agreement, and in order to induce the Company to enter into this Agreement, Parent's controlling shareholder Stichting InBev AK (the Stichting), has delivered to the Company, a letter agreement (the Parent Shareholder Commitment) addressed to the board of directors of the Company with respect to certain commitments made by such shareholder in connection with this Agreement and the transactions contemplated by this Agreement; and

WHEREAS, the Company, Parent and Merger Sub desire to make certain representations, warranties, covenants and agreements in connection with this Agreement;

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

#### ARTICLE I

#### The Merger; Closing; Effective Time

- 1.1. <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the Effective Time (as defined in Section 1.3), Merger Sub shall be merged with and into the Company and the separate corporate existence of Merger Sub shall thereupon cease. The Company shall be the surviving corporation in the Merger (sometimes hereinafter referred to as the <u>Surviving Corporation</u>), and the separate corporate existence of the Company, with all its rights, privileges, immunities, powers and franchises, shall continue unaffected by the Merger, except as set forth in Article II. The Merger shall have the effects specified in the Delaware General Corporation Law, as amended (the <u>DGCL</u>).
- 1.2. <u>Closing</u>. Unless otherwise mutually agreed in writing between the Company and Parent, the closing for the Merger (the <u>Closing</u>) shall take place at the offices of Sullivan & Cromwell LLP, 125 Broad Street, New York, New York, at 9:00 A.M. (Eastern Time) on the third business day (the <u>Closing Date</u>) following the day on which the last to be satisfied or waived of the conditions set forth in Article VII (other than those conditions that by their nature are to

be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions) shall be satisfied or waived in accordance with this Agreement. For purposes of this Agreement, the term **business day** shall mean any day ending at 11:59 P.M. (Eastern Time) other than a Saturday or Sunday or a day on which banks are required or authorized to close in New York, New York or London, England.

1.3. *Effective Time*. As soon as practicable following the Closing, the Company and Parent will cause a certificate of merger (the **Delaware Certificate of Merger** ) to be executed, acknowledged and filed with

A-4

#### **Table of Contents**

the Secretary of State of the State of Delaware as provided in Section 251 of the DGCL. The Merger shall become effective at the time when the Delaware Certificate of Merger has been duly filed with the Secretary of State of the State of Delaware or at such later time as may be agreed by the parties in writing and specified in the Delaware Certificate of Merger (the **\_Effective Time** ).

#### **ARTICLE II**

## Certificate of Incorporation and By-Laws of the Surviving Corporation

- 2.1. <u>The Certificate of Incorporation</u>. As of the Effective Time, the certificate of incorporation of the Surviving Corporation (the <u>Charter</u>) shall be amended in its entirety to read as set forth in Annex B which Parent shall provide to the Company prior to the Closing and which shall be reasonably satisfactory to the Company, until thereafter amended (subject to Section 6.12(b)) as provided therein or by applicable Law.
- 2.2. <u>The By-Laws</u>. The parties hereto shall take all actions necessary so that the by-laws of Merger Sub in effect immediately prior to the Effective Time shall be the by-laws of the Surviving Corporation (the <u>By-Laws</u>), until thereafter amended (subject to Section 6.12(b)) as provided therein or by applicable Law.

#### **ARTICLE III**

## Directors and Officers of the Surviving Corporation

- 3.1. <u>Directors</u>. The parties hereto shall take all actions necessary so that the board of directors of Merger Sub at the Effective Time shall, from and after the Effective Time, be the directors of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws.
- 3.2. <u>Officers</u>. The parties hereto shall take all actions necessary so that the officers of the Company at the Effective Time shall, from and after the Effective Time, be the officers of the Surviving Corporation until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Charter and the By-Laws.

#### ARTICLE IV

## Effect of the Merger on Capital Stock; Exchange of Certificates

- 4.1. <u>Effect on Capital Stock</u>. At the Effective Time, as a result of the Merger and without any action on the part of the holder of any capital stock of the Company:
- (a) <u>Merger Consideration</u>. Each share of common stock, par value \$1.00 per share, of the Company (a <u>Share</u> or, collectively, the <u>Shares</u>) issued and outstanding immediately prior to the Effective Time other than (i) Shares owned by Parent, Merger Sub or any other direct or indirect wholly owned subsidiary of Parent and Shares owned by the Company or any direct or indirect wholly owned subsidiary of the Company, and in each case not held on behalf of third parties, and (ii) Shares that are owned by stockholders (<u>Dissenting Stockholders</u>) who have perfected and not withdrawn a demand for appraisal rights pursuant to Section 262 of the DGCL (each of the Shares described in clauses (i) and (ii), an <u>Excluded Share</u> and, collectively, the <u>Excluded Shares</u>) shall be converted into the right to receive \$70.00 in cash per Share (the <u>Per Share Merger Consideration</u>). At the Effective Time, all of the Shares (other than Excluded Shares) shall cease to be outstanding, shall be cancelled and shall cease to exist, and each

certificate (a <u>Certificate</u>) formerly representing any of the Shares or non-certificated Shares represented by book-entry (<u>Book-Entry Shares</u>) (other than Excluded Shares) shall thereafter represent only the right to receive the Per Share Merger Consideration, without interest.

(b) <u>Cancellation of Excluded Shares</u>. Each Excluded Share shall, by virtue of the Merger and without any action on the part of the holder of the Excluded Share, cease to be outstanding, be cancelled

A-5

#### **Table of Contents**

without payment of any consideration therefor and shall cease to exist, subject to any rights the holder thereof may have under Section 4.2(f), except for Excluded Shares owned by Parent, the Company or any direct or indirect wholly owned subsidiary of Parent or the Company, and in each case not held on behalf of third parties, which shall remain outstanding and unaffected by the Merger.

(c) <u>Merger Sub</u>. At the Effective Time, each share of common stock, par value \$0.01 per share, of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one share of common stock, par value \$0.01 per share, of the Surviving Corporation.

## 4.2. Exchange of Certificates.

- (a) <u>Paying Agent</u>. Simultaneously with the Effective Time, Parent shall deposit, or shall cause to be deposited, with a paying agent selected by Parent with the Company s prior approval (such approval not to be unreasonably withheld or delayed) (the <u>Paying Agent</u>), for the benefit of the holders of Shares, a cash amount in immediately available funds sufficient for the Paying Agent to make payments (i) under Section 4.1(a) and (ii) in respect of Company Options pursuant to Section 4.3(a) and Company Awards pursuant to Section 4.3(b) (such aggregate cash amount in clauses (i) and (ii) being hereinafter referred to as the <u>Exchange Fund</u>). The Paying Agent shall invest the Exchange Fund as directed by Parent; <u>provided</u> that such investments shall be in obligations of or guaranteed by the United States of America in commercial paper obligations rated A-1 or P-1 or better by Moody s Investors Service, Inc. or Standard & Poor s Corporation, respectively, in certificates of deposit, bank repurchase agreements or banker s acceptances of commercial banks with capital exceeding \$1 billion, or in money market funds having a rating in the highest investment category granted by a recognized credit rating agency at the time of acquisition or a combination of the foregoing and, in any such case, no such instrument shall have a maturity exceeding three months. Any interest and other income resulting from such investments shall become a part of the Exchange Fund, and any amounts in excess of the amounts payable under Section 4.1(a) shall be promptly returned to Parent.
- (b) <u>Exchange Procedures</u>. Promptly after the Effective Time (and in any event within (x) five business days in the case of record holders and (y) three business days in the case of the Depository Trust Company (\_DTC ) on behalf of street holders), the Surviving Corporation shall cause the Paying Agent to mail (or in the case of DTC, deliver) to each holder of record of Shares (other than holders of Excluded Shares) (i) a letter of transmittal in customary form specifying that delivery shall be effected, and risk of loss and title to the Certificates or Book-Entry Shares shall pass, only upon delivery of the Certificates (or affidavits of loss in lieu of the Certificates as provided in Section 4.2(e)) or Book-Entry Shares to the Paying Agent, such letter of transmittal to be in such form and have such other provisions as Parent and the Company may reasonably agree, and (ii) instructions for use in effecting the surrender of the Certificates (or affidavits of loss in lieu of the Certificates as provided in Section 4.2(e)) or Book-Entry Shares in exchange for the Per Share Merger Consideration. Upon surrender of a Certificate (or affidavit of loss in lieu of the Certificate as provided in Section 4.2(e)) or Book-Entry Shares to the Paying Agent in accordance with the terms of such letter of transmittal, duly executed, the holder of such Certificate or Book-Entry Shares shall be entitled to receive in exchange therefor a cash amount in immediately available funds (after giving effect to any required tax withholdings as provided in Section 4.2(g)) equal to (x) the number of Shares represented by such Certificate (or affidavit of loss in lieu of the Certificate as provided in Section 4.2(e)) or Book-Entry Shares multiplied by (y) the Per Share Merger Consideration, and the Certificate or Book-Entry Shares so surrendered shall forthwith be cancelled. No interest will be paid or accrued on any amount payable upon due surrender of the Certificates or Book-Entry Shares. In the event of a transfer of ownership of Shares that is not registered in the transfer records of the Company, a check for any cash to be exchanged upon due surrender of the Certificate or Book-Entry Shares may be issued to such transferee if the Certificate or Book-Entry Shares formerly representing such Shares is presented to the Paying Agent, accompanied by all documents reasonably required to evidence and effect such transfer and to evidence that any applicable stock transfer taxes have been paid or are not applicable.

(c) <u>Transfers</u>. From and after the Effective Time, there shall be no transfers on the stock transfer books of the Company of the Shares that were outstanding immediately prior to the Effective Time. If, after the Effective Time, any Certificate or Book-Entry Shares are presented to the Surviving Corporation, Parent or the

A-6

#### **Table of Contents**

Paying Agent for transfer, such Certificates or Book-Entry Shares shall be cancelled and exchanged for the cash amount in immediately available funds to which the holder of the Certificate or Book-Entry Shares is entitled pursuant to this Article IV.

- (d) <u>Termination of Exchange Fund</u>. Any portion of the Exchange Fund (including the proceeds of any investments of the Exchange Fund) that remains unclaimed by the stockholders of the Company for nine (9) months after the Effective Time shall be delivered to the Surviving Corporation. Any holder of Shares (other than Excluded Shares) who has not theretofore complied with this Article IV shall thereafter look only to the Surviving Corporation for payment of the Per Share Merger Consideration (after giving effect to any required tax withholdings as provided in Section 4.2(g)) upon due surrender of its Certificates (or affidavits of loss in lieu of the Certificates) or Book-Entry Shares, without any interest thereon. Notwithstanding the foregoing, none of the Surviving Corporation, Parent, the Paying Agent or any other Person shall be liable to any former holder of Shares for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar Laws. For the purposes of this Agreement, the term <u>Person</u> shall mean any individual, corporation (including not-for-profit), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, Governmental Entity (as defined in Section 5.1(d)(i)) or other entity of any kind or nature.
- (e) <u>Lost, Stolen or Destroyed Certificates</u>. In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent, the posting by such Person of a bond in customary amount and upon such terms as may be required by Parent as indemnity against any claim that may be made against it or the Surviving Corporation with respect to such Certificate, the Paying Agent will issue a check in the amount (after giving effect to any required tax withholdings) equal to the number of Shares represented by such lost, stolen or destroyed Certificate multiplied by the Per Share Merger Consideration.
- (f) <u>Appraisal Rights</u>. No Person who has perfected a demand for appraisal rights pursuant to Section 262 of the DGCL shall be entitled to receive the Per Share Merger Consideration with respect to the Shares owned by such Person unless and until such Person shall have effectively withdrawn or lost such Person s right to appraisal under the DGCL. Each Dissenting Stockholder shall be entitled to receive only the payment provided by Section 262 of the DGCL with respect to Shares owned by such Dissenting Stockholder. The Company shall give Parent (i) prompt notice of any written demands for appraisal, attempted withdrawals of such demands, and any other instruments served pursuant to applicable Law that are received by the Company relating to stockholders—rights of appraisal and (ii) the opportunity to direct all negotiations and proceedings with respect to demand for appraisal under the DGCL. The Company shall not, except with the prior written consent of Parent, voluntarily make any payment with respect to any demands for appraisal, offer to settle or settle any such demands or approve any withdrawal of any such demands.
- (g) <u>Withholding Rights</u>. Each of Parent, the Surviving Corporation and the Paying Agent shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of Shares such amounts as it reasonably determines in good faith it is required to deduct and withhold with respect to the making of such payment under the Internal Revenue Code of 1986, as amended (the <u>Code</u>), or any other applicable state, local or foreign Tax (as defined in Section 5.1(n)) Law. To the extent that amounts are so withheld by Parent, the Surviving Corporation or the Paying Agent, as the case may be, such withheld amounts (i) shall be remitted by Parent, the Surviving Corporation or the Paying Agent, as applicable, to the applicable Governmental Entity, and (ii) shall be treated for all purposes of this Agreement as having been paid to the holder of Shares in respect of which such deduction and withholding was made by Parent, the Surviving Corporation or the Paying Agent, as the case may be.

#### 4.3. Treatment of Stock Plans.

(a) <u>Treatment of Options</u>. At the Effective Time, each outstanding option to purchase Shares (a <u>Company Option</u>) under the Stock Plans (as defined in Section 5.1(b)(i)), vested or unvested, shall be cancelled and converted into the right to receive, as soon as reasonably practicable (and in no event later than three (3) business days) after the Effective Time, an amount in cash equal to the product of (x) the total number of Shares subject to the Company Option times (y) the excess, if any, of the Per Share Merger

A-7

#### **Table of Contents**

Consideration over the exercise price per Share under such Company Option, less applicable Taxes required to be withheld with respect to such payment.

- (b) <u>Company Awards</u>. At the Effective Time, each right of any kind, contingent or accrued, to acquire or receive Shares or benefits measured by the value of Shares, and each award of any kind consisting of Shares that may be held, awarded, outstanding, payable or reserved for issuance under the Stock Plans and any other Company Benefit Plans (as defined in Section 5.1(h)(i)), other than Company Options (the <u>Company Awards</u>), vested or unvested, shall be cancelled and converted into the right to receive, as soon as reasonably practicable (but in no event later than three (3) business days) after the Effective Time, an amount in cash equal to (x) the number of Shares subject to such Company Awards immediately prior to the Effective Time times (y) the Per Share Merger Consideration (or, if the Company Award provides for payments to the extent the value of the Shares exceed a specified reference price, the amount, if any, by which the Per Share Merger Consideration exceeds such reference price), less applicable Taxes required to be withheld with respect to such payment.
- (c) <u>Termination of Plan</u>. The Company s Global Employee Stock Purchase Plan shall terminate as of immediately prior to the Effective Time and no new share offering date shall commence after the date hereof. Any amounts withheld and contributed to a savings account on behalf of a participating employee of the Company or its Subsidiaries and held in such savings account immediately prior to the Effective Time shall be refunded to such participating employee as soon as reasonably practicable (but in no event later than three (3) business days).
- (d) <u>Corporate Actions</u>. At or prior to the Effective Time, the Company, the board of directors of the Company or the compensation committee of the board of directors of the Company, as applicable, shall adopt any resolutions and take any actions which are reasonably necessary to effectuate the provisions of Sections 4.3(a), 4.3(b) and 4.3(c). The Company shall take all reasonable actions necessary to ensure that from and after the Effective Time neither Parent nor the Surviving Corporation will be required to deliver Shares or other capital stock of the Company to any Person pursuant to or in settlement of Company Options or Company Awards or otherwise.
- 4.4. <u>Adjustments to Prevent Dilution</u>. In the event that the Company changes the number of Shares or securities convertible or exchangeable into or exercisable for Shares issued and outstanding prior to the Effective Time as a result of a reclassification, stock split (including a reverse stock split), stock dividend or distribution, recapitalization, merger, issuer tender or exchange offer, or other similar transaction, the Per Share Merger Consideration shall be equitably adjusted.

## **ARTICLE V**

#### Representations and Warranties

- 5.1. <u>Representations and Warranties of the Company</u>. Except as set forth in the Company Reports (as defined in Section 5.1(e)(i)) filed with the Securities and Exchange Commission (the <u>SEC</u>) prior to the date of this Agreement (excluding, in each case, any disclosures set forth in any risk factor section or in any other section to the extent they are forward looking statements or cautionary, predictive or forward-looking in nature) or in the corresponding sections or subsections of the disclosure schedule delivered to Parent by the Company prior to entering into this Agreement (the <u>Company Disclosure Schedule</u>) (it being agreed that disclosure of any item in any section or subsection of the Company Disclosure Schedule shall be deemed disclosure with respect to any other section or subsection to which the relevance of such item is reasonably apparent), the Company hereby represents and warrants to Parent and Merger Sub that:
- (a) <u>Organization, Good Standing and Qualification</u>. Each of the Company and its Subsidiaries is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has

all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation or other legal entity in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the

A-8

#### **Table of Contents**

failure to be so organized, qualified or in good standing, or to have such power or authority, is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement. The Company has made available to Parent complete and correct copies of the Company s certificate of incorporation and by-laws, each as amended to the date of this Agreement, and each as so delivered is in full force and effect.

As used in this Agreement, the term (i) <u>Subsidiary</u> means, with respect to any Person, any other Person of which at least a majority of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other persons performing similar functions is directly or indirectly owned or controlled by such Person and/or by one or more of its Subsidiaries; (ii) <u>Significant Subsidiary</u> is as defined in Rule 1.02(w) of Regulation S-X promulgated pursuant to the Securities Exchange Act of 1934, as amended (the <u>Exchange Act</u>); (iii) <u>Joint Venture</u> means Tsingtao Brewery Company, Ltd.; and (iv) <u>Material Adverse Effect</u> means a material adverse effect on the financial condition, business or results of operations of the Company and its Subsidiaries taken as a whole; <u>provided</u>, that none of the following, in and of itself or themselves, shall constitute (or be taken into account in determining the occurrence of) a Material Adverse Effect:

- (A) effects resulting from changes in the economy or financial, credit, banking, currency, commodities or capital markets generally in the United States or other countries in which the Company conducts material operations or any changes in currency exchange rates, interest rates, monetary policy or inflation;
- (B) effects resulting from changes that are the result of factors generally affecting the beer, packaging or theme park industries;
- (C) effects resulting from changes in Law or in United States generally accepted accounting principles (<u>GAAP</u>) or rules and policies of the Public Company Accounting Oversight Board;
- (D) any act of God or other calamity, national or international, political or social conditions (including the engagement by any country in hostilities, whether commenced before or after the date hereof, and whether or not pursuant to the declaration of a national emergency or war), or the occurrence of any military or terrorist attack;
- (E) effects resulting from any failure by the Company to meet any estimates of revenues or earnings on or after the date of this Agreement, <u>provided</u> that the exception in this clause (E) shall not prevent or otherwise affect any change, effect, circumstance or development underlying such failure from being taken into account in determining whether a Material Adverse Effect has occurred:
- (F) (i) the announcement or the existence of this Agreement and the transactions contemplated hereby (including any related or resulting loss of or change in relationship with any customer, supplier, distributor, wholesaler or other business partner, or departure of any employee or officer, or any litigation or other proceeding), including by reason of the identity of Parent or any plans or intentions of Parent with respect to the conduct of the business of any of the Company or its Subsidiaries or (ii) compliance with the terms of, or any actions taken pursuant to, this Agreement, or any failures to take action which is prohibited by this Agreement, or such other changes or events to which Parent has expressly consented in writing; or
- (G) any item or items set forth in Section 5.1(a)(G) of the Company Disclosure Letter;

<u>provided</u>, <u>however</u>, that, with respect to clauses (A), (B), (C) and (D), any effects resulting from any change, event, circumstance or development that disproportionately adversely affects the Company and its Subsidiaries compared to other companies operating in the beer, packaging or theme park industries, as the case may be, shall be considered for purposes of determining whether a Material Adverse Effect has occurred, but only to the extent of such

disproportionate effect.

## (b) Capital Structure.

(i) The authorized capital stock of the Company consists of 1,600,000,000 Shares, of which, as of the close of business on June 30, 2008, 719,042,325 Shares (including outstanding shares of restricted stock) were

A-9

#### **Table of Contents**

outstanding and 774,890,269 Shares were held in treasury and 40,000,000 shares of preferred stock, of which no shares are outstanding. All of the outstanding Shares have been duly authorized and are validly issued, fully paid and nonassessable. Other than (i) 819,677 Shares reserved for issuance under the Company s 1989 Incentive Stock Plan, 1998 Incentive Stock Plan and Global Employee Stock Purchase Plan and (ii) 35,110,917 Shares reserved for delivery out of the Company s treasury pursuant to the 2007 Equity and Incentive Plan, the Stock Plan for Non-Employee Directors, the 2006 Restricted Plan for Non-Employee Directors, the 2008 Long-Term Incentive Plan for Non-Employee Directors, and the Non-Employee Director Elective Stock Purchase Plan (the plans referred to in clauses (i) and (ii) being collectively referred to as the **Stock Plans** ), the Company has no Shares reserved for issuance or delivery. As of the close of business on June 30, 2008, there were 88,718,052 options outstanding and 930,953 shares of restricted stock outstanding under the Stock Plans. Each of the outstanding shares of capital stock or other securities of each of the Company s wholly-owned Subsidiaries, which are set forth in Section 5.1(b)(i)(A) of the Company Disclosure Schedule (Wholly Owned Subsidiaries), are duly authorized, validly issued, fully paid and nonassessable and owned by the Company or by a direct or indirect wholly owned Subsidiary of the Company, free and clear of any lien, charge, pledge, security interest, claim or other encumbrance (each, a Lien ) other than transfer restrictions arising pursuant to applicable securities Laws. Each of the outstanding shares of capital stock or other securities that are directly or indirectly owned by the Company of each of the Company s Subsidiaries that are not Wholly Owned Subsidiaries, which are set forth in Section 5.1(b)(i)(A) of the Company Disclosure Schedule (Non-Wholly Owned Subsidiaries), are duly authorized, validly issued, fully paid and nonassessable and owned by the Company or by a direct or indirect Wholly Owned Subsidiary, free and clear of any Lien other than transfer restrictions arising pursuant to applicable securities Laws. Except as set forth in Sections 5.1(b)(i)(A) or 5.1(b)(i)(B) of the Company Disclosure Schedule, there are no preemptive or other outstanding rights, options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or rights of any kind that obligate the Company or any of its Subsidiaries to issue or sell any shares of capital stock or other securities of the Company or any of its Subsidiaries or any securities or obligations convertible or exchangeable into or exercisable for, or giving any Person a right to subscribe for or acquire, any securities of the Company or any of its Subsidiaries, and no securities or obligations evidencing such rights are authorized, issued or outstanding. The Company does not have outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the stockholders of the Company on any matter.

(ii) In all material respects, each Company Option (A) was granted in compliance with all applicable Laws and all of the terms and conditions of the Stock Plan pursuant to which it was issued, (B) has an exercise price per Share equal to or greater than the fair market value of a Share on the date of such grant, (C) has a grant date identical to the date on which the Company s board of directors or compensation committee actually awarded such Company Option, (D) qualifies for the Tax (as defined in Section 5.1(n)) and accounting treatment afforded to such Company Option in the Company s Tax Returns (as defined in Section 5.1(n)) and the Company Reports (as defined in Section 5.1(e)), respectively, and (E) is not deferred compensation under Section 409A of the Code.

## (c) Corporate Authority; Approval and Fairness.

(i) The Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Merger, subject only to adoption of this Agreement by the holders of a majority of the outstanding Shares entitled to vote on such matter at a stockholders meeting duly called and held for such purpose (the **Requisite Company Vote**). This Agreement has been duly executed and delivered by the Company and constitutes a valid and binding agreement of the Company enforceable against the Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar Laws of general applicability relating to or affecting creditors rights and to general equity principles (the **Bankruptcy and Equity Exception**).

(ii) Subject to Section 6.2, the board of directors of the Company has (A) unanimously determined that the Merger is fair to, and in the best interests of, the Company and its stockholders, approved and declared

A-10

#### **Table of Contents**

advisable this Agreement and the Merger and the other transactions contemplated hereby and resolved to recommend adoption of this Agreement to the holders of Shares (the <u>Company Recommendation</u>), (B) directed that this Agreement be submitted to the holders of Shares for their adoption and (C) received the opinions of its financial advisors, Goldman, Sachs & Co. and Citigroup Global Markets, Inc., to the effect that, as of the date of this Agreement and based upon and subject to the qualifications and assumptions set forth therein, the Per Share Merger Consideration is fair from a financial point of view to such holders (other than Parent and its Subsidiaries) of Shares. True and correct copies of the foregoing opinions have been (or, if not received by the Company s board of directors prior to the execution and delivery of this Agreement, promptly will be) delivered to Parent. It is agreed and understood that such opinion is for the benefit of the Company s board of directors and may not be relied upon by Parent or Merger Sub.

## (d) Governmental Filings; No Violations; Certain Contracts.

- (i) Other than the filings and/or notices pursuant to Section 1.3, under the Exchange Act and the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the **HSR Act** ) and any notices to, approvals of, or consents or clearances by, any other Governmental Antitrust Entity (collectively, the **Company Approvals** ), no notices, reports or other filings are required to be made by the Company with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by the Company from, any domestic, multinational or foreign governmental or regulatory authority, agency, commission, body, court or other legislative, executive or judicial governmental entity (each a **Governmental Entity** ), in connection with the execution, delivery and performance of this Agreement by the Company and the consummation of the Merger and the other transactions contemplated hereby, or in connection with the continuing operation of the business of the Company and its Subsidiaries following the Effective Time, except those that the failure to make or obtain are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement.
- (ii) The execution, delivery and performance of this Agreement by the Company does not, and the consummation of the Merger and the other transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the certificate of incorporation or by-laws of the Company or the comparable governing documents of any of its Significant Subsidiaries, (B) with or without notice, lapse of time or both, a material breach or violation of, assuming (solely with respect to performance of this Agreement and consummation of the Merger and the other transactions contemplated hereby), compliance with the matters referred to in Section 5.1(d)(i), any Law to which the Company or any of its Significant Subsidiaries is subject, (C) with or without notice, lapse of time or both, a breach or violation of, a termination, cancellation or modification (or right of termination, cancellation or modification) or default under, the payment of additional fees, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of the Company or any of its Significant Subsidiaries pursuant to any agreement, lease, license, contract, settlement, consent, note, mortgage or indenture not otherwise terminable by the other party thereto on 90 days or less notice (each, a Contract ) binding upon the Company or any of its Subsidiaries, (D) a breach or violation of or a termination, cancellation or modification (or right of termination, cancellation or modification) or default or right of first refusal or similar right under any investment agreement, shareholders agreement or any other similar agreement with respect to a Joint Venture to which the Company or any of its Subsidiaries is a party or by which any of them are otherwise bound, or (E) any change in the rights or obligations of any party under any Contract binding upon the Company or any of its Subsidiaries or Joint Ventures, except, in the case of clause (C), (D) or (E) above, for any such breach, violation, termination, default, creation, acceleration or change that is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement.

#### (e) Company Reports; Financial Statements.

(i) The Company has filed or furnished, as applicable, on a timely basis, all forms, statements, certifications, reports and documents required to be filed or furnished by it with the SEC pursuant to the Exchange Act or the Securities Act of 1933, as amended (the <u>Securities Act</u>), since December 31, 2006

A-11

#### **Table of Contents**

- (the <u>Applicable Date</u>) (the forms, statements, reports and documents filed or furnished since the Applicable Date and those filed or furnished subsequent to the date of this Agreement, including any amendments thereto, the <u>Company</u> <u>Reports</u>). Each of the Company Reports, at the time of its filing or being furnished complied or, if not yet filed or furnished, will comply in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act of 2002 (the <u>Sarbanes-Oxley Act</u>), and any rules and regulations promulgated thereunder applicable to the Company Reports. As of their respective dates (or, if amended prior to the date of this Agreement, as of the date of such amendment), the Company Reports did not, and any Company Reports filed with or furnished to the SEC subsequent to the date of this Agreement will not, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances in which they were made, not misleading. The Company s derivatives activities and business are substantially as described in the Company s 2007 Annual Report.
- (ii) The Company is in compliance in all material respects with the applicable listing and corporate governance rules and regulations of the New York Stock Exchange (the <u>NYSE</u>). Except as permitted by the Exchange Act, including Section 13(k)(2) and Section 13(k)(3) thereof or the rules of the SEC, since the enactment of the Sarbanes-Oxley Act, neither the Company nor any of its Subsidiaries has made, arranged or modified (in any material way) any extensions of credit in the form of a personal loan to any executive officer or director of the Company or any of their respective Affiliates. For purposes of this Agreement, the term <u>Affiliate</u> when used with respect to any party shall mean any Person who is an affiliate of that party within the meaning of Rule 405 promulgated under the Securities Act.
- (iii) The Company maintains disclosure controls and procedures required by Rule 13a-15 or 15d-15 under the Exchange Act. Such disclosure controls and procedures are designed to provide reasonable assurance that information required to be disclosed by the Company is recorded and reported on a timely basis to the individuals responsible for the preparation of the Company s filings with the SEC and other public disclosure documents. The Company maintains internal control over financial reporting (as defined in Rule 13a-15 or 15d-15, as applicable, under the Exchange Act). Such internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP and includes policies and procedures that (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the asset of the Company, (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that receipts and expenditures of the Company are being made only in accordance with authorizations of management and directors of the Company, and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company s assets that could have a material effect on its financial statements. The Company has disclosed, based on the most recent evaluation of its chief executive officer and its chief financial officer prior to the date of this Agreement, to the Company s auditors and the audit committee of the Company s board of directors (A) any significant deficiencies in the design or operation of its internal controls over financial reporting that are reasonably likely to adversely affect the Company s ability to record, process, summarize and report financial information and has identified for the Company s auditors and audit committee of the Company s board of directors any material weaknesses in internal control over financial reporting and (B) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company s internal control over financial reporting.
- (iv) Each of the consolidated balance sheets included in or incorporated by reference into the Company Reports (including the related notes and schedules) fairly presents, or, in the case of Company Reports filed after the date of this Agreement, will fairly present, in all material respects, the consolidated financial position of the Company and its consolidated Subsidiaries as of its date, and each of the consolidated statements of income, changes in shareholders equity and cash flows included in or incorporated by reference into the Company Reports (including any related notes and schedules) fairly presents, or in the case of Company Reports filed after the date of this Agreement, will fairly present, in all material respects, the results of operations, retained earnings and changes in financial position, as the

case may be, of such companies for the periods set forth therein (subject, in the case of unaudited statements, to notes and normal year-end audit

A-12

#### **Table of Contents**

adjustments), in each case in accordance with GAAP consistently applied during the periods involved, except as may be noted therein.

- (f) <u>Absence of Certain Changes</u>. Since December 31, 2007, (i) the Company and its Subsidiaries have conducted their respective businesses in all material respects in the ordinary course of business consistent with past practice, and (ii) there has not been:
- (A) any change in the financial condition, business or results of their operations or any circumstance, occurrence or development of which the Company has knowledge which, individually or in the aggregate, has had or is reasonably likely to have a Material Adverse Effect;
- (B) other than regular quarterly dividends on Shares of \$0.33 per Share (or with respect to periods after the date hereof, \$0.37 per Share), any declaration, setting aside or payment of any dividend or other distribution with respect to any shares of capital stock of the Company or any of its Subsidiaries (except for dividends or other distributions by any Subsidiary to the Company or to any Subsidiary of the Company);
- (C) any material change in any method of accounting or accounting practice by the Company or any of its Subsidiaries:
- (D) except as expressly permitted by this Agreement, (1) any increase in the compensation or benefits payable or to become payable to its officers or employees (except for increases in the ordinary course of business and consistent with past practice) or (2) any establishment, adoption, entry into or amendment of any collective bargaining, bonus, profit sharing, equity, thrift, compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee, except to the extent required by applicable Law; or
- (E) any agreement to do any of the foregoing.

As used in this Agreement, the term **knowledge** when used in the phrases to the knowledge of the Company, the Company s knowledge, of which the Company has knowledge, or the Company has no knowledge or words of similar import shall mean the knowledge of the Persons listed in Section 5.1(f) of the Company Disclosure Schedule.

(g) Litigation and Liabilities. There are no civil, criminal, administrative or other actions, suits, claims, oppositions, litigations, hearings, arbitrations, investigations or other proceedings (\_Actions ) pending or, to the knowledge of the Company, threatened against the Company or any of its Subsidiaries or obligations or liabilities of the Company or any of its Subsidiaries, whether or not accrued, contingent or otherwise, that would be required to be reflected or reserved against in a consolidated balance sheet of the Company and its Subsidiaries (\_Liabilities ) or the notes thereto if such balance sheet was prepared as of the date hereof, except (i) as reflected or reserved against in the Company s consolidated balance sheets (and the notes thereto) included in the Company Reports filed after the Applicable Date but prior to the date of this Agreement, (ii) for obligations or Liabilities incurred in the ordinary course of business since December 31, 2007, or (iii) for obligations or Liabilities incurred pursuant to the transactions contemplated by this Agreement, or (iv) for those that are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions of any judgment, order, writ, injunction, decree, award, stipulation or settlement (\_Iudgment ) of any Governmental Entity which is, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement.

#### (h) Employee Benefits.

(i) All material benefit and compensation plans, employment agreements, contracts, policies or arrangements covering current or former employees of the Company and its Subsidiaries (the <u>Employees</u>) and current or former directors of the Company, including, but not limited to, employee benefit plans within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>), and deferred compensation, severance, stock option, stock purchase, stock appreciation rights, Company stock-

A-13

#### **Table of Contents**

based, incentive and bonus plans (the <u>Company Benefit Plans</u>), other than Company Benefit Plans maintained outside of the United States primarily for the benefit of Employees working outside of the United States (such plans hereinafter being referred to as <u>Company Non-U.S. Benefit Plans</u>), are listed in Section 5.1(h)(i) of the Company Disclosure Schedule. True and complete copies of all Company Benefit Plans listed in Section 5.1(h)(i) of the Company Disclosure Schedule, including any trust instruments, insurance contracts, actuarial reports and, with respect to any employee stock ownership plan, loan agreements forming a part of any Company Benefit Plans, and all amendments thereto have been provided or made available to Parent or will be made available to Parent within ten (10) days following the date hereof.

- (ii) All Company Benefit Plans, other than multiemployer plans within the meaning of Section 3(37) of ERISA (each, a Multiemployer Plan ) and Company Non-U.S. Benefit Plans (collectively, Company U.S. Benefit Plans ) are in material compliance with ERISA, the Code and other applicable Laws. Each Company U.S. Benefit Plan which is subject to ERISA (a Company ERISA Plan ) that is an employee pension benefit plan within the meaning of Section 3(2) of ERISA (a Company Pension Plan ) intended to be qualified under Section 401(a) of the Code, has received a favorable determination letter from the Internal Revenue Service (the RS ) covering all tax Law changes prior to the Economic Growth and Tax Relief Reconciliation Act of 2001 or has applied to the IRS for such favorable determination letter within the applicable remedial amendment period under Section 401(b) of the Code, and the Company is not aware of any circumstances likely to result in the loss of the qualification of such Plan under Section 401(a) of the Code except for any such loss that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any of its Subsidiaries has engaged in a transaction with respect to any Company ERISA Plan that, assuming the taxable period of such transaction expired as of the date of this Agreement, could subject the Company or any Subsidiary to a tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA except that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- (iii) Except for those that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect, (A) no liability under Subtitle C or D of Title IV of ERISA has been or is expected to be incurred by the Company or any of its Subsidiaries with respect to any ongoing, frozen or terminated single-employer plan , within the meaning of Section 4001(a)(15) of ERISA, currently or formerly maintained by any of them, or the single-employer plan of any entity which is considered one employer with the Company under Section 4001 of ERISA or Section 414 of the Code (an **\_ERISA Affiliate** ); (B) the Company and its Subsidiaries have not incurred and do not expect to incur any material withdrawal liability with respect to a Multiemployer Plan under Subtitle E of Title IV of ERISA (regardless of whether based on contributions of an ERISA Affiliate); (C) no notice of a reportable event , within the meaning of Section 4043 of ERISA for which the reporting requirement has not been waived or extended, other than pursuant to Pension Benefit Guaranty Corporation (**\_PBGC** ) Reg. Section 4043.33 or 4043.66, has been required to be filed for any Company Pension Plan or by any ERISA Affiliate within the 12-month period ending on the date hereof or will be required to be filed in connection with the transaction contemplated by this Agreement.
- (iv) As of the date of this Agreement, there is no pending or, to the knowledge of the Company threatened, claims or litigation relating to the Company Benefit Plans, except that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any of its Subsidiaries has any material obligations for retiree health and life benefits under any Company ERISA Plan or collective bargaining agreement.
- (v) Neither the execution of this Agreement, shareholder approval of this Agreement nor the consummation of the transactions contemplated hereby will (x) entitle any employees of the Company or any of its Subsidiaries to severance pay or any increase in severance pay upon any termination of employment after the date of this Agreement, or (y) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Company Benefit Plans. Section 5.1(h)(v) of the Company Disclosure Schedule separately lists

the potential cash severance amounts (including each component of each such cash severance amount) payable to each executive officer upon a qualifying termination after the Effective Time.

A-14

#### **Table of Contents**

- (vi) Neither the execution of this Agreement, shareholder approval of this Agreement nor the consummation of the transactions contemplated hereby will result in payments under any of the Company Benefit Plans which would not be deductible under Section 280G of the Code. Section 5.1(h)(v) of the Company Disclosure Schedule separately lists any Company Benefit Plans that provide for a Section 4999 gross-up or cut back.
- (vii) All Company Benefit Plans that are nonqualified deferred compensation plans (within the meaning of Section 409A of the Code) have been maintained and administered in material good faith operational compliance with the requirements of Section 409A of the Code and any regulations or other guidance issued thereunder. No Company Benefit Plan provides for a Section 409A gross-up or indemnity.
- (viii) All Company Non-U.S. Benefit Plans comply in all material respects with applicable Law, except as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. As of the date of this Agreement, there is no pending or, to the knowledge of the Company, threatened material claims or litigation relating to Company Non-U.S. Benefit Plans.
- (i) Compliance with Laws; Licenses. The businesses of each of the Company and its Subsidiaries (including the ownership and maintenance of all its assets, including the Owned Real Property (as defined in Section 5.1(k)(i))) have not, since December 31, 2006, been, and are not being, conducted in violation of any applicable federal, state, local, multinational or foreign law, statute or ordinance, common law, or any rule, regulation, directive, treaty, policy, standard, Judgment, agency requirement, license or permit of any Governmental Entity (collectively, <u>Laws</u>), except for violations that, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement. The Company has not taken any action which would cause it to be in material violation of the Foreign Corrupt Practices Act of 1977, as amended, or any rules and regulations thereunder. No investigation, audit or review by any Governmental Entity with respect to the Company or any of its Subsidiaries or any of their assets is pending or, to the knowledge of the Company, threatened, nor has any Governmental Entity notified the Company of its intention to conduct the same, except for such investigations or reviews the outcome of which are not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement. As of the date hereof, to the knowledge of the Company, the Company has not received any notice or communication of any material noncompliance with any applicable Laws that has not been cured as of the date of this Agreement. The Company and its Subsidiaries each has obtained and is in compliance with all permits, licenses, certifications, approvals, registrations, consents, authorizations, franchises, variances, exemptions and orders issued or granted by a Governmental Entity necessary to conduct its business as presently conducted and operate the Owned Real Properties (as defined in Section 5.1(k)(i)), except those the absence of which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect or prevent or materially impair the consummation of the transactions contemplated by this Agreement.

## (j) Material Contracts.

- (i) Except for this Agreement and except for Contracts filed as exhibits to the Company Reports, as of the date of this Agreement, none of the Company or its Subsidiaries is a party to or bound by:
- (A) any Contract that would be required to be filed by the Company as a material contract pursuant to Item 601(b)(10) of Regulation S-K under the Securities Act;
- (B) any Contract involving the payment or receipt of amounts by the Company or any of its Subsidiaries of more than \$50 million in any calendar year;

(C) any Contract that contains a put, call or similar right pursuant to which the Company or any of its Subsidiaries could be required to purchase or sell, as applicable, any equity interests of any Person or assets that have a fair market value or purchase price of more than \$50 million;

(D) any material agreement relating to the formation, creation, operation, management or control of any Joint Venture;

A-15

#### **Table of Contents**

- (E) any Contract between the Company or any of its Subsidiaries and any director or executive officer of the Company or any Person beneficially owning five percent or more of the outstanding Shares required to be disclosed pursuant to Rule 404 of Regulation S-K under the Exchange Act;
- (F) any Contract with any party with annual revenues of \$300 million or more pursuant to which the Company or any of its Subsidiaries is a party containing a standstill or similar agreement pursuant to which one party has agreed not to acquire assets or securities of the other party or any of its Affiliates; and
- (G) any non-competition Contract or other Contract that (1) limits or purports to limit in any material respect the type of business in which the Company or its Subsidiaries may engage, the type of goods or services which the Company or its Subsidiaries may manufacture, produce, import, export, offer for sale, sell or distribute or the manner or locations in which any of them may so engage in any business or use their assets, or (2) grants most favored nation status that apply to the Company and its Subsidiaries, except in each of clauses (1) and (2) as would not reasonably be expected to materially impair the manner in which the Company and its Subsidiaries operate the beer business as of the date hereof;

Each such Contract described in clauses (A) through (G) above and each such Contract that would be a Material Contract but for the exception of being filed as an exhibit to the Company Reports is referred to herein as a <u>Material</u> Contract.

(ii) Each of the Material Contracts is valid and binding on the Company or its Subsidiaries, as the case may be, and, to the knowledge of the Company, each other party thereto, and is in full force and effect, except for such failures to be valid and binding or to be in full force and effect as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect. There is no breach or default under any Material Contracts by the Company or its Subsidiaries and no event has occurred that with the lapse of time or the giving of notice or both would constitute a breach or default thereunder by the Company or its Subsidiaries, in each case except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect.

## (k) Properties.

- (i) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, with respect to the parcels of real property for the Company's principal domestic facilities owned in fee by the Company or any of its Subsidiaries, and all buildings, structures, improvements, and fixtures thereon, together with all rights of way, easements, privileges and appurtenances pertaining or belonging thereto, including any right, title and interest of the Company or any of its Subsidiaries in and to any street or other property adjoining any portion of such property (the **Owned Real Property**), (A) the Company or one of its Subsidiaries, as applicable, has good and marketable title to the Owned Real Property, free and clear of any Encumbrance, (B) there are no outstanding options or rights of first refusal to purchase the Owned Real Property, or any portion of the Owned Real Property or interest therein, and (C) the major structural elements of the improvements comprising the Owned Real Property, including mechanical, electrical, heating, ventilation, air conditioning or plumbing systems, elevators or parking elements, are in sufficiently good condition (except for ordinary wear and tear) to allow the business of the Company and its Subsidiaries to be operated in the ordinary course of business and consistent with past practices of the Company as currently operated.
- (ii) With respect to the real property with respect to which the Company s principal domestic facilities are leased, subleased or licensed to the Company or its Subsidiaries (the <u>Leased Real Property</u>), the lease, sublease or license for such property is valid, legally binding, enforceable and in full force and effect, except in each case, for such invalidity, failure to be binding or unenforceability that is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect.

(iii) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company and its Subsidiaries have good and marketable title to, or a valid and binding leasehold interest in, all their respective domestic properties and assets (excluding Owned Real

A-16

#### **Table of Contents**

Property, Leased Real Property and Intellectual Property (as defined in Section 5.1(p))), in each case free and clear of all Encumbrances.

- (iv) For purposes of this Section 5.1(k) only, **Encumbrance** means any Lien, mortgage, easement, covenant, or other restriction or title matter or encumbrance of any kind in respect of such asset but specifically excludes (a) specified encumbrances described in Section 5.1(k)(iv) of the Company Disclosure Schedule; (b) encumbrances for current Taxes or other governmental charges not yet due and payable; (c) mechanics, carriers, workmen s, repairmen s or other like encumbrances arising or incurred in the ordinary course of business as to which there is no default on the part of the Company or any of its Subsidiaries and reflected on or specifically reserved against or otherwise disclosed in the Company s consolidated balance sheets (and the notes thereto) included in the Company Reports filed prior to the date of this Agreement; and (d) other encumbrances that do not, individually or in the aggregate, materially impair the continued use, operation, value or marketability of the specific parcel of Owned Real Property to which they relate or the conduct of the business of the Company and its Subsidiaries as presently conducted.
- (1) Takeover Statutes. No business combination, fair price, moratorium, control share acquisition or other similar anti-takeover statute or regulation (including Section 203 of the DGCL) or any similar anti-takeover provision in the Company s certificate of incorporation (including Article EIGHTH) or by-laws (each, a Takeover Statute ) is applicable to the Company, the Shares, the Merger or the other transactions contemplated by this Agreement. The Company and the board of directors of the Company has taken all action so that (i) Parent will not be an interested stockholder or subject to restrictions or delays in entering into or consummating a business combination with the Company (in each case as such term is used in Section 203 of the DGCL) as a result of the execution of this Agreement or the consummation of the transactions in the manner contemplated hereby, (ii) Parent will not be an Interested Shareholder or subject to restrictions or delays in entering into or consummating a Business Transaction with the Company (in each case as such term is used in the Company s certificate of incorporation) as a result of the execution of this Agreement or the consummation of the transactions in the manner contemplated hereby, and (iii) Section 203 of the DGCL and Article EIGHTH of the Company s certificate of incorporation is not applicable to the Company, the Shares, the Merger or the other transactions contemplated by this Agreement.
- (m) Environmental Matters. Except for such matters that, alone or in the aggregate, are not reasonably likely to have a Material Adverse Effect: (i) the Company and its Subsidiaries are in compliance, and have complied at all times with, all applicable Environmental Laws; (ii) no property currently or, to the knowledge of the Company, formerly owned or operated by the Company or any of its Subsidiaries has been contaminated with any Hazardous Substance in a manner that could reasonably be expected to require remediation or other action pursuant to any Environmental Law; (iii) neither the Company nor any of its Subsidiaries has received any written notice, demand, letter, claim or request for information alleging that the Company or any of its Subsidiaries is in violation of or liable under any Environmental Law; and (iv) neither the Company nor any of its Subsidiaries is subject to any order, decree or injunction with any Governmental Entity or agreement with any third party concerning liability under any Environmental Law or relating to Hazardous Substances. As used herein, the term **Environmental Law** means any federal, state or local statute, Law, regulation, order, decree, permit or authorization relating to: (A) the protection of health, safety or the environment or (B) the handling, use, transportation, disposal, release or threatened release of any Hazardous Substance. This Section 5.1(m) constitutes the only representations and warranties of the Company with respect to any Environmental Law. As used herein, the term Hazardous Substance means any substance that is: (A) listed, classified, regulated under any Environmental Law as hazardous substance, toxic substance, pollutant, contaminant or oil or (B) any petroleum product or by-product, asbestos-containing material, polychlorinated biphenyls or radioactive material.
- (n) <u>Taxes</u>. The Company and each of its Subsidiaries (i) have prepared in good faith and duly and, in cases where the statute of limitations would still be open, timely filed (taking into account any extension of time within which to file) all income, franchise, and similar Tax Returns (as defined below) and all other material Tax Returns required to be

filed by any of them and all such filed Tax Returns are complete and accurate in all material respects; (ii) have paid all Taxes (as defined below) that are shown as due on such filed Tax Returns and any Material Taxes that the Company or any of its Subsidiaries are obligated to withhold

A-17

#### **Table of Contents**

from amounts owing to any employee, creditor or third party, except with respect to matters contested in good faith; and (iii) have not waived any statute of limitations with respect to Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency which waiver or extension are still in effect. As of the date of this Agreement, there are not pending or, to the knowledge of the Company, threatened, any audits, examinations, investigations or other proceedings in respect of Taxes or Tax matters. There are not, to the knowledge of the Company, any unresolved questions or claims concerning the Company s or any of its Subsidiaries Tax liability that are, individually or in the aggregate, reasonably likely to have a Material Adverse Effect and are not disclosed or provided for in the Company Reports. The Company has made available to Parent true and correct copies of the United States federal income Tax Returns filed by the Company and its Subsidiaries for each of the fiscal years ended December 31, 2006, 2005 and 2004. Neither the Company nor any of its Subsidiaries has engaged in any transaction that is the same as, or substantially similar to, a transaction which is a reportable transaction for purposes of Treasury Regulations Section 1.6011-4(b) (including any transaction which the IRS has determined to be a listed transaction for purposes of 1.6011-4(b)(2)). Neither the Company nor any of Subsidiaries has engaged in a transaction of which it made disclosure to any taxing authority to avoid penalties under Section 6662(d) of the Code or any comparable provision of state, foreign or local law. Neither the Company nor any of its Subsidiaries has participated in any tax amnesty or similar program offered by any taxing authority to avoid the assessment of penalties or other additions to Tax.

As used in this Agreement, (i) the term <u>Tax</u> (including, with correlative meaning, the term <u>Taxs</u>) includes all federal, state, local and foreign income, profits, franchise, gross receipts, environmental, customs duty, capital stock, net worth, severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value added, goods and services, occupancy, transfer and other taxes, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of such penalties and additions, and (ii) the term <u>Tax Return</u> includes all returns and reports (including elections, declarations, disclosures, schedules, estimates and information returns) required to be supplied to a Tax authority relating to Taxes.

(o) <u>Labor Matters</u>. Section 5.1(o) of the Company Disclosure Schedules sets forth a list of all labor and collective bargaining agreements, Contracts or other agreements with a labor union, works council or labor organization to which the Company or any of its Subsidiaries is party or by which any of them are otherwise bound pertaining to certain Employees in the United States (collectively, the **Company Labor Agreements** ). As of the date of this Agreement, except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect, neither the Company nor any of its Significant Subsidiaries is the subject of any proceeding that asserts that the Company or any of its Subsidiaries has committed an unfair labor practice (nor, to the knowledge of the Company, is there a basis for any such claim), nor is there pending or, to the knowledge of the Company, threatened, nor has there been, since January 1, 2007, any labor strike, dispute, walk-out, work stoppage, labor picketing, lockout or material slowdown involving the Company or any of its Subsidiaries. Except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect, the consummation of the Merger and the other transactions contemplated by this Agreement will not entitle any third party (including any labor union or labor organization) to any payments under any Company Labor Agreements. Except as would not, or would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect, the Company and its Subsidiaries have complied in all respects with the reporting requirements of the Labor Management Reporting and Disclosure Act.

## (p) Intellectual Property.

(i) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, (A) the Company and its Subsidiaries own or have sufficient rights to use all Intellectual Property that is used in their respective businesses as currently conducted (the **Company IP**) free and clear of all liens and security

interests, and will have the same rights after the Closing Date; and (B) all of the registrations and applications included in the Company IP owned by, and to the knowledge of the Company, the Company IP exclusively licensed by the Company and its Subsidiaries, are subsisting.

A-18

#### **Table of Contents**

- (ii) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, (A) neither the conduct of the business of the Company and/or the conduct of the business of each of its Subsidiaries nor the Company IP infringes, dilutes, misappropriates or otherwise violates any Intellectual Property rights of any third party; and (B) to the knowledge of the Company, no third party is infringing, diluting, misappropriating or otherwise violating any Company IP owned or exclusively licensed by the Company or its Subsidiaries.
- (iii) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company and its Subsidiaries take and have taken commercially reasonable measures to maintain, preserve and protect the confidentiality of all Trade Secrets, and to the Company s knowledge, such Trade Secrets have not been used, disclosed or discovered by any Person except pursuant to valid and appropriate non-disclosure and/or license agreements or pursuant to obligations to maintain confidentiality arising by operation of law.
- (iv) Except in any such case as is not, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company s and its Subsidiaries computers, Software, firmware, middleware, servers, systems, networks and all other information technology equipment operate and perform as required by the Company and each of its Subsidiaries in connection with their respective businesses as currently conducted and have not malfunctioned or failed within the past three years, without such malfunction or failure having been resolved.

For purposes of this Agreement: <a href="Intellectual Property">Intellectual Property</a> means all foreign, multinational and domestic (A) trademarks, service marks, brand names, corporate names, Internet domain names, logos, symbols, trade dress, trade names, and all other source indicators and all goodwill associated therewith and symbolized thereby; (B) patents and proprietary inventions and discoveries; (C) confidential and proprietary information, trade secrets and know-how, (including confidential and proprietary processes, technology, research, recipes, schematics, business methods, formulae, drawings, prototypes, models, designs, customer lists and supplier lists) (collectively, <a href="Irrade Secrets">Irrade Secrets</a> ); (D) copyrights and works of authorship in any media (including copyrights in software, databases, source code, object code and all documentation related thereto, Internet site content, advertising and marketing materials and art work); (E) all other intellectual property rights; and (F) all applications and registrations, invention disclosures, and extensions, revisions, restorations, substitutions, modifications, renewals, divisions, continuations, continuations-in-part, reissues and re-examinations related to any of the foregoing; <a href="Registered">Registered</a> means issued by, filed with, registered with, renewed by or the subject of a pending application before any Governmental Entity or Internet domain name registrar; and <a href="Software">Software</a> means any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code or object code, and (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise.

- (q) <u>Brokers and Finders</u>. Neither the Company nor any of its officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finders—fees in connection with the Merger or the other transactions contemplated in this Agreement, except that the Company has employed Goldman, Sachs & Co., Citigroup Global Capital Markets Inc. and Moelis & Company as its financial advisors. The Company has made available to Parent a complete and accurate copy of all agreements pursuant to which any advisor to the Company is entitled to any fees and expenses in connection with any of the transactions contemplated by this Agreement.
- (r) <u>Solvency</u>. Immediately prior to the Effective Time (and before giving effect to any financing that is to be put in place in connection with the transactions contemplated hereby), the Company is Solvent. <u>Solvent</u> means, with respect to any Person and a specified date of determination, that at such date: (i) the present fair saleable value of such Person s assets is in excess of the total amount of such Person s probable liabilities on its existing debts and obligations (including contingent liabilities) as they become absolute and matured; (ii) such Person is able to pay its debts as they become due; and (iii) such Person does not have unreasonably small capital to carry on such Person s business as

theretofore operated and all businesses in which such Person is then about to engage.

A-19

#### **Table of Contents**

- (s) *No Additional Representations*. Except for those representations and warranties expressly set forth in this Section 5.1 and except as otherwise expressly set forth in this Agreement, neither the Company nor any other person acting on behalf of the Company makes any representation or warranty of any kind or nature, express or implied, in connection with the transactions contemplated by this Agreement. Neither the Company nor any of its Subsidiaries has made or makes any representation or warranty with respect to any projections, estimates or budgets made available to Merger Sub or its affiliates of future revenues, future results of operations (or any component thereof), future cash flows or future financial condition (or any component thereof) of the Company and its Subsidiaries or the future business and operations of the Company and its Subsidiaries.
- 5.2. <u>Representations and Warranties of Parent and Merger Sub</u>. Except as set forth in the corresponding sections or subsections of the disclosure schedule delivered to the Company by Parent prior to entering into this Agreement (the <u>Parent Disclosure Schedule</u>) (it being agreed that disclosure of any item in any section or subsection of the Parent Disclosure Schedule shall be deemed disclosure with respect to any other section or subsection to which the relevance of such item is reasonably apparent), Parent hereby represents and warrants to the Company that:
- (a) <u>Organization, Good Standing and Qualification</u>. Each of Parent and Merger Sub is a legal entity duly organized, validly existing and in good standing under the Laws of its respective jurisdiction of organization and has all requisite corporate or similar power and authority to own, lease and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation in each jurisdiction where the ownership, leasing or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, qualified or in such good standing, or to have such power or authority, would not, individually or in the aggregate, reasonably be expected to prevent or materially impair the ability of Parent and Merger Sub to consummate the Merger and the other transactions contemplated by this Agreement. Parent has made available to the Company complete and correct copies of Parent s articles of incorporation and Merger Sub s certificate of incorporation and by-laws, each as amended to the date of this Agreement, and each as so delivered is in full force and effect. All of the capital stock of Merger Sub is owned by InBev Worldwide S.à.r.l., a limited liability company organized under the laws of Luxembourg and a wholly owned subsidiary of Parent.
- (b) <u>Corporate Authority</u>. No vote of holders of capital stock of Merger Sub is necessary to approve this Agreement and the Merger and the other transactions contemplated hereby. Each of Parent and Merger Sub has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Merger, subject only to approval of the Merger (in accordance with Section 23 of Parent s by-laws) and certain other transactions contemplated by this Agreement, by 75% (or 50% with respect to certain of the other transactions) of the outstanding ordinary shares of Parent (<u>Parent Common Stock</u>) attending in person or being represented by a proxyholder at a shareholders meeting of Parent duly called and held for such purpose (the <u>Requisite Parent Vote</u>) at which a quorum is present. This Agreement has been duly executed and delivered by each of Parent and Merger Sub and is a valid and binding agreement of Parent and Merger Sub, enforceable against each of Parent and Merger Sub in accordance with its terms, subject to the Bankruptcy and Equity Exception. At or prior to the execution and delivery of this Agreement, Parent has delivered to the Company a true and correct copy of the Parent Shareholder Commitment.

## (c) Governmental Filings; No Violations; Etc.

(i) Other than any filing with and approval of the Belgian Banking, Finance and Insurance Commission and/or Euronext Brussels (or similar regulators or (market) authorities in other jurisdictions) which may be required in conjunction with the Parent Prospectus or the Equity Financing (collectively, the <u>Equity Financing Filings</u>), the filings and/or notices pursuant to Section 1.3, under the Exchange Act and the HSR Act and any notices to, approvals of, or consents or clearances by, any other Governmental Antitrust Entity (the <u>Parent Approvals</u>), no notices, reports

A-20

#### **Table of Contents**

filings are required to be made by Parent or Merger Sub with, nor are any consents, registrations, approvals, permits or authorizations required to be obtained by Parent or Merger Sub from, any Governmental Entity in connection with the execution, delivery and performance of this Agreement by Parent and Merger Sub and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby, except those that the failure to make or obtain are not, individually or in the aggregate, reasonably likely to prevent or materially impair the ability of Parent or Merger Sub to consummate the Merger and the other transactions contemplated by this Agreement.

(ii) The execution, delivery and performance of this Agreement by Parent and Merger Sub do not, and the consummation by Parent and Merger Sub of the Merger and the other transactions contemplated hereby will not, constitute or result in (A) a breach or violation of, or a default under, the articles of incorporation or by-laws of Parent or Merger Sub or the comparable governing instruments of any of their respective Subsidiaries, (B) with or without notice, lapse of time or both, a material breach or violation of, assuming (solely with respect to performance of this Agreement and consummation of the Merger and the other transactions contemplated hereby) compliance with the matters referred to in Section 5.2(c)(i), any Law to which Parent or any of its Subsidiaries is subject or (C) with or without notice, lapse of time or both, a breach or violation of, a termination, cancellation or modification (or right of termination, cancellation or modification) or default under, the payment of additional fees, the creation or acceleration of any obligations under or the creation of a Lien on any of the assets of Parent or any of its Subsidiaries pursuant to any Contract binding upon Parent or any of its Subsidiaries, except in the case of clause (C) for any such breach, violation, termination, default, creation or acceleration that would not, individually or in the aggregate, reasonably be likely to prevent or materially impair the ability of Parent or Merger Sub to consummate the Merger and the other transactions contemplated hereby.

### (d) Available Funds.

- (i) Parent and Merger Sub have available or will have available to them, as of the date all of the conditions to Parent s obligation to close pursuant to Section 7.1 and 7.2 have been satisfied (the **Financing Date**), all funds necessary for the payment to the Paying Agent of the aggregate amount of the Exchange Fund and any other amounts required to be paid in connection with the consummation of the Merger and the other transactions contemplated by this Agreement and to pay all related fees and expenses.
- (ii) Parent has made available to the Company true and complete copies of the following financing documents for the Merger: (A) a true and correct copy of the \$45,000,000,000 Senior Facilities Agreement dated 12th July 2008 (the Senior Facilities Agreement ) among Parent and InBev Worldwide S.à.r.l., as Borrowers, and Banco Santander, S.A., Barclays Capital, BNP Paribas, Deutsche Bank AG, London Branch, Fortis Bank SA/NV, ING Bank N.V., J.P. Morgan Bank plc, Mizuho Corporate Bank, Ltd., The Bank of Tokyo-Mitsubishi UFJ, Ltd., and The Royal Bank of Scotland plc, as Mandated Lead Arrangers and Bookrunners, Fortis Bank SA/NV, as Agent, and Fortis Bank SA/NV, as Issuing Bank, and the other lenders from time to time party thereto (the **Senior Lenders** ), and (B) a true and correct copy of the \$5,600,000,000 Bridge Facility Agreement dated 12th July 2008 (the **Bridge Facility** and, together with the Senior Facilities Agreement, the **Facilities** ) among Parent and InBev Worldwide S.à.r.l., as Borrowers, and Banco Santander, S.A., BNP Paribas, Deutsche Bank AG, London Branch, Fortis Bank SA/NV, ING Bank N.V., J.P. Morgan plc, and The Royal Bank of Scotland plc, as Mandated Lead Arrangers and Bookrunners and Fortis Bank SA/NV as Agent (the **Bridge Lenders** and, together with the Senior Lenders, the **Lenders**), and all annexes, schedules and appendixes containing any funding conditions to the funding under the Senior Facilities Agreement and the Bridge Facility, pursuant to which Facilities the Lenders have agreed to lend the amounts contemplated by the Facilities for the purpose of, among other things, consummating the Merger and the other transactions contemplated by this Agreement (\_Financing\_). Each of the Facilities is in full force and effect, and has not been withdrawn or terminated and to Parent s knowledge, is enforceable against the Lenders in accordance

#### **Table of Contents**

with its terms, subject to the Bankruptcy and Equity Exception. Parent will not release or consent to the termination of the obligations of the Lenders under the Senior Facilities Agreement or the Bridge Facility Agreement, except for assignments and replacements of an individual Lender under the terms of or in connection with the syndication of the Facilities. There are no conditions precedent related to the funding under the Facilities, except those expressly set forth in the Senior Facilities Agreement or the Bridge Facilities Agreement. As of the date of this Agreement, Parent has no reason to believe that any of the conditions to the Facilities or the funding thereunder will not be satisfied or that the Facilities and Financing will not be made available to Parent as of the Financing Date.

- (iv) The net proceeds contemplated by the Facilities, together with other existing financial commitments and resources of Parent and Merger Sub and cash on hand of Parent and Merger Sub at the Closing, will in the aggregate be sufficient for Parent and Merger Sub to make all payments under (i) Section 4.1(a) in respect of Shares, (ii) Section 4.3(a) in respect of Company Options and (iii) Section 4.3(b) in respect of Company Awards, and any other amounts required to be paid in connection with the consummation of the Merger and the other transactions contemplated by this Agreement and to pay all related fees and expenses.
- (e) <u>Capitalization of Merger Sub</u>. The authorized capital stock of Merger Sub consists solely of 100 shares of common stock, par value \$0.01 per share, all of which are validly issued and outstanding. All of the issued and outstanding capital stock of Merger Sub is, and at the Effective Time will be, owned by Parent or a direct or indirect wholly owned Subsidiary of Parent. Merger Sub has not conducted any business prior to the date of this Agreement and has no, and prior to the Effective Time will have no, assets, liabilities or obligations of any nature other than those incident to its formation and pursuant to this Agreement and the Merger and the other transactions contemplated by this Agreement.
- (f) <u>Litigation and Judgments</u>. There are no Actions pending or, to the knowledge of Parent, threatened against Parent or any of its Subsidiaries or Liabilities of Parent or its Subsidiaries, except for those that are not, individually or in the aggregate, reasonably likely to prevent or materially impair the consummation of the transactions contemplated by this Agreement. Neither Parent nor any of its Subsidiaries is a party to or subject to the provisions of any Judgment of any Governmental Entity which is, individually or in the aggregate, reasonably likely to prevent or materially impair the consummation of the transactions contemplated by this Agreement.
- (g) <u>Solvency</u>. As of the Effective Time, assuming (A) satisfaction of the conditions to Parent s and Merger Sub s obligation to consummate the Merger, or waiver of such conditions, (B) the accuracy in all respects of the representations and warranties set forth in Section 5.1 hereof without regard to any materiality qualifiers (including Section 5.1(r)), (C) the accuracy of any estimates, projections or forecasts provided by the Company to Parent prior to the Effective Time and (D) assuming that Parent was provided prior to the date hereof all material information relating to any acceleration of indebtedness of the Company and its Subsidiaries or other costs associated with indebtedness of the Company and its Subsidiaries, immediately after giving effect to the transactions contemplated by this Agreement (including the financing under the Facilities, the payment of the aggregate Per Share Merger Consideration and the funding of any obligations of the Surviving Corporation or its subsidiaries which become due or payable by the Surviving Corporation or its subsidiaries in connection with, or as a result of, the Merger, and the payment of all related fees and expenses), the Surviving Corporation will be Solvent.
- (h) <u>Share Ownership</u>. As of the date of this Agreement, Parent and Merger Sub do not beneficially own, within the meaning of Rule 13d-3 of the Exchange Act, more than 250 Shares.
- (i) <u>Brokers and Finders</u>. Neither Parent nor Merger Sub nor any of their respective officers, directors or employees has employed any broker or finder or incurred any liability for any brokerage fees, commissions or finders fees in connection with the Merger or the other transactions contemplated in this Agreement, except that Parent has employed Lazard Frères SAS, J.P. Morgan plc, Deutsche Bank AG and BNP PARIBAS as its financial advisors.

# A-22

#### **Table of Contents**

(j) <u>No Additional Representations</u>. Except for those representations and warranties expressly set forth in this Section 5.2 and except as otherwise expressly set forth in this Agreement, none of Parent, Merger Sub or any other person acting on behalf of Parent or Merger Sub makes any representation or warranty of any kind or nature, express or implied, in connection with the transactions contemplated by this Agreement.

#### ARTICLE VI

### Covenants

## 6.1. Interim Operations.

- (a) Except as required by applicable Law or as expressly contemplated by this Agreement, the Company covenants and agrees as to itself and its Subsidiaries that, after the date of this Agreement and prior to the Effective Time, the business of it and its Subsidiaries shall be conducted in all material respects in the ordinary and usual course and, to the extent consistent therewith, it and its Subsidiaries shall use their respective commercially reasonable efforts to preserve their business organizations intact and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, licensors, licensees, distributors, creditors, lessors, employees and business associates (other than as announced by the Company prior to the date hereof). Without limiting the generality of, and in furtherance of, the foregoing, from the date of this Agreement until the Effective Time, except (A) as otherwise expressly required or permitted by this Agreement or as required by Law, (B) as Parent may approve in writing (such approval not to be unreasonably withheld or delayed) or (C) as set forth in Section 6.1 of the Company Disclosure Schedule, the Company will not and will not permit its Subsidiaries to:
- (i) adopt or propose any change in its certificate of incorporation or by-laws or other applicable governing instruments;
- (ii) merge or consolidate the Company or any of its Subsidiaries with any other Person, except for any such transactions among wholly owned Subsidiaries of the Company, or other than in the ordinary course restructure, reorganize or completely or partially liquidate or otherwise enter into any Contracts imposing material changes or restrictions on its assets, operations or businesses, other than in the ordinary course;
- (iii) acquire, directly or indirectly, whether by purchase, merger, consolidation or acquisition of stock or assets or otherwise, any assets, securities, properties, interests, or businesses or make any investment (whether by purchase of stock or securities, contributions to capital, loans to, or property transfers), in each case, other than (A) in the ordinary course of business (it being understood and agreed that the acquisition of all or substantially all of the assets of any Person is not in the ordinary course of business), or (B) if not in the ordinary course of business, with a value or purchase price (including the value of assumed liabilities) not in excess of \$50 million in any transaction or related series of transactions or acquisitions pursuant to Contracts in effect as of the date of this Agreement and listed on Section 6.1 of the Company Disclosure Schedule (it being understood that capital expenditures are not meant to be captured by this Section 6.1(a)(iii));
- (iv) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or encumbrance of, any shares of capital stock of the Company or any of its Subsidiaries (other than the issuances, sales, pledges, dispositions, grants, transfers, leases, licenses, guaranties or encumbrances of shares by a wholly owned Subsidiary of the Company to the Company or another wholly owned Subsidiary or the issuance or transfer of Shares pursuant to outstanding awards under Company Stock Plans), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities;

(v) create or incur (A) any lien or other security interest on any Company Intellectual Property owned or exclusively licensed or that is material and non-exclusively licensed by the Company or any of

A-23

#### **Table of Contents**

its Subsidiaries or (B) any Lien on any other assets of the Company or any of its Subsidiaries having a value in excess of \$50 million:

- (vi) make any loans, advances, guarantees or capital contributions to or investments in any Person (other than the Company or any direct or indirect wholly owned Subsidiary of the Company) in excess of \$10 million in the aggregate;
- (vii) declare, set aside, make or pay any dividend or other distribution, payable in cash, stock, property or otherwise, with respect to any of its capital stock (except for dividends paid by any Subsidiary to the Company or to any other Subsidiary, or regular quarterly dividends not to exceed \$0.37 per share, declared and paid consistent with prior timing) or enter into any Contract with respect to the voting of its capital stock;
- (viii) reclassify, split, combine, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock;
- (ix) incur, alter, amend or modify, any indebtedness for borrowed money or guarantee such indebtedness of another Person, or issue or sell any debt securities or warrants or other rights to acquire any debt security of the Company or any of its Subsidiaries, or permit any Subsidiary of the Company to guarantee any indebtedness of the Company, other than in the case of indebtedness for borrowed money incurred in the ordinary course of business consistent with past practices (A) that does not exceed \$200 million in the aggregate, (B) debt of the Company or a Subsidiary that is in replacement of existing indebtedness for borrowed money of the Company or such Subsidiary, as applicable, (C) guarantees incurred in compliance with this Section 6.1 by the Company of indebtedness of wholly owned Subsidiaries of the Company or (D) interest rate swaps on customary commercial terms consistent with past practice and in compliance with the Company s risk management policies in effect on the date of this Agreement and not to exceed \$500 million of notional debt in the aggregate;
- (x) make or authorize any capital expenditure in excess of \$875 million in the aggregate during any 12 month period;
- (xi) other than in the ordinary course of business or in connection with an acquisition permitted under Section 6.1(a)(iii), enter into any Contract that would have been a Material Contract had it been entered into prior to this Agreement;
- (xii) make any material changes with respect to accounting policies or procedures, except as required by changes in applicable generally accepted accounting principles;
- (xiii) settle any Action before a Governmental Entity for an amount in the aggregate in excess of the amount set forth on Section 6.1(a)(xiii) of the Company Disclosure Schedule (excluding amounts covered by insurance) or for any obligation or liability of the Company in excess of such amount or agree to any material limitation or restriction on any aspect of the conduct of the Company s or its Subsidiaries business (or, after giving effect to the Merger, Parent s or its Subsidiaries business);
- (xiv) other than in the ordinary course, amend or modify, in any material respect, or terminate any Material Contract or IP Contract, or cancel, modify or waive any material debts or claims held by it or waive any material rights;
- (xv) except as in Section 6.1(a)(xv) of the Company Disclosure Schedule, make any Tax election, amend any Tax Return, settle or finally resolve any controversy with respect to Taxes or change any method of Tax accounting in each case, where the sum of amount of Taxes in question with respect to all such actions exceeds \$50 million;

(xvi) (A) with regard to Intellectual Property owned by the Company or any of its Subsidiaries, transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, disclose, cancel, abandon or allow to lapse or expire or otherwise dispose of any material Intellectual Property, other than licenses or other non-material Contracts granted in the ordinary course of business, or cancellation, abandonment,

A-24

#### **Table of Contents**

allowing to lapse or expire such Intellectual Property that is no longer used or useful in any of the Company s or its Subsidiaries respective businesses or pursuant to Contracts in effect prior to the date of this Agreement; and (B) with regard to other assets, transfer, sell, lease, license, mortgage, pledge, surrender, encumber, divest, cancel, abandon or allow to lapse or expire or otherwise dispose of any material assets, licenses, operations, rights, product lines, businesses or interests therein of the Company or its Subsidiaries, including capital stock of any of its Subsidiaries, except in connection with services provided in the ordinary course of business and sales of obsolete assets and except for sales, leases, licenses or other dispositions of assets with a fair market value not in excess of \$50 million in the aggregate, other than pursuant to Contracts in effect prior to the date of this Agreement;

(xvii) except as required pursuant to existing written, binding plans or agreements in effect prior to the date of this Agreement or as set forth in Section 5.1(h)(i) of the Company Disclosure Schedule, or as otherwise required by applicable Law, (A) grant or provide any severance or termination payments or benefits to any director, officer or employee of the Company or any of its Subsidiaries, (B) increase the compensation, bonus or pension, welfare, severance or other benefits of, pay any bonus to, or make any new equity awards to any director, officer or employee of the Company or any of its Subsidiaries, in each case other than to non-officer employees in the ordinary course of business consistent with past practice, (C) establish, adopt, amend or terminate any Company Benefit Plan (except as required by Law) or amend the terms of any outstanding equity-based awards, (D) take any action to accelerate the vesting or payment, or fund or in any other way secure the payment, of compensation or benefits under any Company Benefit Plan, to the extent not already provided in any such Company Benefit Plan, (E) materially change any actuarial or other assumptions used to calculate funding obligations with respect to any Company Benefit Plan or to change the manner in which contributions to such plans are made or the basis on which such contributions are determined, except as may be required by GAAP, or (F) forgive any loans to directors, officers or employees of the Company or any of its Subsidiaries;

(xviii) agree, authorize or commit to do any of the foregoing.

- (b) Prior to making any widely distributed written communications to the employees of the Company or any of its Subsidiaries pertaining to material compensation or benefit matters that are affected by the transactions contemplated by this Agreement, the Company shall use its reasonable best efforts to provide Parent with a copy of the intended communication, Parent shall have a reasonable period of time to review and comment on the communication, and the Company shall consider such comments prior to distributing any such communication to its employees.
- (c) Parent and Merger Sub (1) shall not take, and Parent shall not permit any of its Subsidiaries to take, any action that is reasonably likely to prevent or materially impair the consummation of the Merger and (2) shall not enter into any acquisition agreement, or make any acquisition, that is reasonably likely to prevent, materially delay or impair the consummation of the Merger. Parent shall not issue any Parent Common Stock (or subscription rights with respect thereto) prior to obtaining the Requisite Parent Vote unless the Stichting agrees to vote any Parent Common Stock received by it, or received by others who are bound to vote together with the Stichting, in favor of the proposal to approve the Merger and the other transactions contemplated hereby.
- (d) Nothing contained herein shall give to Parent or Merger Sub, directly or indirectly, rights to control or direct the Company s operations prior to the Effective Time in violation of applicable Law. Prior to the Effective Time, the Company shall exercise, consistent with the terms and conditions hereof, complete control and supervision of its operations.

### 6.2. Acquisition Proposals.

(a) <u>No Solicitation or Negotiation</u>. The Company agrees that, except as expressly permitted by this Section 6.2, neither it nor any of its Subsidiaries nor any of the officers and directors of it or any of its Subsidiaries shall, and that

it shall use its reasonable best efforts to instruct and cause its and its Subsidiaries employees, investment bankers, attorneys, accountants and other advisors or representatives (such directors,

A-25

#### **Table of Contents**

officers, employees, investment bankers, attorneys, accountants and other advisors or representatives, collectively, **Representatives** ) not to, directly or indirectly:

- (i) initiate, solicit or knowingly encourage any inquiries or the making of any proposal or offer that constitutes, or could reasonably be expected to lead to, any Acquisition Proposal (as defined below);
- (ii) engage in, continue or otherwise participate in any discussions or negotiations regarding, or provide any non-public information or data to any Person relating to, any Acquisition Proposal; or
- (iii) otherwise knowingly facilitate any effort or attempt to make an Acquisition Proposal.

Notwithstanding anything in the foregoing to the contrary, prior to the time that, but not after, the Requisite Company Vote is obtained, if the Company has otherwise complied in all material respects with this Section 6.2, the Company may (A) provide information in response to a request therefor by a Person (other than any Affiliate of the Company) who has made an unsolicited written Acquisition Proposal that the Company Board of Directors believes in good faith to be bona fide providing for the acquisition of more than 50% of the assets (on a consolidated basis) or total voting power of the equity securities of the Company if the Company receives from the Person so requesting such information an executed confidentiality agreement on terms not more favorable to such other Person than those contained in the Confidentiality Agreement (as defined in Section 9.7); and promptly discloses (and, if applicable, provides copies of) any such information to Parent to the extent not previously provided to Parent; (B) engage or participate in any discussions or negotiations with any Person who has made such an unsolicited bona fide written Acquisition Proposal of the type described in clause (A) above; or (C) after having complied with Section 6.2(c), approve, recommend, or otherwise declare advisable or propose to approve, recommend or declare advisable (publicly or otherwise) such an Acquisition Proposal; (x) in each such case referred to in clause (A) or (B) above, the board of directors of the Company has determined in good faith based on the information then available and (after consultation with its financial advisor and outside legal counsel) that such Acquisition Proposal either constitutes a Superior Proposal (as defined below) or is reasonably likely to result in a Superior Proposal; and (y) in the case referred to in clause (C) above, the board of directors of the Company determines in good faith (after consultation with its financial advisor and outside legal counsel) that such Acquisition Proposal is a Superior Proposal.

### (b) *Definitions*. For purposes of this Agreement:

<u>Acquisition Proposal</u> means (i) any proposal or offer with respect to a merger, joint venture, partnership, consolidation, dissolution, liquidation, tender offer, recapitalization, reorganization, share exchange, business combination or similar transaction involving the Company or any of its Significant Subsidiaries and (ii) any acquisition by any Person, or proposal or offer, which if consummated would result in any Person becoming the beneficial owner of, directly or indirectly, in one or a series of related transactions, 15% or more of the total voting power or of any class of equity securities of the Company or those of any of its Significant Subsidiaries, or 15% or more of the consolidated total assets (including equity securities of its Subsidiaries) of the Company, in each case other than the transactions contemplated by this Agreement.

<u>Superior Proposal</u> means a bona fide Acquisition Proposal in connection with which there has been no material violation of this Section 6.2 that would result in any Person (or its stockholders) becoming the beneficial owner, directly or indirectly, of more than 50% of the assets (on a consolidated basis) or more than 50% of the total voting power of the equity securities of the Company that the board of directors of the Company has determined in its good faith judgment is reasonably likely to be consummated in accordance with its terms, taking into account all legal, financial and regulatory aspects of the proposal and the Person making the proposal, and if consummated, would result in a transaction more favorable to the Company s stockholders from a financial point of view than the transaction contemplated by this Agreement (after taking into account any revisions to the terms of the transaction

contemplated by Section 6.2(c) of this Agreement pursuant to Section 6.2(c)).

A-26

#### **Table of Contents**

- (c) <u>No Change of Recommendation or Alternative Acquisition Agreement</u>. The board of directors of the Company and each committee of the board of directors shall not:
- (i) withhold, withdraw, qualify or modify (or publicly propose or resolve to withhold, withdraw, qualify or modify), in a manner adverse to Parent, the Company Recommendation with respect to the Merger; or
- (ii) except as expressly permitted by, and after compliance with, Section 8.3(a) hereof, cause or permit the Company to enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other agreement (other than a confidentiality agreement referred to in Section 6.2(a) entered into in compliance with Section 6.2(a) (an <u>Alternative Acquisition Agreement</u>) relating to any Acquisition Proposal.

Notwithstanding anything to the contrary set forth in this Agreement, prior to the time, but not after, the Requisite Company Vote is obtained, the board of directors of the Company may withhold, withdraw, qualify or modify the Company Recommendation or approve, recommend or otherwise declare advisable any Superior Proposal not knowingly solicited, entered into or agreed to in breach of this Section 6.2 and made after the date of this Agreement, if the board of directors of the Company determines in good faith, after consultation with outside legal counsel, that failure to do so would be inconsistent with its fiduciary obligations under applicable Law (a **Change of Recommendation** ); provided, however, that no Change of Recommendation may be made until after at least 72 hours following Parent s receipt of notice from the Company advising that management of the Company currently intends to recommend to its board of directors that it take such action and the basis therefor, including all required information under Section 6.2(f). In determining whether to make a Change of Recommendation in response to a Superior Proposal, the board of directors of the Company shall take into account any changes to the terms of this Agreement proposed by Parent and any other information provided by Parent. Any material amendment to any Acquisition Proposal will be deemed to be a new Acquisition Proposal for purposes of this Section 6.2.

- (d) <u>Certain Permitted Disclosure</u>. Nothing contained in this Section 6.2 shall be deemed to prohibit the Company from complying with its disclosure obligations under United States federal or state Law with regard to an Acquisition Proposal; <u>provided</u>, <u>however</u>, that if such disclosure includes a Change of Company Recommendation or has the substantive effect of withdrawing or adversely modifying the Company Recommendation, such disclosure shall be deemed to be a Change of Recommendation and Parent shall have the right to terminate this Agreement as set forth in Section 8.4 (it being understood that any stop, look or listen communication that contains only the information set forth in Rule 14d-9(f) shall not be deemed a Change of Company Recommendation or be deemed to have the substantive effort of withdrawing or adversely modifying the Company Recommendation).
- (e) <u>Existing Discussions</u>. The Company agrees that it will immediately cease and cause to be terminated any existing activities, discussions or negotiations with any Persons conducted heretofore with respect to any Acquisition Proposal. The Company agrees that it will take the necessary steps to promptly inform the individuals or entities referred to in the first sentence of this Section 6.2(e) of the obligations undertaken in this Section 6.2 and in the Confidentiality Agreement. The Company also agrees that it will promptly request each Person that has executed a confidentiality agreement on or after January 1, 2007, in connection with such Person s consideration of acquiring the Company or any of its Subsidiaries to return or destroy all confidential information heretofore furnished to such Person by or on behalf of the Company or any of its Subsidiaries.
- (f) *Notice*. The Company agrees that it will promptly (and, in any event, within 24 hours) notify Parent if any proposals or offers with respect to an Acquisition Proposal are received by, any such information is requested from, or any such discussions or negotiation are sought to be initiated or continued with, it or any of its Representatives indicating, in connection with such notice, the name of such Person and the material terms and conditions of any proposals or offers (including, if applicable, copies of any written requests, proposals or offers, including proposed agreements) and thereafter shall keep Parent informed, on a reasonably current basis (which shall be considered in

light of the circumstances of such proposal or offer and the time by which Parent has the opportunity to respond), of the status and terms of any such proposals or offers (including any amendments thereto).

A-27

#### **Table of Contents**

## 6.3. Proxy Filing; Information Supplied.

- (a) The Company shall prepare and file with the SEC, as promptly as practicable after the date of this Agreement, a proxy statement in preliminary form relating to the Stockholders Meeting (as defined in Section 6.4(a)) (such proxy statement, including any amendment or supplement thereto, the **Proxy Statement** ). The Company agrees, as to it and its Subsidiaries, that (i) the Proxy Statement will comply in all material respects with the applicable provisions of the Exchange Act and the rules and regulations thereunder, and (ii) none of the information supplied by it or any of its Subsidiaries for inclusion or incorporation by reference in the Proxy Statement will, at the date of mailing to stockholders of the Company or at the time of the Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Notwithstanding the foregoing, (i) the Company makes no representation or warranty with respect to the information supplied by or on behalf of Parent or Merger Sub or any of their respective Representatives that is contained or incorporated by reference in the Proxy Statement and (ii) the Company represents and warrants with respect to any projected financial information provided by it or on its behalf, only that the information was prepared or approved in good faith by the Company s management on the basis of assumptions believed by the Company s management to be reasonable as of the time made. Parent agrees that none of the information supplied by it for inclusion or incorporation by reference in the Proxy Statement will, at the date of mailing to stockholders of the Company or at the time of the Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- (b) Parent agrees that none of the information supplied by it for inclusion or incorporation by reference in any written material to be furnished to shareholders of Parent in connection with the Parent Shareholders Meeting (the Parent Shareholder Meeting Materials ) or any prospectus that Parent would subsequently prepare and file in connection with the Equity Financing (the Parent Prospectus ) will, at the date of being put at the disposal of shareholders or investors, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Notwithstanding the foregoing, Parent makes no representation or warranty with respect to the information supplied by or on behalf of the Company or any of its Representatives that is contained or incorporated by reference in the Parent Shareholder Meeting Materials or the Parent Prospectus. The Company agrees that none of the information supplied by it for inclusion or incorporation by reference in the Parent Shareholder Meeting Materials or the Parent Prospectus will, at the date of being put at the disposal of shareholders or investors, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

#### 6.4. Stockholders Meeting.

- (a) The Company will take, in accordance with applicable Law and its certificate of incorporation and by-laws, all action necessary to convene a meeting of holders of Shares (the <u>Stockholders Meeting</u>) as promptly as practicable after the execution of this Agreement to consider and vote upon the adoption of this Agreement, and shall not postpone or adjourn such meeting except to the extent required by Law. Subject to Section 6.2(c) hereof, the board of directors of the Company shall recommend such adoption and shall take all lawful action to solicit such adoption of this Agreement. In the event that subsequent to the date hereof, the board of directors of the Company determines that this Agreement is no longer advisable and makes a Change of Recommendation, the Company shall nevertheless submit this Agreement to the holders of the Shares for adoption at the Stockholders Meeting unless this Agreement shall have been terminated in accordance with its terms prior to the Stockholders Meeting.
- (b) Parent will take, in accordance with applicable Law and its articles of association and by-laws, all actions necessary to convene a meeting of its shareholders at which a vote shall be taken (the **Parent Shareholders Meeting**)

as promptly as practicable after the execution of this Agreement to consider and vote upon the approval of the Merger and certain other matters contemplated by this Agreement, and shall not

A-28

#### **Table of Contents**

postpone or adjourn such meeting except to the extent required by Law. The board of directors of Parent shall recommend such approval and shall take all lawful action to solicit such approval of the Merger. To the extent permitted under applicable Law, Parent shall use its best efforts to obtain the approval of the Merger by its shareholders.

## 6.5. Filings; Other Actions; Notification.

- (a) <u>Proxy Statement</u>. The Company shall promptly notify Parent of the receipt of all comments of the SEC with respect to the Proxy Statement and of any request by the SEC for any amendment or supplement thereto or for additional information and shall promptly provide to Parent copies of all correspondence between the Company and/or any of its Representatives and the SEC with respect to the Proxy Statement. The Company and Parent shall each use its reasonable best efforts to promptly provide responses to the SEC with respect to all comments received on the Proxy Statement by the SEC and the Company shall cause the definitive Proxy Statement to be mailed as promptly as possible after the date the SEC staff advises that it has no further comments thereon or that the Company may commence mailing the Proxy Statement.
- (b) <u>Cooperation</u>. Subject to the terms and conditions set forth in this Agreement, the Company and Parent shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under this Agreement and applicable Laws to consummate and make effective the Merger and the other transactions contemplated by this Agreement as soon as practicable, including preparing and filing as promptly as practicable all documentation to effect all necessary notices, reports and other filings and to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity in order to consummate the Merger or any of the other transactions contemplated by this Agreement. Notwithstanding the foregoing, nothing in this Agreement, including this Section 6.5(b), shall require, or be construed to require, Parent to proffer to, or agree to, sell, divest, lease, license, transfer, dispose of or otherwise encumber or hold separate, or actually to, sell, divest, lease, license, transfer, dispose of or otherwise encumber or hold separate, before or after the Effective Time, any assets, licenses, operations, rights, product lines, businesses or interest therein of Parent, the Company or any of their respective Affiliates (or to consent to any sale, divestiture, lease, license, transfer, disposition or other encumberment by the Company of any of its assets, licenses, operations, rights, product lines, businesses or interest therein or to any agreement by the Company to take any of the foregoing actions) or to agree to any material changes (including through a licensing arrangement) or restriction on, or other impairment of, Parent s ability to own or operate any of such assets, licenses, operations, rights, product lines, businesses or interests therein or Parent s ability to vote, transfer, receive dividends or otherwise exercise full ownership rights with respect to the stock of the Surviving Corporation (any of the foregoing, **Detriments**); provided, however, that in order to ensure that Governmental Antitrust Entities with the authority to clear, authorize or otherwise approve the consummation of the Merger do so by the Termination Date, Parent shall agree to one or more Detriments provided that such Detriments would not, individually or in the aggregate, result in a reduction of assets, categories of assets, businesses or investments that generated in the aggregate more than 5% of the sum of the 2007 gross sales and investment income of Parent and the Company, including their respective Subsidiaries, on a combined basis net of any ongoing annual royalty payments or other ongoing revenue contributions expected to be received as a result of effectuating such Detriments ( Detriments Limit ). If Governmental Antitrust Entities require that Parent agree to Detriments that would in the aggregate exceed the Detriment Limit and if, as a result, any administrative or judicial action or proceeding is instituted or threatened to be instituted by one or more Governmental Antitrust Entities challenging the Merger as violative of law, Parent and the Company shall, and shall cause their respective Affiliates to, use their reasonable best efforts to cooperate and to contest and resist, except insofar as Parent and the Company otherwise may agree, one or more of such actions or proceedings; provided, however, that Parent in its sole discretion may decide which actions or proceedings to contest, if such selected actions or proceedings, if successfully contested, in the aggregate would result in the Detriments not exceeding the Detriments Limit. The

Company and Parent will each request early termination of the waiting period with respect to the Merger under the HSR Act. Subject to the terms and conditions set forth in this Agreement, without limiting the generality of the undertakings pursuant to this Section 6.5(b),

A-29

#### **Table of Contents**

each of the Company and Parent agree to promptly provide to each and every Governmental Antitrust Entity non-privileged information and documents (i) requested by any Governmental Antitrust Entity or (ii) that are necessary, proper or advisable to permit consummation of the transactions contemplated by this Agreement. Subject to applicable Laws relating to the exchange of information, Parent shall have the right to direct all matters with any Governmental Antitrust Entity, provided that it shall keep the Company informed and shall afford the Company a reasonable opportunity to participate therein. The Company shall not initiate any meeting or discussion with, or make any submission to, any Governmental Antitrust Entity with respect to any filings, applications, litigation, investigation, or other inquiry or proceeding regarding the Merger or filings under any pre-merger notification rules or in connection with any actual or threatened claim or proceeding unless Parent has approved in advance such initiation and the circumstances of such meeting, discussion or the content and submission of any such filing. The Company shall not participate in any meeting or discussion initiated by any Governmental Antitrust Entity with respect to any filings, applications, investigation, or other inquiry regarding the Merger or filings under any pre-merger notification rules, without giving Parent reasonable prior notice of the meeting or discussion and, to the extent permitted by the relevant Governmental Antitrust Entity, the opportunity to attend and participate. In addition, the Company shall not initiate any offer to any Governmental Antitrust Entity with respect to any proposed Detriment. Parent and the Company shall have the right to review in advance all of the information relating to Parent or the Company, as the case may be, and any of their respective Subsidiaries, that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Merger and the other transactions contemplated by this Agreement; provided, however, that the right to review shall not include any documents created in the ordinary course of business that are submitted to such Governmental Entity and each of Parent and the Company may limit access to work product and privileged materials to outside counsel and agreed upon inside counsel for Parent and the Company. In exercising the foregoing rights, each of the Company and Parent shall act reasonably and as promptly as practicable. If any Detriment agreed to by Parent requires action by or with respect to the Company or its Subsidiaries or its or their businesses or assets, and such action would constitute a breach of this Agreement, Parent hereby agrees to consent to the taking of such action by the Company and any such action may, at the discretion of the Company, be conditioned upon consummation of the Merger.

- (c) <u>Information</u>. The Company and Parent each shall, upon request by the other, furnish the other with all information concerning itself, its Subsidiaries, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with the Proxy Statement, the Parent Prospectus, the Parent Shareholders Meeting Materials or any other statement, filing, notice or application made by or on behalf of Parent, the Company or any of their respective Subsidiaries to any third party and/or any Governmental Entity in connection with the Merger and the transactions contemplated by this Agreement.
- (d) *Status*. Subject to applicable Laws and as required by any Governmental Entity, the Company and Parent each shall keep the other apprised of the status of matters relating to completion of the transactions contemplated hereby, including promptly furnishing the other with copies of notices or other communications received by Parent or the Company, as the case may be, or any of its Subsidiaries, from any Governmental Entity with respect to the Merger and the other transactions contemplated by this Agreement. The Company shall give prompt notice to Parent of any change, fact or condition that is reasonably expected to result in a Material Adverse Effect or of any failure of any condition to Parent so obligations to effect the Merger. Parent shall give prompt notice to the Company of any change, fact or condition that is reasonably expected to prevent or materially impair the consummation of the transactions contemplated by this Agreement or of any failure of any condition to the Company so obligations to effect the Merger.
- (e) <u>Antitrust Matters</u>. Subject to the terms and conditions set forth in this Agreement, without limiting the generality of the undertakings pursuant to this Section 6.5, each of the Company and Parent agrees to promptly provide to each and every federal, state, local or foreign court or Governmental Entity with jurisdiction over enforcement of any applicable antitrust or competition Laws (<u>Governmental Antitrust Entity</u>) non-privileged information and documents (i) requested by any Governmental Antitrust Entity or (ii) that are necessary, proper or advisable to permit

consummation of the transactions contemplated by this Agreement.

A-30

#### **Table of Contents**

- 6.6. Access and Reports. Subject to applicable Law, upon reasonable notice, the Company shall (and shall cause its Subsidiaries to) afford Parent s officers and other authorized Representatives reasonable access, during normal business hours throughout the period prior to the Effective Time, to its employees, properties, books, contracts and records and, during such period, the Company shall (and shall cause its Subsidiaries to) furnish promptly to Parent all information concerning its business, properties and personnel as may reasonably be requested, provided that no investigation pursuant to this Section 6.6 shall affect or be deemed to modify any representation or warranty made by the Company herein, and provided, further, that the foregoing shall not require the Company (i) to permit any inspection, or to disclose any information, that in the reasonable judgment of the Company would result in the disclosure of any trade secrets of third parties or violate any of its obligations with respect to confidentiality if the Company shall have used reasonable best efforts to obtain the consent of such third party to such inspection or disclosure or (ii) to disclose any privileged information of the Company or any of its Subsidiaries. All requests for information made pursuant to this Section 6.6 shall be directed to the executive officer or other Person designated by the Company. All such information shall be governed by the terms of the Confidentiality Agreement (as defined in Section 9.7).
- 6.7. <u>Stock Exchange Delisting</u>. Prior to the Closing Date, the Company shall cooperate with Parent and use reasonable best efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable on its part under applicable Laws and rules and policies of the NYSE to enable the delisting by the Surviving Corporation of the Shares from the NYSE and the deregistration of the Shares under the Exchange Act as promptly as practicable after the Effective Time.

#### 6.8. Publicity.

- (a) The initial press release regarding the Merger shall be a joint press release. After the initial press release, the Company and Parent each shall, to the extent feasible, consult with each other prior to issuing any press releases or otherwise making public announcements with respect to the Merger and the other transactions contemplated by this Agreement and prior to making any filings with any third party and/or any Governmental Entity (including any national securities exchange or interdealer quotation service) with respect thereto, except as may be required by Law or by obligations pursuant to any listing agreement with or rules of any national securities exchange or interdealer quotation service or by the request of any Governmental Entity.
- (b) Notwithstanding the foregoing, each of Parent and the Company may (i) make or cause to be made any press release or similar public announcement or communication as may be required to comply with the requirements of any applicable Laws or the rules and regulations of each stock exchange upon which the securities of one of the parties is listed and (ii) disclose any information concerning the transactions contemplated hereby which it deems appropriate in its reasonable judgment, in light of its status as a publicly owned company, including to securities analysts, institutional investors, in press interviews and to employees; provided, that, to the extent practicable, Parent and the Company will try in good faith to remain within the bounds of the parties prior disclosures; provided, further, that in the case of clauses (i) and (ii) above to the extent in the good faith judgment of such party it is reasonably practicable to do so), such party (x) provides the other party with a reasonable opportunity in light of the circumstances to review such party s intended communication and (y) consider in good faith modifications to the intended communication that are requested by the other party.

#### 6.9. Employee Benefits.

(a) Parent acknowledges that approval of this Agreement by stockholders of the Company or consummation of the Merger, as applicable, shall constitute a change in control for purposes of all applicable Company Benefit Plans.

(b) Parent agrees that, during the period commencing at the Effective Time and ending on the later of (i) the first anniversary of the Effective Time and (ii) December 31, 2009, the employees of the Company and its Subsidiaries will continue to be provided with compensation and employee benefits (under compensation and employee benefit plans) which are not less favorable in the aggregate than those provided by the Company and its Subsidiaries to such employees as of immediately prior to the date hereof. Parent shall cause any compensation and employee benefit plans which the employees of the Company and its Subsidiaries are

A-31

#### **Table of Contents**

entitled to participate in to credit, for purposes of eligibility and vesting thereunder (but not for purposes of benefit accrual), service by employees of the Company and its Subsidiaries as if such service were with Parent, to the same extent such service was credited under a comparable plan of the Company. Notwithstanding the foregoing, nothing contained herein shall (1) be treated as an amendment of any particular Company Benefit Plan, (2) give any third party any right to enforce the provisions of this Section 6.9 or (3) obligate Parent, the Surviving Corporation or any of their Affiliates to (A) maintain any particular benefit plan or (B) retain the employment of any particular employee.

- (c) Parent shall, or shall cause the Surviving Corporation or Parent s or the Surviving Corporation s Subsidiaries, as applicable, to honor the terms of all Company Benefit Plans existing as of immediately prior to the Effective Time, including the Company Benefit Plans set forth on Section 6.9(c) of the Company Disclosure Schedule.
- (d) Except as set forth on Section 6.9(d) of the Company Disclosure Schedule, for non-union employees of the Company and its Subsidiaries who are involuntarily terminated without cause during the period beginning on the Effective Time and ending on the later of (x) one year from the Effective Time and (y) December 31, 2009, other than those employees who are given a notice of termination prior to the Effective Time. Following the Closing, Parent shall, or shall cause the Surviving Corporation or Parent s or the Surviving Corporation s Subsidiaries, as applicable, to pay severance benefits under such circumstances and in such amounts to such employees that are not less favorable than the severance benefits payable under the Company s Severance Pay Program as in effect immediately prior to the Effective Time. For purposes of the foregoing, an employee who elects to terminate employment pursuant to an early retirement incentive program shall not be deemed to have been involuntarily terminated without cause.
- (e) Parent shall, or shall cause the Surviving Corporation or Parent s or the Surviving Corporation s Subsidiaries, as applicable, to pay the annual bonuses for the 2008 calendar year to employees of the Company and its Subsidiaries who (1) remain employed through December 31, 2008 or (2) are involuntarily terminated without cause, between the Effective Time and December 31, 2008 (other than employees who are given a notice of termination prior to the Effective Time). Such bonuses shall be determined as follows: (i) the formula portion shall be payable based on the performance for the 2008 year in accordance with the Company s practices and policies in effect on the date hereof (including the Company s practice with respect to applying the formula) and (ii) the discretionary portion shall be payable at not less than the target rate for the particular employee. The aggregate bonus shall be payable in accordance with Company practice (reduced by the portion which, pursuant to the applicable Company Benefit Plans, is paid on the change in control and any other portion that is paid prior to the Effective Time). For purposes of the foregoing, an employee who elects to terminate employment pursuant to an early retirement incentive program shall not be deemed to have been involuntarily terminated without case.
- (f) Parent shall, or shall cause the Surviving Corporation or Parent s or the Surviving Corporation s Subsidiaries, as applicable, to: (i) waive any preexisting condition limitations otherwise applicable to employees of the Company and its Subsidiaries and their eligible dependents under any plan of Parent or any Subsidiary of Parent that provides health benefits in which Company Employees may be eligible to participate following the Closing Date, other than any limitations that were in effect with respect to such employees as of the Effective Time under the analogous Company Benefit Plan, (ii) honor any deductible, co-payment and out-of-pocket maximums incurred by the employees of the Company and its Subsidiaries and their eligible dependents under the health plans in which they participated immediately prior to the Effective Time during the portion of the calendar year prior to the Effective Time in satisfying any deductibles, co-payments or out-of-pocket maximums under health plans of Parent, the Surviving Corporation or any of their respective Subsidiaries in which they are eligible to participate after the Effective Time in the same plan year in which such deductibles, co-payments or out-of-pocket maximums were incurred and (iii) waive any waiting period limitation or evidence of insurability requirement that would otherwise be applicable to an employee of the Company and its Subsidiaries and his or her eligible dependents on or after the Effective Time, in each case to the extent such employee of the Company and its Subsidiaries or eligible dependent had satisfied any similar limitation or requirement under an analogous Company Benefit Plan prior to the Effective Time.

# A-32

#### **Table of Contents**

(g) The Surviving Corporation and its Subsidiaries, as the case may be, shall honor the terms of all Company Labor Agreements existing as of immediately prior to the Effective Time, including the Company Labor Agreements set forth on Section 6.9(g) of the Company Disclosure Schedule.

## 6.10. Debt and Equity Financing.

- (a) Parent shall satisfy on a timely basis all conditions applicable to Parent under the Facilities and shall obtain and consummate the financing contemplated by the Facilities on the Financing Date or, if such financing is unavailable, obtain and consummate alternative financing. Parent shall give the Company prompt notice upon becoming aware of any termination of the Facilities other than expiry in accordance with their terms on March 20, 2009 and of the occurrence of any event or circumstance that would prevent Parent from satisfying one or more of the conditions to initial utilization in either of the Facilities. In the event that Parent becomes aware of any event or circumstance that makes any portion of the financing of the Facilities unavailable prior to the Termination Date on the terms and conditions set forth in the Facilities, Parent shall secure as promptly as practicable any such unavailable portion from alternative sources.
- (b) Prior to the Closing, the Company shall use commercially reasonable efforts to provide Parent and its Representatives with reasonable access during normal business hours, to Parent s books and records, personnel and auditors and shall provide (and shall use all reasonable commercial efforts to cause its Representatives to provide) to Parent and Merger Sub, the following cooperation, as reasonably requested by Parent in connection with the Facilities and any equity or equity-linked financings required hereunder (\_Equity Financings ) (provided that such requested cooperation does not unreasonably interfere with the on-going operations of the Company and its Subsidiaries), (i) using commercially reasonable efforts to make its personnel reasonably available for participation in meetings, presentations, road shows, due diligence sessions (including accounting due diligence sessions) and sessions with rating agencies; (ii) using commercially reasonable efforts to provide Parent with all information within the Company s possession as is reasonably requested by Parent for the preparation of materials for rating agency presentations, offering documents, private placement memoranda, bank information memoranda, prospectuses and similar documents required in connection therewith (including preparation of combined financial information on a pro forma basis); (iii) using commercially reasonable efforts to provide Parent with all information within the Company s possession as is reasonably requested by Parent for preparation of additional definitive financing documents, and consider in good faith Parent s reasonable requests for certificates, legal opinions or documents as may be reasonably required under the Facilities or in connection with the Equity Financing (it being understood that the failure of the Company to provide any such certificate, legal opinion or document will not constitute a breach of this Agreement by the Company); (iv) furnishing Parent and its lenders or the underwriters of the Equity Financing with financial and other pertinent information regarding the Company s and its Subsidiaries businesses that is within the Company s possession as may be reasonably requested by Parent (including providing monthly financial statements (excluding footnotes) and in the form that the Company customarily prepares such financial statements within the time such statements are customarily prepared); (v) using commercially reasonable efforts to enable Parent to obtain accountants comfort letters, as reasonably requested by Parent (whether addressed to the underwriters of the Equity Financing and/or Parent) (it being understood that the failure of the Company s accountants to agree to provide such comfort letters or the failure to provide such comfort letters will not constitute a breach of this Agreement by the Company, in each case so long as the Company has used such commercially reasonable efforts); and (vi) using commercially reasonable efforts to permit the lenders under the Facilities or the underwriters of the Equity Financing, as well as Parent to evaluate the assets of the Company and its Subsidiaries, and the cash management and accounting systems, policies and procedures associated therewith. Notwithstanding anything to the contrary, nothing in this Agreement or Section 6.10(b) shall require any cooperation or assistance by the Company, its Subsidiaries or any of their respective Representatives with respect to any action set forth in this Section 6.10(b) to the extent it would, in the Company s reasonable judgment, interfere unreasonably with the business or operations of the Company or any of its Subsidiaries or would require any action of the Board of Directors of the Company or any of its Subsidiaries.

(c) Each of Parent and Merger Sub acknowledges and agrees that the Closing is not conditioned on the availability of the funding under the Facilities or any alternative financing arrangement.

A-33

#### **Table of Contents**

- (d) Parent acknowledges and agrees that the Company and its Affiliates and their respective Representatives shall not have any responsibility for, or incur any liability to any Person under, the Facilities or any financing that Parent may raise in connection with the transactions contemplated by this Agreement, and that Parent and Merger Sub shall indemnify and hold harmless the Company, its Affiliates and its Representatives from and against any and all losses, damages, claims, costs or expenses suffered or incurred by any of them in connection with the Financing and any information utilized in connection therewith (other than, in the case of the Company, to the extent that any such matters arise from a knowing and material breach of the Company s obligations under this Section 6.10(b)). Parent shall promptly, upon request by the Company, reimburse the Company for all reasonable out of pocket third party costs incurred by the Company or any of its Subsidiaries in connection with cooperation pursuant to this Section 6.10.
- 6.11. *Expenses*. The Surviving Corporation shall pay all charges and expenses, including those of the Paying Agent, in connection with the transactions contemplated in Article IV, and Parent shall reimburse the Surviving Corporation for such charges and expenses. Except as otherwise provided in Section 6.10 and Section 8.5, whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the Merger and the other transactions contemplated by this Agreement shall be paid by the party incurring such expense.
- 6.12. Indemnification; Directors and Officers Insurance. (a) From and after the Effective Time, each of Parent and the Surviving Corporation agrees that it will indemnify, defend and hold harmless each present and former director and officer of the Company or any of its Subsidiaries and any fiduciary under any Company Plan (in each case, when acting in such capacity), determined as of the Effective Time (the **Indemnified Parties** ), against any costs or expenses (including attorneys fees and disbursements), judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the Indemnified Party is or was an officer, director, employee or fiduciary of the Company or any of its Subsidiaries or a fiduciary under any Company Plan, whether asserted or claimed prior to, at or after the Effective Time (including with respect to any acts or omissions in connection with this Agreement and the transactions and actions contemplated hereby), to the fullest extent that the Company would have been permitted under Delaware law and its certificate of incorporation or by-laws (and, to the extent not contrary to Delaware law or its certificate of incorporation, any indemnification agreement) in effect on the date of this Agreement to indemnify such Person (and Parent or the Surviving Corporation shall also promptly advance expenses as incurred to the fullest extent that the Company would have been permitted under Delaware law or its certificate of incorporation or by-laws (and, to the extent not contrary to Delaware law or its certificate of incorporation, any indemnification agreement) in effect on the date of this Agreement; provided that the Person to whom expenses are advanced provides an undertaking, if and only to the extent required by Delaware law of the Company s certification of incorporation or by-laws, to repay such advances if it is ultimately determined that such Person is not entitled to indemnification); and provided, further, that any determination required to be made with respect to whether an officer s or director s conduct complied with the standards set forth under Delaware law and the Company s certificate of incorporation and by-laws shall be made by independent counsel selected by the Indemnified Party. In the event of any claim, action, suit, proceeding or investigation, (x) neither Parent nor the Surviving Corporation shall settle, compromise or consent to the entry of any judgment in any claim, action, suit, proceeding or investigation (and in which indemnification could be sought by Indemnified Parties hereunder), unless such settlement, compromise or consent includes an unconditional release of such Indemnified Party from all liability arising out of such claim, action, suit, proceeding or investigation or such Indemnified Party otherwise consents, and (y) the Surviving Corporation shall cooperate in the defense of such matter. For the avoidance of doubt, the parties agree that this Section 6.12 does not purport to limit any rights that any Indemnified Party may have under any employment agreement, indemnification agreement or Company Plan.
- (b) The Charter and By-laws of the Surviving Corporation shall contain provisions no less favorable with respect to indemnification and advancement of expenses of individuals who were directors and officers prior to the Effective Time than are set forth, as of the date of this Agreement, in the Company s certificate of incorporation and by-laws,

which provisions shall not be amended, repealed or otherwise modified for a period

A-34

#### **Table of Contents**

of six years from the Effective Time in any manner that would adversely affect the rights thereunder of any such individual.

- (c) Prior to the Effective Time, the Company shall and, if the Company is unable to, Parent shall cause the Surviving Corporation as of the Effective Time to obtain and fully pay for tail insurance policies (providing only for the Side A coverage for Indemnified Parties where the existing policies also include Side B coverage for the Company) with a claims period of at least six years from and after the Effective Time from an insurance carrier with the same or better credit rating as the Company s current insurance carrier with respect to directors and officers liability insurance and fiduciary liability insurance (collectively, **D&O Insurance**), for the persons who, as of the date of this Agreement, are covered by the Company s existing D&O Insurance, with terms, conditions, retentions and levels of coverage at least as favorable as the Company s existing D&O Insurance with respect to matters existing or occurring at or prior to the Effective Time (including in connection with this Agreement or the transactions or actions contemplated hereby), and Parent shall cause the Surviving Corporation to maintain such D&O Insurance in full force and effect for their full terms. If the Company and the Surviving Corporation for any reason fail to obtain such tail insurance policies as of the Effective Time, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, continue to maintain in effect, at no expense to the beneficiaries, for a period of at least six years from and after the Effective Time for the persons who, as of the date of this Agreement, are covered by the Company s D&O Insurance in place as of the date of this Agreement with terms, conditions, retentions and levels of coverage at least as favorable as provided in the Company s existing policies as of the date of this Agreement, or, if such insurance is unavailable, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, purchase the best available D&O Insurance for such six-year period from an insurance carrier with the same or better credit rating as the Company s current insurance carrier with respect to the Company s existing D&O Insurance with terms, conditions, retentions and with levels of coverage at least as favorable as provided in the Company s existing policies as of the date of this Agreement. Notwithstanding anything in the foregoing, in no event shall Parent or the Surviving Corporation be required to expend for such policies an annual premium amount in excess of 300% of the annual premiums currently paid by the Company for such insurance; and provided further, that if the annual premiums of such insurance coverage exceed such amount, the Surviving Corporation shall obtain a policy with the greatest coverage available for a cost not exceeding such amount.
- (d) If Parent or the Surviving Corporation or any of their respective successors or assigns (i) shall consolidate with or merge into any other corporation or entity and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any individual, corporation or other entity, then, and in each such case, proper provisions shall be made so that the successors and assigns of Parent or the Surviving Corporation shall assume all of the obligations set forth in this Section 6.12.
- (e) The provisions of this Section 6.12 are intended to be for the benefit of, and shall be enforceable by, each of the Indemnified Parties and their respective successors, heirs and legal representatives, shall be binding on all successors and assigns of Parent and the Surviving Corporation and shall not be amended in any matter that is adverse to the Indemnified Parties (including their successors, heirs and legal representatives) without the consent of the Indemnified Party (including the successors, heirs and legal representatives) affected thereby.
- (f) The rights of the Indemnified Parties under this Section 6.12 shall be in addition to any rights such Indemnified Parties may have under the certificate of incorporation or by-laws of the Company or any of its Subsidiaries, or under any applicable Contracts or Laws, and Parent shall, and shall cause the Surviving Corporation to, honor and perform under all indemnification agreements entered into by the Company or any of its Subsidiaries.
- 6.13. <u>Takeover Statutes</u>. If any Takeover Statute is or may become applicable to the Merger or the other transactions contemplated by this Agreement, the Company and its board of directors shall grant all such approvals and take all such actions as are necessary or advisable so that such transactions may be consummated as promptly as practicable

on the terms contemplated by this Agreement and otherwise act to

A-35

#### **Table of Contents**

eliminate or minimize the effects of such statute, regulation or provision in the Company s certificate of incorporation or by-laws on such transactions.

6.14. *Communications*. As promptly as practicable following the date of this Agreement and in compliance with applicable Laws, Parent and the Company shall develop a joint plan for communication to the Company s employees, wholesalers and other strategic Persons about this Agreement and the transactions contemplated by this Agreement.

## 6.15. Release and Termination of Litigation.

- (a) Immediately after execution of this Agreement, the parties hereto shall voluntarily dismiss, with each party bearing its own costs and litigation expenses, all actions or proceedings pending between themselves and their affiliates related in any way to the removal or removability of directors or the transactions contemplated by this Agreement, and each party shall thereafter sign and deliver all such further papers as may be necessary in connection with such dismissals. From and after the Effective Time, each party hereto shall be deemed to have forever released and discharged the other party from and against all claims or causes of action that were or could have been asserted against one another in all such actions or proceedings.
- (b) Parent hereby withdraws its requests, dated July 8, 2008, that (i) the Company set a record date in connection with Parent s previously announced consent solicitation to remove and replace the Company s directors (the Consent Solicitation ) and (ii) the Company furnish Parent with certain stockholder list material pursuant to DGCL Section 220. Parent further agrees to withdraw and terminate the Consent Solicitation and not to pursue or take any action in furtherance thereof so long as this Agreement is in effect.
- 6.16. *Retention Plan*. As promptly as practicable after the date hereof, Parent and the Company shall cooperate with each other to establish an employee retention program as detailed in Section 6.16 of the Company Disclosure Schedule. Parent and the Company shall establish as at the Closing Date such retention program.

#### 6.17. Other Matters.

- (a) <u>Headquarters</u>. Effective upon the Closing, the Company s current headquarters in St. Louis, Missouri will be the Surviving Corporation s headquarters, such headquarters will also be Parent s headquarters for North America (excluding Cuba) and the global home of the flagship Budweiser brand.
- (b) *Name of Parent*. Effective upon the Closing, the current name of the Company shall be the name of the Surviving Corporation, and the name of Parent shall be amended to be Anheuser-Busch InBev N.V./S.A.
- (c) <u>Board Representation</u>. Parent shall, after consultation with the board of directors of the Company, nominate and cause to be elected two current or former directors of the Company to the board of directors of Parent, and each such director shall be confirmed for a three-year term at the first annual general meeting of Parent following the Closing.
- (d) <u>Preservation of Company Heritage</u>. Following the Closing, Parent shall cause the Surviving Corporation to preserve the Company s heritage and continue to support philanthropic and charitable causes in St. Louis and other communities in which the Company operates, including The Grant s Farm and the Clydesdales operations.
- (e) <u>U.S. Breweries</u>. Following the Closing, Parent will confirm the Surviving Corporation s good faith commitment that it will not close any of the Company s current twelve (12) breweries located in the United States, provided there are no new or increased federal or state excise taxes or other unforeseen extraordinary events which negatively impact the Company s business.

(f) <u>Wholesalers</u>. Parent reaffirms its commitment to the three-tier distribution system in the United States and agrees to work with the Company s existing wholesaler panel to strengthen the relationship between the Surviving Corporation and its wholesalers.

A-36

#### **Table of Contents**

(g) <u>Stadium</u>. The Company will honor its obligations under the Naming Rights and Sponsorship Agreement, dated August 3, 2004, as amended, between Busch Media Group, Inc., as authorized agent for Anheuser-Busch, Incorporated and Cardinals Ballpark, LLC relating to Busch Stadium.

### ARTICLE VII

#### Conditions

- 7.1. <u>Conditions to Each Party</u> s <u>Obligation to Effect the Merger</u>. The respective obligation of each party to effect the Merger is subject to the satisfaction or waiver at or prior to the Effective Time of each of the following conditions:
- (a) <u>Stockholder Approvals</u>. This Agreement shall have been duly adopted by holders of Shares constituting the Requisite Company Vote in accordance with applicable Law and the certificate of incorporation and by-laws of the Company. The holders of shares of Parent Common Stock constituting the Requisite Parent Vote, at the Parent Shareholders Meeting, shall have adopted a resolution approving the Merger (including the change of name of Parent).
- (b) <u>Regulatory Consents</u>. The waiting period applicable to the consummation of the Merger under the HSR Act shall have expired or been earlier terminated and all other Company Approvals and Parent Approvals (excluding the Equity Financing Filings) shall have been obtained or received without exceeding the Detriments Limit.
- (c) <u>No Injunction</u>. No court or other Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any Law (whether temporary, preliminary or permanent) that is in effect and restrains, enjoins or otherwise prohibits consummation of the Merger or the other transactions contemplated by this Agreement (collectively, an <u>Order</u>).
- 7.2. <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to effect the Merger are also subject to the satisfaction or waiver by Parent at or prior to the Effective Time of the following conditions:
- (a) Representations and Warranties. (i) The representations and warranties of the Company set forth in this Agreement that are qualified by reference to Material Adverse Effect shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date); (ii) the representations and warranties of the Company set forth in this Agreement that are not qualified by reference to Material Adverse Effect shall be true and correct as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date), provided, however, that notwithstanding anything herein to the contrary, the condition set forth in this Section 7.2(a)(ii) shall be deemed to have been satisfied even if any representations and warranties of the Company (other than those set forth in Section 5.1(b)(i) (solely with respect to the first four sentences), Section 5.1(c) (Corporate Authority; Approval and Fairness) and Section 5.1(l) (Takeover Statutes), which must be true and correct in all material respects) are not so true and correct, unless the failure of such representations and warranties of the Company to be so true and correct, individually or in the aggregate, has had or is reasonably likely to have a Material Adverse Effect; and (iii) Parent shall have received at the Closing a certificate signed on behalf of the Company by the Chief Executive Officer of the Company to the effect that such executive officer has read this Section 7.2(a) and the conditions set forth in this Section 7.2(a) have been satisfied.

(b) <u>Performance of Obligations of the Company</u>. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent shall have received a certificate signed on behalf of the Company by the Chief Executive Officer of the Company to such effect.

A-37

#### **Table of Contents**

- (c) <u>No Material Adverse Effect</u>. Since the date of this Agreement, there shall not have occurred any change, event, circumstance or development that has had, or is reasonably likely to have, a Material Adverse Effect.
- 7.3. <u>Conditions to Obligation of the Company</u>. The obligation of the Company to effect the Merger is also subject to the satisfaction or waiver by the Company at or prior to the Effective Time of the following conditions:
- (a) <u>Representations and Warranties</u>. (i) The representations and warranties of Parent set forth in this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of such date and time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date), and (ii) the Company shall have received at the Closing a certificate signed on behalf of Parent by the Chief Executive Officer of Parent to the effect that such Chief Executive Officer has read this Section 7.3(a) and the conditions set forth in this Section 7.3(a) have been satisfied.
- (b) <u>Performance of Obligations of Parent and Merger Sub</u>. Each of Parent and Merger Sub shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and the Company shall have received a certificate signed on behalf of Parent and Merger Sub by the Chief Executive Officer of Parent to such effect.

### **ARTICLE VIII**

### **Termination**

- 8.1. <u>Termination by Mutual Consent</u>. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after the adoption of this Agreement by the stockholders of the Company referred to in Section 7.1(a), by mutual written consent of the Company and Parent by action of their respective boards of directors.
- 8.2. Termination by Either Parent or the Company. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time by action of the board of directors of either Parent or the Company if (a) the Merger shall not have been consummated by March 19, 2009 (the \_\_Termination Date\_\_\_\_\_), whether such date is before or after the date of adoption of this Agreement by the stockholders of the Company or Parent referred to in Section 7.1(a), (b) the adoption of this Agreement by the stockholders of the Company referred to in Section 7.1(a) shall not have been obtained at the Stockholders Meeting or at any adjournment or postponement of the Stockholders Meeting, (c) the adoption of a resolution approving the Merger referred to in Section 7.1(a) shall not have been obtained at the Parent Shareholders Meeting (provided, that Parent shall not have the right to terminate this Agreement if it fails to obtain the Requisite Parent Approval to the extent that the Stichting is in breach of the Parent Shareholder Commitment) or (d) any Order permanently restraining, enjoining or otherwise prohibiting consummation of the Merger shall become final and non-appealable (whether before or after the adoption of this Agreement by the stockholders of the Company or Parent referred to in Section 7.1(a)); provided that the right to terminate this Agreement pursuant to this Section 8.2 shall not be available to any party that has breached in any material respect its obligations under this Agreement in any manner that shall have resulted in the occurrence of the failure of a condition to the consummation of the Merger.
- 8.3. <u>Termination by the Company</u>. This Agreement may be terminated by the Company and the Merger may be abandoned:
- (a) at any time prior to (but not after) the time the Requisite Company Vote is obtained, if (i) the board of directors of the Company authorizes the Company, subject to complying with the terms of this Agreement, to enter into an

Alternative Acquisition Agreement with respect to a Superior Proposal and the Company notifies Parent in writing that it intends to enter into such an agreement, attaching the most current version of such agreement to such notice (it being understood that giving Parent such notice does not entitle Parent to claim that a Change of Recommendation has occurred), (ii) 72 hours have passed since Parent s receipt of the Company s written notification of its intention to enter into a binding agreement for a

A-38

#### **Table of Contents**

Superior Proposal, and the board of directors of the Company determines, in good faith after consultation with its financial advisors, that the Superior Proposal remains a Superior Proposal after taking into account any revised offer that may be made by Parent, and (iii) the Company concurrently with such termination pays to Parent in immediately available funds any fees required to be paid pursuant to Section 8.5. The Company agrees (x) that it will not enter into the binding agreement referred to in clause (ii) above until after the 72 hour period described in clause (iii) above has terminated and (y) during such 72 hour period, to negotiate in good faith with Parent with respect to any revisions to the terms of the transaction contemplated by this Agreement proposed by Parent in response to a Superior Proposal, if any.

- (b) if there has been a breach of any representation, warranty, covenant or agreement made by Parent or Merger Sub in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Section 7.3(a) or 7.3(b) would not be satisfied and such breach or condition is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by the Company to Parent and (y) the Termination Date.
- (c) if all of the conditions set forth in Section 7.1 and Section 7.2 have been satisfied (other than those conditions that by their nature are to be satisfied by action taken at the Closing, <u>provided</u> that such conditions are capable of being satisfied) and Parent or Merger Sub has failed to consummate the Merger within thirty (30) days after satisfaction of such conditions.
- 8.4. *Termination by Parent*. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time by action of the board of directors of Parent if (a) the board of directors of the Company shall have made a Change of Recommendation, (b) at any time following receipt of an Acquisition Proposal, the Company s board of directors shall have failed to reaffirm its approval or recommendation of this Agreement and the Merger as promptly as practicable (but in any event within ten (10) business days after receipt of any reasonable written request to do so from Parent), (c) a tender offer or exchange offer for outstanding shares of the Company s common stock shall have been publicly disclosed (other than by Parent or an Affiliate of Parent) and, prior to the earlier of (x) the date prior to the date of the Stockholders Meeting and (y) eleven (11) business days after the commencement of such tender or exchange offer pursuant to Rule 14d-2 under the Exchange Act, the Company s board of directors fails at that time or at any time thereafter to recommend unequivocally against acceptance of such offer or (d) there has been a breach of any representation, warranty, covenant or agreement made by the Company in this Agreement, or any such representation and warranty shall have become untrue after the date of this Agreement, such that Section 7.2(a) or 7.2(b) would not be satisfied and such breach or condition is not curable or, if curable, is not cured within the earlier of (x) thirty (30) days after written notice thereof is given by Parent to the Company and (y) the Termination Date.
- 8.5. Effect of Termination and Abandonment. (a) Except as provided in paragraphs (b) or (c) below, in the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article VIII, this Agreement shall become void and of no effect with no liability to any Person on the part of any party hereto (or of any of its Representatives or Affiliates); provided, however, and notwithstanding anything in the foregoing to the contrary, that (i) no such termination shall relieve (x) any party hereto of any liability or damages to the other party hereto resulting from any willful or intentional material breach of this Agreement or (y) Parent or Merger Sub of any liability or damages to the Company resulting from the failure by Parent or Merger Sub to obtain the Financing (including a breach of Section 5.2(d) and Section 6.10) and (ii) the provisions set forth in this Section 8.5 and the second sentence of Section 9.1 shall survive the termination of this Agreement.
- (b) In the event that (i) a bona fide Acquisition Proposal shall have been made to the Company or any of its Subsidiaries or to its stockholders generally or any Person shall have publicly announced an intention (whether or not conditional) to make a bona fide Acquisition Proposal with respect to the Company or any of its Subsidiaries (and such Acquisition Proposal or publicly announced intention shall not have been publicly withdrawn without

qualification at least (A) thirty (30) business days prior to, with respect to any termination pursuant to Section 8.2(a), the date of termination, and (B) at least ten (10) business days prior to, with respect to termination pursuant to Section 8.2(b), the date of the Stockholders Meeting) and thereafter this Agreement is terminated by either Parent or the Company pursuant to Section 8.2(a) (Drop Dead) or 8.2(b) (No Stockholder Approval), (ii) this Agreement is terminated (A) by Parent pursuant to Section 8.4 or (B) by the Company pursuant to Section 8.2(b) and, on or prior to the date of the Stockholders Meeting, any event giving

A-39

#### **Table of Contents**

rise to Parent s right to terminate under Section 8.4 shall have occurred or (iii) this Agreement is terminated by the Company pursuant to Section 8.3(a) (Fiduciary Out), then, in each case, the Company shall promptly, but in no event later than two days after the date of such termination, pay Parent a termination fee of \$1,250,000,000 (the <u>Termination Fee</u> ) (provided, however, that the Termination Fee to be paid pursuant to clause (iii) shall be paid as set forth in Section 8.3) payable by wire transfer of same day funds; provided, however, that no Termination Fee shall be payable to Parent pursuant to clause (i) of this paragraph (b) unless and until within 12 months of such termination, (1) the Company or any of its Subsidiaries shall have entered into an Alternative Acquisition Agreement with respect to, or shall have consummated or shall have approved or recommended to the Company s stockholders, an Acquisition Proposal or (2) there shall have been consummated an Acquisition Proposal (substituting in both instances 50% for 15%, it being understood for the avoidance of doubt that any series of related transactions shall be aggregated and considered as a whole in determining such percentages in the definition of Acquisition Proposal ). The Company acknowledges that the agreements contained in this Section 8.5(b) are an integral part of the transactions contemplated by this Agreement, and that, without these agreements, Parent and Merger Sub would not enter into this Agreement; accordingly, if the Company fails to promptly pay the amount due pursuant to this Section 8.5(b), and, in order to obtain such payment, Parent or Merger Sub commences a suit that results in a judgment against the Company for the fee set forth in this Section 8.5(b) or any portion of such fee, the Company shall pay to Parent or Merger Sub its costs and expenses (including attorneys fees) in connection with such suit, together with interest on the amount of the fee at the prime rate published in the Money Rates section of The Wall Street Journal in effect on the date such payment was required to be made. Notwithstanding anything to the contrary in this Agreement, the parties hereby acknowledge that in the event that the Termination Fee becomes payable and is paid by the Company pursuant to this Section 8.5(b), the Termination Fee shall be Parent s and Merger Sub s sole and exclusive remedy for monetary damages under this Agreement.

(c) In the event that this Agreement is terminated by the Company or Parent pursuant to Section 8.2(c), then, Parent shall promptly, but in no event later than two days after the date of such termination, pay the Company the Termination Fee, payable by wire transfer of same day funds. Parent acknowledges that the agreements contained in this Section 8.5(c) are an integral part of the transactions contemplated by this Agreement, and that, without these agreements, the Company would not enter into this Agreement; accordingly, if Parent fails to promptly pay the amount due pursuant to this Section 8.5(c), and, in order to obtain such payment, the Company commences a suit that results in a judgment against Parent and/or Merger Sub for the fee set forth in this Section 8.5(c) or any portion of such fee, Parent shall pay to the Company its costs and expenses (including attorneys fees) in connection with such suit, together with interest on the amount of the fee at the prime rate published in the Money Rates section of The Wall Street Journal in effect on the date such payment was required to be made. Notwithstanding anything to the contrary in this Agreement, the parties hereby acknowledge that in the event that the Termination Fee becomes payable and is paid by Parent pursuant to this Section 8.5(c) (provided that (i) neither Parent nor Merger Sub has breached any obligation under Section 6.1(c) (but only with respect to (x) clause (2) of the first sentence thereof and (y) the second sentence thereof) or Section 6.4(b) of this Agreement and (ii) the Stichting has not breached any obligation under the Parent Stockholder Commitment), then the Termination Fee shall be the Company s sole and exclusive remedy for monetary damages under this Agreement.

#### ARTICLE IX

#### Miscellaneous and General

9.1. <u>Survival</u>. This Article IX and the agreements of the Company, Parent and Merger Sub contained in Article IV and Sections 6.9 (Employee Benefits), 6.11 (Expenses), 6.12 (Indemnification; Directors and Officers Insurance) and Section 6.17 (Other Matters) shall survive the consummation of the Merger. This Article IX and the agreements of the Company, Parent and Merger Sub contained in Section 6.11 (Expenses) and Section 8.5 (Effect of Termination and Abandonment) and the Confidentiality Agreement (as defined in Section 9.7) shall survive the termination of this

Agreement. All other representations, warranties, covenants and agreements in this Agreement shall not survive the consummation of the Merger or the termination of this Agreement.

A-40

#### **Table of Contents**

- 9.2. <u>Modification or Amendment</u>. Subject to the provisions of the applicable Laws, at any time prior to the Effective Time, the parties hereto may modify or amend this Agreement, by written agreement executed and delivered by duly authorized officers of the respective parties.
- 9.3. <u>Waiver of Conditions</u>. The conditions to each of the parties obligations to consummate the Merger are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable Laws.
- 9.4. *Counterparts*. This Agreement may be executed in any number of counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts shall together constitute the same agreement.
- 9.5. <u>GOVERNING LAW AND VENUE; WAIVER OF JURY TRIAL; SPECIFIC PERFORMANCE</u>. (a) THIS AGREEMENT SHALL BE DEEMED TO BE MADE IN AND IN ALL RESPECTS SHALL BE INTERPRETED, CONSTRUED AND GOVERNED BY AND IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES THEREOF.
- (b) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 9.5.
- (c) (i) The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that, except where this Agreement is terminated in accordance with Article VIII, the parties shall be entitled to an injunction or injunctions to prevent breaches or threatened breaches of this Agreement and to specifically enforce the terms and provisions of this Agreement (including Section 6.10) and any other agreement or instrument executed in connection herewith and this right shall include the right of the Company to cause Parent and Merger Sub to seek to enforce the terms of the Facilities against the Lenders to the fullest extent permissible pursuant to such Facilities and applicable Laws and to thereafter cause the Merger to be consummated, in each case, if the conditions set forth in Section 7.1 and Section 7.2 have been satisfied or waived (other than conditions which by their nature cannot be satisfied until Closing, but subject to the satisfaction or waiver of those conditions at Closing). Any action or proceeding for any such remedy shall be brought exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, only if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), and each party waives any requirement for the securing or posting of any bond in connection with any such remedy, it being acknowledged that the rights of the Parent and Merger Sub against the Lenders are governed by the Facilities and English Law. The parties further agree that (x) by seeking the remedies provided for in this Section 9.5(c), a party shall not in any respect waive its right to seek any other form of relief that may be available to a party under this Agreement, including monetary damages in the event that this Agreement has been terminated or in the event that the remedies provided for in this Section 9.5(c) are not available or otherwise are not granted and (y) nothing contained in this Section 9.5(c) shall require any party to institute any proceeding for (or limit any party s right to institute any proceeding for) specific

performance under this Section 9.5(c) before exercising any termination right under Article VIII (and pursuing damages after such termination) nor shall the commencement of any Action pursuant to this Section 9.5(c) or anything contained in this Section 9.5(c) restrict or limit any party s right to terminate this Agreement in accordance with the terms of Article VIII or pursue any other remedies under this Agreement that may be available then or thereafter.

A-41

#### **Table of Contents**

- (ii) Each of the parties hereto (A) irrevocably consents to the service of the summons and complaint and any other process in any action or proceeding relating to the transactions contemplated by this Agreement, on behalf of itself or its property, in accordance with Section 9.6 or in such other manner as may be permitted by Law, of copies of such process to such party, and nothing in this Section 9.5(c) shall affect the right of any party to serve legal process in any other manner permitted by Law, (B) irrevocably and unconditionally consents and submits itself and its property in any action or proceeding to the exclusive general jurisdiction of the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, only if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware) in the event any dispute arises out of this Agreement or the transactions contemplated by this Agreement, or for recognition and enforcement of any judgment in respect thereof, (C) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court, (D) agrees that any actions or proceedings arising in connection with this Agreement or the transactions contemplated by this Agreement shall be brought, tried and determined only in the Delaware Court of Chancery (or, only if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware), (E) waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same and (F) agrees that it will not bring any action relating to this Agreement or the transactions contemplated by this Agreement in any court other than the aforesaid courts. Each of Parent, Merger Sub and the Company agrees that a final judgment in any action or proceeding in such court as provided above shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law.
- (iii) Each of Parent and Merger Sub hereby irrevocably appoint the persons listed under Sullivan & Cromwell LLP in Section 9.6 as agent for service of process, to receive on behalf of such party service of copies of the summons and complaint and any other process which may be served in any action or proceeding arising out of or in connection with this Agreement or the transactions contemplated hereby and agrees that process may be served on such persons by the methods provided for giving notice in Section 9.6.
- 9.6. <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the others shall be in writing and delivered personally or sent by registered or certified mail, postage prepaid, by facsimile or overnight courier:

#### If to Parent or Merger Sub:

InBev N.V./S.A.
Brouwerijplein 1
3000 Leuven, Belgium
Attention: Sabine Chalmers, Chief Legal Officer

Fax: +32 16 50 71 11

with a copy to

Sullivan & Cromwell LLP, 125 Broad Street, New York, NY 10004 Attention: James C. Morphy Francis J. Aquila George Sampas

Fax: (212) 558-3588

#### If to the Company:

Anheuser-Busch Companies Inc. One Busch Place, St Louis, MO 63118

Attention: Gary Rutledge Fax: 314-577-2900

A-42

#### **Table of Contents**

with a copy to

Skadden, Arps, Slate, Meagher & Flom LLP Four Times Square, New York, NY 10036

Attention: Paul T. Schnell Thomas W. Greenberg

Fax: (212) 735-2000

or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above. Any notice, request, instruction or other document given as provided above shall be deemed given to the receiving party upon actual receipt, if delivered personally; three (3) business days after deposit in the mail, if sent by registered or certified mail; upon confirmation of successful transmission if sent by facsimile (<u>provided</u> that if given by facsimile such notice, request, instruction or other document shall be followed up within one business day by dispatch pursuant to one of the other methods described herein); or on the next business day after deposit with an overnight courier, if sent by an overnight courier.

- 9.7. Entire Agreement. This Agreement (including any exhibits hereto), the Company Disclosure Schedule, the Parent Shareholder Commitment and the Confidentiality Agreement, dated July 10, 2008, between Parent and the Company (the Confidentiality Agreement ) constitute the entire agreement, and supersede all other prior agreements, understandings, representations and warranties, both written and oral, among the parties, with respect to the subject matter hereof. EACH PARTY HERETO AGREES THAT, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS AGREEMENT, NEITHER PARENT AND MERGER SUB NOR THE COMPANY MAKES ANY OTHER REPRESENTATIONS OR WARRANTIES, AND EACH HEREBY DISCLAIMS ANY OTHER REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, OR AS TO THE ACCURACY OR COMPLETENESS OF ANY OTHER INFORMATION, MADE BY, OR MADE AVAILABLE BY, ITSELF OR ANY OF ITS REPRESENTATIVES, WITH RESPECT TO, OR IN CONNECTION WITH, THE NEGOTIATION, EXECUTION OR DELIVERY OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE OTHER OR THE OTHER S REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION WITH RESPECT TO ANY ONE OR MORE OF THE FOREGOING.
- 9.8. No Third Party Beneficiaries. Except for: (a) only following the Effective Time, the right of the Company s shareholders to receive (i) the Per Share Merger Consideration in respect of Shares pursuant to Section 4.1(a), (ii) the aggregate consideration payable in respect of Company Options pursuant to Section 4.3(a) and (iii) the aggregate consideration payable in respect of Company Awards pursuant to Section 4.3(b); (b) the right of the Company on behalf of its shareholders to pursue damages (including claims for damages based on loss of the economic benefits of the transaction to the Company s stockholders) in the event of Parent s or Merger Sub s breach of this Agreement (whether or not the Agreement has been terminated pursuant to Article VIII), which right is hereby expressly acknowledged and agreed by Parent and Merger Sub; and (c) the right of the Indemnified Parties to enforce the provisions of Section 6.12 (Indemnification; Directors and Officers Insurance) only, (1) Parent and the Company hereby agree that their respective representations, warranties and covenants set forth herein are solely for the benefit of the other party hereto, in accordance with and subject to the terms of this Agreement, and (2) this Agreement is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, including the right to rely upon the representations and warranties set forth herein. The third-party beneficiary rights referenced in clause (b) of the preceding sentence may be exercised only by the Company (on behalf of its stockholders as their agent) through actions expressly approved by the Company s Board of Directors, and no shareholder of the Company whether purporting to act in its capacity as a shareholder or purporting to assert any right (derivatively or otherwise) on behalf of the Company, shall have any right or ability to exercise or cause the exercise of any such right.

9.9. <u>Obligations of Parent and of the Company</u>. Whenever this Agreement requires a Subsidiary of Parent to take any action, such requirement shall be deemed to include an undertaking on the part of Parent to cause such Subsidiary to take such action. Whenever this Agreement requires a Subsidiary of the Company to take any action, such requirement shall be deemed to include an undertaking on the part of the Company to

A-43

#### **Table of Contents**

cause such Subsidiary to take such action and, after the Effective Time, on the part of the Surviving Corporation to cause such Subsidiary to take such action.

- 9.10. <u>Transfer Taxes</u>. All transfer, documentary, sales, use, stamp, registration and other such Taxes and fees (including penalties and interest) incurred in connection with the Merger shall be paid by Parent and Merger Sub when due, and Parent and Merger Sub will indemnify the Company against liability for any such taxes.
- 9.11. <u>Definitions</u>. Each of the terms set forth in Annex A is defined in the Section of this Agreement set forth opposite such term.
- 9.12. <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application of such provision to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application of such provision, in any other jurisdiction.
- 9.13. *Interpretation: Construction*. (a) The table of contents and headings herein are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to a Section, Annex or Exhibit, such reference shall be to a Section of, Annex to or Exhibit to this Agreement unless otherwise indicated. Whenever the words include, includes or including are used in this Agreement, they shall be deemed to be followed by the words without limitation. A reference in this Agreement to \$ or dollars is to U.S. dollars.
- (b) The parties have participated jointly in negotiating and drafting this Agreement. In the event that an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (c) Subject to the introductory language to Section 5.1 and Section 5.2, each party to this Agreement has or may have set forth information in its respective disclosure schedule in a section of such disclosure schedule that corresponds to the section of this Agreement to which it relates. The fact that any item of information is disclosed in a disclosure schedule to this Agreement shall not constitute an admission by such party that such item is material, that such item has had or would have a Material Adverse Effect or that the disclosure of such be construed to mean that such information is required to be disclosed by this Agreement.
- 9.14. *Assignment*. This Agreement shall not be assignable by operation of law or otherwise; <u>provided</u>, <u>however</u>, that Parent may designate, by written notice to the Company, another wholly owned direct or indirect subsidiary to be a Constituent Corporation in lieu of Merger Sub, in which event all references herein to Merger Sub shall be deemed references to such other subsidiary, except that all representations and warranties made herein with respect to Merger Sub as of the date of this Agreement shall be deemed representations and warranties made with respect to such other subsidiary as of the date of such designation and, as applicable pursuant to Section 7.3(a), as of the Closing Date; <u>provided</u> that any such designation shall not materially impede or delay the consummation of the transactions contemplated by this Agreement or otherwise materially impede the rights of the stockholders of the Company under this Agreement. Any purported assignment in violation of this Agreement is void.

#### **Table of Contents**

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first written above.

ANHEUSER-BUSCH COMPANIES, INC.

By: /s/ August A. Busch IV

Name: August A. Busch IV

Title: President and Chief Executive Officer

INBEV N.V./S.A.

By: /s/ Carlos Brito

Name: Carlos Brito

Title: Chief Executive Officer

By: /s/ Sabine Chalmers

Name: Sabine Chalmers

Title: Chief Legal Officer

PESTALOZZI ACQUISITION CORP.

By: /s/ Sabine Chalmers

Name: Sabine Chalmers

Title: Secretary

A-45

#### ANNEX A

## **DEFINED TERMS**

Terms	Section
Acquisition Proposal	6.2(b)
Actions	5.1(g)
Affiliate	5.1(e)(ii)
Agreement	Preamble
Alternative Acquisition Agreement	6.2(c)(ii)
Applicable Date	5.1(e)(i)
Bankruptcy and Equity Exception	5.1(c)(i)
Book-Entry Shares	4.1(a)
Bridge Facility	5.2(d)(ii)
Bridge Lenders	5.2(d)(ii)
business day	1.2
By-Laws	2.2
Certificate	4.2(a)
Change of Recommendation	6.2(c)(iii)
Charter	2.1
Closing	1.2
Closing Date	1.2
Code	4.2(g)
Company	Preamble
Company Approvals	5.1(d)(i)
Company Awards	4.3(b)
Company Benefit Plans	5.1(h)(i)
Company Disclosure Schedule	5.1
Company ERISA Plan	5.1(h)(ii)
Company IP	5.1(p)(i)
Company Labor Agreements	5.1(o)
Company Non-U.S. Benefit Plans	5.1(h)(i)
Company Option	4.3(a)
Company Pension Plan	5.1(h)(ii)
Company Recommendation	5.1(c)(ii)
Company Reports	5.1(e)(i)
Company U.S. Benefit Plans	5.1(h)(ii)
Confidentiality Agreement	9.7
Consent Solicitation	6.15(b)
Constituent Corporations	Preamble
Contract	5.1(d)(ii)
D&O Insurance	6.12(c)
Delaware Certificate of Merger	1.3
Detriments	6.5(b)
Detriments Limit	6.5(b)
DGCL	1.1
Dissenting Stockholders	4.1(a)

## A-46

## **Table of Contents**

Terms	Section
DTC	4.2(b)
Effective Time	1.3
Employees	5.1(h)(i)
Encumbrance	5.1(k)(iv)
Environmental Law	5.1(m)
Equity Financing Filings	5.2(c)(i)
Equity Financings	6.10(b)
ERISA A SCILLAR	5.1(h)(i)
ERISA Affiliate	5.1(h)(iii)
Exchange Act	5.1(a)
Exchange Fund Excluded Share	4.1(a) 4.1(a)
Excluded Shares	4.1(a) 4.1(a)
Facilities	5.2(d)(ii)
Financing	5.2(d)(ii)
Financing Date	5.2(c)(i)5.2(d)(i)
GAAP	5.1(a)
Governmental Antitrust Entity	6.5(e)
Governmental Entity	5.1(d)(i)
Hazardous Substance	5.1(m)
HSR Act	5.1(d)(i)
Indemnified Parties	6.12
Intellectual Property	5.1(p)(iv)
IRS	5.1(h)(ii)
Joint Venture	5.1(a)
Judgment	5.1(g)
knowledge	5.1(f)
Laws	5.1(i)
Leased Real Property Lenders	5.1(k)(ii) 5.2(d)(ii)
Liabilities	5.1(g)
Lien	5.1(g) 5.1(b)(i)
Material Adverse Effect	5.1(a)
Material Contract	5.1(j)
Merger	Preamble
Merger Sub	Preamble
Multiemployer Plan	5.1(h)(ii)
Non-Wholly Owned Subsidiaries	5.1(b)(i)
NYSE	5.1(e)(ii)
Order	7.1(c)
Owned Real Property	5.1(k)(i)
Parent	Preamble
Parent Approvals	5.2(c)(i)
Parent Common Stock	5.2(b)
A-47	

## **Table of Contents**

Terms	Section
Parent Disclosure Schedule	5.2
Parent Prospectus	6.3(b)
Parent Shareholder Commitment	Preamble
Parent Shareholder Meeting Materials	6.3(b)
Parent Shareholders Meeting	6.4(b)
Paying Agent	4.2(a)
PBGC	5.1(h)(iii)
Per Share Merger Consideration	4.1(a)
Person	4.2(d)
Proxy Statement	6.3(a)
Registered	5.1(p)(iv)
Representatives	6.2(a)
Requisite Company Vote	5.1(c)(i)
Requisite Parent Vote	5.2(b)
Sarbanes-Oxley Act	5.1(e)(i)
SEC	5.1
Securities Act	5.1(e)(i)
Senior Facilities Agreement	5.2(d)(ii)
Senior Lenders	5.2(d)(ii)
Share	4.1(a)
Shares	4.1(a)
Significant Subsidiary	5.1(a)
Software	5.1(p)(iv)
Solvent	5.1(r)
Stichting	Preamble
Stock Plans	5.1(b)(i)
Stockholders Meeting	6.4(a)
Subsidiary	5.1(a)
Superior Proposal	6.2(b)
Surviving Corporation	1.1
Takeover Statute	5.1(1)
Tax	5.1(n)
Tax Return	5.1(n)
Taxes	5.1(n)
Termination Date	8.2
Termination Fee	8.5(b)
Trade Secrets	5.1(p)(iv)
Wholly Owned Subsidiaries	5.1(b)(i)
A-48	

Annex B

#### PERSONAL AND CONFIDENTIAL

July 13, 2008

Board of Directors Anheuser-Busch Companies, Inc. One Busch Place St. Louis, MO 63118-1852

#### Ladies and Gentlemen:

You have requested our opinion as to the fairness from a financial point of view to the holders (other than InBev NV/SA ( InBev ) and its direct and indirectly owned subsidiaries) of outstanding shares of common stock, par value \$1.00 per share (the Shares ), of Anheuser-Busch Companies, Inc. (the Company ) of the \$70.00 per Share in cash to be received by such holders pursuant to the Agreement and Plan of Merger, dated as of July 13, 2008 (the Agreement ), by and among InBev, Pestalozzi Acquisition Corp., a wholly owned subsidiary of InBev, and the Company.

Goldman, Sachs & Co. and its affiliates are engaged in investment banking and financial advisory services, securities trading, investment management, principal investment, financial planning, benefits counseling, risk management, hedging, financing, brokerage activities and other financial and non-financial activities and services for various persons and entities. In the ordinary course of these activities and services, Goldman, Sachs & Co., and its affiliates may at any time make or hold long or short positions and investments, as well as actively trade or effect transactions, in the equity, debt and other securities (or related derivative securities) and financial instruments (including bank loans and other obligations) of the Company, InBev and any of their respective affiliates or any currency or commodity that may be involved in the transaction contemplated by the Agreement (the Transaction ) for their own account and for the accounts of their customers. We have acted as financial advisor to the Company in connection with, and have participated in certain of the negotiations leading to, the Transaction. We expect to receive fees for our services in connection with the Transaction, the principal portion of which is contingent upon consummation of the Transaction, and the Company has agreed to reimburse our expenses and indemnify us against certain liabilities arising out of our engagement. In addition, we have provided certain investment banking and other financial services to the Company and its affiliates from time to time, including having acted as co-manager with respect to a public offering of the Company s Senior Unsecured 5.60% Investment Grade Bonds due March 2017 (aggregate principal amount \$300,000,000) in February 2007; as co-manager with respect to a public offering of the Company s Investment Grade Bonds due September 2037 (aggregate principal amount \$500,000,000) in August 2007; and as joint bookrunning manager with respect to a public debt offering (aggregate principal amount \$500,000,000) in November 2007. We also have provided certain investment banking and other financial services to InBev and its affiliates from time to time, including having acted as an arranger in the lending syndicate in connection with InBev s 2,500,000,000 bank loan in December 2005. We also may provide investment banking and other

B-1

#### **Table of Contents**

financial services to the Company, InBev and their respective affiliates in the future. In connection with the above-described services we have received, and may receive, compensation.

In connection with this opinion, we have reviewed, among other things, the Agreement; annual reports to stockholders and Annual Reports on Form 10-K of the Company for the three fiscal years ended December 31, 2007; certain interim reports to stockholders and Quarterly Reports on Form 10-Q of the Company; certain other communications from the Company and InBev to the stockholders of the Company; certain publicly available research analyst reports for the Company; and certain internal financial analyses and forecasts for the Company prepared by its management and approved for our use by the Company (the Forecasts ). We also have held discussions with senior management and the Board of Directors of the Company regarding their assessment of the past and current business operations, financial condition and future prospects of the Company, including their views on the risks and uncertainties associated with achieving the Forecasts. In addition, we have reviewed the reported price and trading activity for the Shares, companies the securities of which are publicly traded, reviewed the financial terms of certain recent business combinations in the beverage industry specifically and in other industries generally and performed such other studies and analyses, and considered such other factors, as we considered appropriate.

For purposes of rendering this opinion, we have relied upon and assumed, without assuming any responsibility for independent verification, the accuracy and completeness of all of the financial, legal, regulatory, tax, accounting and other information provided to, discussed with or reviewed by us. In addition, we have not made an independent evaluation or appraisal of the assets and liabilities (including any contingent, derivative or off-balance-sheet assets and liabilities) of the Company or any of its subsidiaries and we have not been furnished with any such evaluation or appraisal. Our opinion does not address any legal, regulatory, tax or accounting matters.

Our opinion does not address the underlying business decision of the Company to engage in the Transaction, or the relative merits of the Transaction as compared to any strategic alternatives that may be available to the Company. This opinion addresses only the fairness from a financial point of view, as of the date hereof, of the \$70.00 per Share in cash to be received by the holders (other than InBev and its direct and indirectly owned subsidiaries) of Shares pursuant to the Agreement. We do not express any view on, and our opinion does not address, any other term or aspect of the Agreement or Transaction, including, without limitation, the fairness of the Transaction to, or any consideration received in connection therewith by, the holders of any other class of securities, creditors, or other constituencies of the Company or InBev; nor as to the fairness of the amount or nature of any compensation to be paid or payable to any of the officers, directors or employees of the Company or InBev, or class of such persons in connection with the Transaction, whether relative to the \$70.00 per Share in cash to be received by the holders (other than InBev and its direct and indirectly owned subsidiaries) of Shares pursuant to the Agreement or otherwise. Our opinion is necessarily based on economic, monetary, market and other conditions as in effect on, and the information made available to us as of, the date hereof; we assume no responsibility for updating, revising or reaffirming this opinion based on circumstances, developments or events occurring after the date hereof. Our advisory services and the opinion expressed herein are provided for the information and assistance of the Board of Directors of the Company in connection with its consideration of the Transaction and such opinion does not constitute a recommendation as to how any holder of Shares should vote with respect to the Transaction or any other matter. This opinion has been approved by a fairness committee of Goldman, Sachs & Co.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the \$70.00 per Share in cash to be received by the holders (other than InBev and its direct and indirectly owned subsidiaries) of Shares pursuant to the Agreement is fair from a financial point of view to such holders.

Very truly yours,

/s/ Goldman, Sachs & Co. GOLDMAN, SACHS & CO.

Annex C

388 Greenwich Street New York, NY 10013

July 13, 2008

The Board of Directors Anheuser-Busch Companies, Inc. One Busch Place St. Louis, Missouri 63118

#### Members of the Board:

You have requested our opinion as to the fairness, from a financial point of view, to the holders of the common stock of Anheuser-Busch Companies, Inc. (the Company ) of the Merger Consideration (defined below) to be received by such holders pursuant to the terms and subject to the conditions set forth in the Agreement and Plan of Merger, dated as of July 13, 2008 (the Merger Agreement ), among the Company, a Delaware corporation, InBev NV, a public company organized under the laws of Belgium ( Parent ), and Pestalozzi Acquisition Corp. ( Merger Sub ), a Delaware corporation and a wholly owned subsidiary of Parent. As more fully described in the Merger Agreement, (i) Merger Sub will be merged with and into the Company (the Merger ) and (ii) each outstanding share of the common stock, par value \$1.00 per share, of the Company ( Company Common Stock ) will be converted into the right to receive \$70 in cash (the Merger Consideration ).

In arriving at our opinion, we reviewed the Merger Agreement and held discussions with certain senior officers, directors and other representatives and advisors of the Company and certain senior officers and other representatives and advisors of Parent concerning the business, operations and prospects of the Company. We examined certain publicly available business and financial information relating to the Company as well as certain financial forecasts and other information and data relating to the Company which were provided to or discussed with us by the management of the Company. We reviewed the financial terms of the Merger as set forth in the Merger Agreement in relation to, among other things: current and historical market prices and trading volumes of the Company Common Stock; the historical and projected earnings and other operating data of the Company; and the capitalization and financial condition of the Company. We considered, to the extent publicly available, the financial terms of certain other transactions which we considered relevant in evaluating the Merger and analyzed certain financial, stock market and other publicly available information relating to the businesses of other companies whose operations we considered relevant in evaluating those of the Company. In addition to the foregoing, we conducted such other analyses and examinations and considered such other information and financial, economic and market criteria as we deemed appropriate in arriving at our opinion. The issuance of our opinion has been authorized by our fairness opinion committee.

In rendering our opinion, we have assumed and relied, without independent verification, upon the accuracy and completeness of all financial and other information and data publicly available or provided to or otherwise reviewed by or discussed with us and upon the assurances of the management of the Company that they are not aware of any relevant information that has been omitted or that remains undisclosed to us. With respect to financial forecasts and other information and data relating to the Company provided to or otherwise reviewed by or discussed with us, we have been advised by the management of the Company that such forecasts and other information and data were

reasonably prepared on bases reflecting the best currently available estimates and judgments of the management of the Company as to the future financial performance of the Company.

We have assumed, with your consent, that the Merger will be consummated in accordance with its terms, without waiver, modification or amendment of any material term, condition or agreement and that, in the course of obtaining the necessary regulatory or third party approvals, consents and releases for the Merger, no

C-1

#### **Table of Contents**

delay, limitation, restriction or condition will be imposed that would have a material adverse effect on the Merger, or the parties ability to effect the Merger. We have not made or been provided with an independent evaluation or appraisal of the assets or liabilities (contingent or otherwise) of the Company nor have we made any physical inspection of the properties or assets of the Company. Our opinion does not address the underlying business decision of the Company to effect the Merger, the relative merits of the Merger as compared to any alternative business strategies that might exist for the Company or the effect of any other transaction in which the Company might engage. We also express no view as to, and our opinion does not address, the fairness (financial or otherwise) of the amount or nature or any other aspect of any compensation to any officers, directors or employees of any parties to the Merger, or any class of such persons, relative to the Merger Consideration. Our opinion is necessarily based upon information available to us, and financial, stock market and other conditions and circumstances existing, as of the date hereof.

Citigroup Global Markets Inc. has acted as financial advisor to the Company in connection with the proposed Merger, including the delivery of this opinion, and will receive a fee for such services, a significant portion of which is contingent upon the consummation of the Merger. We and our affiliates in the past have provided services to the Company and Parent unrelated to the proposed Merger, for which services we and such affiliates have received compensation, including, without limitation, (i) for the Company: acting as co-manager on a \$500 million 10-year debt issuance in November 2007; co-manager on a \$500 million debenture offering in August 2007; bookrunner on a \$300 million senior notes issuance in February 2007; lender in connection with a \$55 million financing commitment; joint document agent on a \$500 million revolving credit facility in February 2008, and (ii) for the Parent: in December 2007, providing a fairness opinion with respect to AmBev s acquisition of the remaining stake in Quilmes Industrial SA ( Quinsa ); in July 2007, acting as bookrunner on AmBev s BRL \$300 million senior unsecured notes due 2017; in July 2006, acting as bookrunner on AmBev s BRL \$2 billion local debentures (two tranches) to fund the acquisition of the remaining shares of Quinsa. In the ordinary course of our business, we and our affiliates may actively trade or hold the securities of the Company and Parent for our own account or for the account of our customers and, accordingly, may at any time hold a long or short position in such securities. In addition, we and our affiliates (including Citigroup Inc. and its affiliates) may maintain relationships with the Company, and the Parent and their respective affiliates.

Our advisory services and the opinion expressed herein are provided for the information of the Board of Directors of the Company in its evaluation of the proposed Merger, and our opinion is not intended to be and does not constitute a recommendation to any stockholder as to how such stockholder should vote or act on any matters relating to the proposed Merger.

Based upon and subject to the foregoing, our experience as investment bankers, our work as described above and other factors we deemed relevant, we are of the opinion that, as of the date hereof, the Merger Consideration is fair, from a financial point of view, to the holders of the Company Common Stock.

Very truly yours,

/s/ Citigroup Global Markets Inc.
CITIGROUP GLOBAL MARKETS INC.

C-2

Annex D

#### GENERAL CORPORATION LAW OF THE STATE OF DELAWARE

#### **Delaware Code**

# TITLE 8 Corporations CHAPTER 1. GENERAL CORPORATION LAW Subchapter IX. Merger, Consolidation or Conversion

#### 8 Del. C. § 262. Appraisal rights.

- (a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to § 228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder s shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word stockholder means a holder of record of stock in a stock corporation and also a member of record of a nonstock corporation; the words stock and share mean and include what is ordinarily meant by those words and also membership or membership interest of a member of a nonstock corporation; and the words depository receipt mean a receipt or other instrument issued by a depository representing an interest in one or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.
- (b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to § 251 (other than a merger effected pursuant to § 251(g) of this title), § 252, § 254, § 257, § 258, § 263 or § 264 of this title:
  - (1) Provided, however, that no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the agreement of merger or consolidation, were either (i) listed on a national securities exchange or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in subsection (f) of § 251 of this title.
  - (2) Notwithstanding paragraph (1) of this subsection, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§ 251, 252, 254, 257, 258, 263 and 264 of this title to accept for such stock anything except:
    - (a) Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;

(b)

Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or held of record by more than 2,000 holders;

(c) Cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a. and b. of this paragraph; or

D-1

#### **Table of Contents**

- (d) Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a., b. and c. of this paragraph.
- (3) In the event all of the stock of a subsidiary Delaware corporation party to a merger effected under § 253 of this title is not owned by the parent corporation immediately prior to the merger, appraisal rights shall be available for the shares of the subsidiary Delaware corporation.
- (c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as is practicable.
- (d) Appraisal rights shall be perfected as follows:
  - If a proposed merger or consolidation for which appraisal rights are provided under this section is to (1) be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for such meeting with respect to shares for which appraisal rights are available pursuant to subsection (b) or (c) hereof that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section. Each stockholder electing to demand the appraisal of such stockholder s shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder s shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder s shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or
  - (2) If the merger or consolidation was approved pursuant to § 228 or § 253 of this title, then either a constituent corporation before the effective date of the merger or consolidation or the surviving or resulting corporation within 10 days thereafter shall notify each of the holders of any class or series of stock of such constituent corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder s shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder s shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the

merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send such a second notice to all such holders on or within 10 days after such effective date; provided, however, that if

D-2

#### **Table of Contents**

such second notice is sent more than 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder s shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

- (e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) of this section hereof and who is otherwise entitled to appraisal rights, may commence an appraisal proceeding by filing a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after the effective date of the merger or consolidation, any stockholder who has not commenced an appraisal proceeding or joined that proceeding as a named party shall have the right to withdraw such stockholder s demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) of this section hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder s written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) of this section hereof, whichever is later. Notwithstanding subsection (a) of this section, a person who is the beneficial owner of shares of such stock held either in a voting trust or by a nominee on behalf of such person may, in such person s own name, file a petition or request from the corporation the statement described in this subsection.
- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by 1 or more publications at least 1 week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of

#### **Table of Contents**

the appraisal proceedings; and if any stockholder fails to comply with such direction, the Court may dismiss the proceedings as to such stockholder.

- (h) After the Court determines the stockholders entitled to an appraisal, the appraisal proceeding shall be conducted in accordance with the rules of the Court of Chancery, including any rules specifically governing appraisal proceedings. Through such proceeding the Court shall determine the fair value of the shares exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take into account all relevant factors. Unless the Court in its discretion determines otherwise for good cause shown, interest from the effective date of the merger through the date of payment of the judgment shall be compounded quarterly and shall accrue at 5% over the Federal Reserve discount rate (including any surcharge) as established from time to time during the period between the effective date of the merger and the date of payment of the judgment. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceeding, the Court may, in its discretion, proceed to trial upon the appraisal prior to the final determination of the stockholders entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder s certificates of stock to the Register in Chancery, if such is required, may participate fully in all proceedings until it is finally determined that such stockholder is not entitled to appraisal rights under this section.
- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Payment shall be so made to each such stockholder, in the case of holders of uncertificated stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court s decree may be enforced as other decrees in the Court of Chancery may be enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.
- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney s fees and the fees and expenses of experts, to be charged pro rata against the value of all the shares entitled to an appraisal.
- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder s demand for an appraisal and an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection (e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just; provided, however that this provision shall not affect the right of any stockholder who has not commenced an appraisal proceeding or joined that proceeding as a named party to withdraw such stockholder s demand for appraisal and to accept the terms offered upon the merger or consolidation within 60 days after the effective date of the merger or consolidation, as set forth in subsection (e) of this section.

(l) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of

D-4

#### **Table of Contents**

authorized and unissued shares of the surviving or resulting corporation.(8 Del. C. 1953, § 262; 56 Del. Laws, c. 50; 56 Del. Laws, c. 186, § 24; 57 Del. Laws, c. 148, §§ 27-29; 59 Del. Laws, c. 106, § 12; 60 Del. Laws, c. 371, §§ 3-12; 63 Del. Laws, c. 25, § 14; 63 Del. Laws, c. 152, §§ 1, 2; 64 Del. Laws, c. 112, §§ 46-54; 66 Del. Laws, c. 136, §§ 30-32; 66 Del. Laws, c. 352, § 9; 67 Del. Laws, c. 376, §§ 19, 20; 68 Del. Laws, c. 337, §§ 3, 4; 69 Del. Laws, c. 61, § 10; 69 Del. Laws, c. 262, §§ 1-9; 70 Del. Laws, c. 79, § 16; 70 Del. Laws, c. 186, § 1; 70 Del. Laws, c. 299, §§ 2, 3; 70 Del. Laws, c. 349, § 22; 71 Del. Laws, c. 120, § 15; 71 Del. Laws, c. 339, §§ 49-52; 73 Del. Laws, c. 82, § 21; 76 Del. Laws, c. 145, §§ 11-16.)

D-5

Preliminary Copy Subject to Completion, dated August 15, 2008

[INSERT LOGO]
ADMISSION TICKET
Special Meeting of Stockholders
Anheuser-Busch Companies, Inc.
ADMIT ONE

[INSERT LOGO]
ADMISSION TICKET
Special Meeting of Stockholders
Anheuser-Busch Companies, Inc.
ADMIT ONE

You should present this admission ticket in order to gain admittance to the , 2008 Special Meeting.

This ticket admits only the stockholder(s) listed on the reverse side and is not transferable. If shares are held in the name of a broker, trust, bank, or other nominee, you should bring with you a statement, proxy or letter from the broker, trustee, bank or nominee confirming your beneficial ownership of the shares as of the record date. Cameras, recording devices and other electronic devices will not be permitted at the meeting.

#### **Table of Contents**

Preliminary Copy Subject to Completion, dated August 15, 2008

[INSERT LOGO]

#### **Special Meeting Proxy Card**

## PLEASE REFER TO THE REVERSE SIDE FOR INTERNET AND TELEPHONE VOTING INSTRUCTIONS.

#### A Issues

The Board of Directors recommends a vote FOR the following proposals.

1. Proposal to adopt the Agreement and Plan of Merger by and among InBev N.V./S.A., Pestalozzi Acquisition Corp., and Anheuser-Busch Companies, Inc., as it may be amended from time to time.	For o	Against o	Abstain o	Mark this box with an X if you plan to attend the Special Meeting.	0
2. Proposal to adjourn the Special Meeting to a later date to solicit additional proxies if there are insufficient votes to approve proposal number 1 at the time of the Special Meeting.	For o	Against 0	Abstain o		

## B Authorized Signatures Sign Here This section must be completed for your instructions to be executed.

NOTE: Please sign your name(s) EXACTLY as your name(s) appear(s) on this proxy. All joint holders must sign. When signing as attorney, trustee, executor, administrator, guardian or corporate officer, please provide your FULL title.

Signature 1	Please keep signature within	Signature 2	Please keep signature within	Date (mm/dd/yyyy)
the box		the box		

Title: <u>— —</u>

**Proxy** Anheuser-Busch Companies, Inc.

## THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS OF ANHEUSER-BUSCH COMPANIES, INC.

The undersigned, whose signature appears on the reverse, hereby appoints and each of them, proxies with full power of substitution for and in the name of the undersigned to vote all the shares of common stock of ANHEUSER-BUSCH COMPANIES, INC. which the undersigned would be entitled to vote if personally present at the Special Meeting of Stockholders to be held on , 2008 and at any and all adjournments or postponements thereof, on all matters that may properly come before the meeting. Your shares will be voted as directed on this card. **If signed and no direction is given for any item, it will be voted in favor of items 1 and 2.** The shares represented by this proxy will be voted in the discretion of said proxies with respect to such other business as may properly come before the meeting and any adjournments or postponements thereof. To vote by telephone or Internet, please see below. To vote by mail, please sign and date this card on the reverse side, and mail promptly in the enclosed postage-paid envelope.

This proxy also constitutes your voting instructions for shares held in any of Anheuser-Busch s 401(k) plans (the 401(k) plans ). The undersigned hereby authorizes the respective trustees/administrators of such plans to vote the shares of common stock held in the undersigned s accounts. If a participant fails to instruct the trustee on how to vote his or her shares of common stock, the plan s investment committee will provide instructions as to how those shares should be voted. Participants in the 401(k) Plans must vote by Internet or telephone or complete, date, sign and return this proxy card by Eastern time on , 2008, for shares of common stock represented by this proxy to be voted as directed. Participants may attend the special meeting but may NOT vote their shares of common stock at the meeting.

Your vote is important. By returning your voting instructions promptly, you can avoid the inconvenience of receiving follow-up mailings plus help the Company avoid additional expenses.

## Internet and Telephone Voting Instructions You can vote by telephone OR Internet! Available 24 hours a day 7 days a week!

Instead of mailing your proxy, you may choose one of the two voting methods outlined below to vote your proxy.

To vote using the Internet

instructions.

Call toll free 1 in the United States or Canada any time on a touch tone telephone. There is <b>NO CHARGE</b> to you for the call.	Go to the following web site: http://wwwcom/	
Follow the simple instructions provided by the recorded message.	Enter the information requested on your computer screen and follow the simple	

If you vote by telephone or the Internet, please DO NOT mail back this proxy card. Proxies submitted by telephone or the Internet must be received by , 2008. THANK YOU FOR VOTING

To vote using the Telephone (within U.S. and Canada)